30 APRIL 1967

Technical Library, Bellcomm, Inc. MAY 1 8 1967 LMSC-A847943

Study of Mission Modes and System Analysis for Lunar Exploration

FINAL REPORT

PLANNERS' HANDBOOK

MIMOSA Planning Methodology - Vol. 1

Except Appendixes K and L



(NASA-CE-123333) STUDY OF MISSION MODES AND SYSTEM ANALYSIS FOR LUNAR EXPLORATION (MIMOSA). MIMOSA PLANNING METHODOLOGY, VOLUME 1 (EXCEPT (Lockheed Missiles and Space Co.) 30 Apr. 1967 231 p

LOCKHEED MISSILES & SPACE COMPANY

10 / Q Q

A GROUP DIVISION OF LOCKHEED AIRCRAFT CORPORATION

LMSC-A847943

30 APRIL 1967

Study of Mission Modes and System Analysis for Lunar Exploration

FINAL REPORT

PLANNERS' HANDBOOK

MIMOSA Planning Methodology - Vol. I

Except Appendixes K and L

Prepared Under Contract NAS 8-20262 for GEORGE C. MARSHALL SPACE FLIGHT CENTER HUNTSVILLE, ALABAMA

A GROUP DIVISION OF LOCKHEED AIRCRAFT CORPORATION

FOREWORD

This document is Volume I of MIMOSA Planning Methodology, which constitutes part of the final report on the Study of Mission Modes and Systems Analysis for Lunar Exploration (MIMOSA). This study was conducted by the LMSC MIMOSA team for the George C. Marshall Space Flight Center under contract NAS 8-20262. The entire final report covers work performed from 3 January 1966 to 3 February 1967 and comprises the following parts:

- MIMOSA Summary Digest
- MIMOSA Summary Technical Report

• MIMOSA Technical Report:

Volume I – Lunar Exploration Equipment and Mode Definition Volume II – Candidate Lunar Exploration Programs Volume III – Recommended Lunar Exploration Plan

 MIMOSA Planning Methodology: Volume I – Planners' Handbook
 Volume II – Exploration Equipment Data Book
 Volume III – Scientific Programs

LOCKHEED MISSILES & SPACE COMPANY

ACKNOWLEDGEMENTS

LMSC-A847943

Vol. I

The Study of Mission Modes and Systems Analysis for Lunar Exploration (MIMOSA) was conducted by the MIMOSA team for the George C. Marshall Space Flight Center under contract NAS 8-20262. The contractor team consisted of the following organizations:

- Lockheed Missiles & Space Company. Prime Contractor
- <u>AiResearch Manufacturing Company</u>. Environmental Control and Life Support System
- Bell Aerosystems Company. Lunar Flying Vehicles
- <u>Bendix Corporation, Aerospace Systems Division</u>. Lunar Roving Vehicles Definition and Contributions to Scientific Program Formulation
- <u>General Electric Company, Missile & Space Division</u>. Electrical Power Systems

The technical management of the study was conducted by a government team composed of several organizations and their selected representatives. The Technical Supervisor and Contracting Officers' Representative for the National Aeronautics and Space Administration was David Paul 3rd of the Advanced Systems Office, Marshall Space Flight Center (MSFC). Contributing organizations and their representatives were as follows:

Advanced Systems Office, MSFC

Gordon R. Woodcock	Luther J. Spears
William G. Huber	Robert J. Davies
Terry H. Sharpe	Edward E. Waggoner
Harold D. Trexler	Lynn L. Bradford

iv

LOCKHEED MISSILES & SPACE COMPANY

Propulsion & Vehicle Engineering Laboratory, MSFC Richard A. Love William W. Wales Walter A. Johanson Michael J. Vaccaro Charles R. Darwin Wilburn O. J. Randolph Aero-Astrodynamics Laboratory, MSFC Ronald D. Scott Research Projects Laboratory, MSFC Dr. Nicholas C. Costes James A. Downey III Astrionics Laboratory, MSFC Edward E. Dungan Advanced Manned Missions Program Office, Office of Manned Space Flight, NASA Headquarters Philip E. Culbertson Jerald M. Goldberg Lt. Col. Peter Grosz, USA Donald A. Beattie John F. Kennedy Space Center, NASA Joseph T. Garo Owen L. Sparks A. Robert Raffaelli Warren L. Foss • Manned Spacecraft Center, NASA Thomas R. Kloves Clark Covington Norman Belasco Milton A. Silveria Astrogeology Branch, U. S. Geological Survey Dr. Joseph T. O'Connor Extraterrestrial Research Agency, U. S. Army Corps of Engineers Donald W. Butler James E. Malcolm

Dr. Harold Masursky of the Astrogeology Branch, U.S. Geological Survey, was consulted regarding geological experimentation, and Dr. William G. Tifft, Steward Observatory, University of Arizona, was consulted regarding optical astronomy. In addition, Dr. Alfred H. Webber of St. Louis University provided general advice on scientitic experimentation. The study also drew on certain previous and concurrent studies for information. The most significant of these were as follows:

- <u>Scientific Mission Support for Extended Lunar Exploration</u>. North American Aviation, Inc., Contract NAS 8-20258 (William McKaig, Study Leader)
- Lunar Surface Mobility Systems Comparison and Evolution Study (MOBEV). Bendix Corporation, Aerospace Systems Division, Contract NAS 8-20334 (Carmelo J. Moscolino, Study Leader)
- <u>Early Lunar Shelter Design and Comparison Study</u>. AiResearch Manufacturing Company, Contract NAS 8-20261 (William L. Burriss, Study Leader)

The study was aided by an independent effort on the part of Mr. Wes C. Schmill of Atomics International, a Division of North American Aviation, Inc., who contributed to the definition of nuclear power systems.

LOCKHEED MISSILES & SPACE COMPANY

CONTENTS

Chapter			Page
	FORI	EWORD	iii
	ACKI	NOWLEDGEMENTS	iv
	ILLU	STRATIONS	xiii
	TAB	LES	xiv
	INTR	ODUCTION	xv
1	LUNA	AR PROGRAM ANALYSIS APPROACH	1-1
	1.1	General Planning Procedure	1-1
	1.2	Planners Handbook Organization	1-4
2	PLAN	NNING-QUESTION DETERMINATION	2-1
	2.1	Science Questions	2-1
	2.2	Equipment Questions	2-1
	2.3	Operational Questions	2-2
	2.4	Resources Allocation Questions	2-2
3	SCIE	NTIFIC PROGRAM FORMULATION	3-1
	3.1	Identification of Scientific Goals	3-1
	3.2	Identification of Major Disciplines	3-2
	3.3	Scientific-Investigation Matrix	3-4
	3.4	Selection of Scientific Experiments	3-4
	3.5	Specification of Major Scientific Equipment	3-6
	3.6	Lunar-Surface Exploration Patterns	3-6
	3.7	Distribution of Scientific Experiments	3-9
	3.8	Scientific Program Specification	3-10
4	EXPI DEFI	LORATION EQUIPMENT AND EQUIPMENT-EVOLUTION	4-1
	4.1	Equipment Description	4-1
		4.1.1 Transportation Systems	4-1
		4.1.2 Mission Equipment	4-3

. 1

÷

Chapter			Page
	4.2	Computer Input Preparation for Exploration Equipment	4-3
		4.2.1 Transportation Systems	4-3
		4.2.2 All Mission Equipment	4-5
•		4.2.3 Shelter Only	4-6
		4.2.4 LRV/Trailer Only	4-6
		4.2.5 LFV Only	4-7
		4.2.6 Major Scientific Equipment Only	4-8
		4.2.7 All Systems	4-8
	4.3	Equipment-Evolution Requirements	4-9
	4.4	Sample Equipment Evolutions	4-10
5	MISSI	ON FORMULATION	5-1
	5.1	Mission Sizing	5-1
	5.2	Traverse Formulation	5-2
	5.3	Sample Missions	5-3
	5.4	Computer	5-5
6	MODI	E FORMULATION	6-1
	6.1	Mode Requirements	6-1
	6.2	Sample Mode	6-2
	6.3	Computer Input Preparation	6-2
	6.4	"Working" Modes	6-6
	6.5	"Zero-Launch" Modes	6-6
	6.6	Emergency Modes	6-9
7	MISSI	ION-MODE COMPARISON	7-1
	7.1	Hand Analysis	7-1
	7.2	Computer Input Preparation	7-2
	7.3	Sample Mission-Mode Comparison	7-3
		7.3.1 Computer Printout	7-3
		7.3.2 The Mayday Message	7-15
		7.3.3 Excessive Cargo Message	7-15
		7.3.4 Decimal Point Errors	7-16
		7.3.5 Routine Statements	7-16
		7.3.6 Excessive Earth-Return Mass Statement	7-17

viii

Chapter				Page
		7.3.7	Other Messages	7-17
		7.3.8	Second Iteration	7-18
8	EXPL	ORATION	PROGRAM GENERATION	8-1
	8.1	Mission	Scheduling	8-1
	8.2	Compute	er Usage	8-1
9	PROG	RAM DAT	LA SUMMARY	9-1
10	MIMO	SA COMI	PUTER PROGRAM INPUT INSTRUCTIONS	10-1
	10.1	General	Computer Information	10-1
		10.1.1	Man/Computer Interface	10-3
		10.1.2	Tape Handling Philosophy	10-5
		10.1.3	Sequence Numbers	10-5
	•	10.1.4	Skip, Last, and Stop Cards	10-7
	10.2	Stacking	g Cards for Computer Runs	10-8
	10.3	General	Input Instructions	10-9
		10.3.1	Tape Assignment	10-9
		10.3.2	Selection of Routine	10-17
	10.4	Input an	d Update Control Control Cards	10-17
	10.5	Mission	-Mode Tape Generation	10-19
		10.5.1	Mission Input	10-19
		10.5.2	Mode Input	10-23
		10.5.3	Emergency Mode Input	10-23
	10.6	System,	Experiment Tape Generation	10-26
		10.6.1	System Input	10-26
		10.6.2	Subsystem State-of-the-Art Advance Input	10-31
		10.6.3	Experiment Input	10-32
	10.7	Mission	Input Revision	10-33
		10.7.1	Procedure for Changing Existing Data (Except Identifiers)	10-34
		10.7.2	Procedure for Deleting Whole Missions	10-35
		10.7.3	Procedure for Adding Whole Missions	10-35
		10.7.4	Procedure for Copying a Selected Subset of Missions onto Another Tape	10-36

••

Chapter

•

	10.8	Mode, S Revisior	ystem, Subsystem, and Experiment Input	10-36
		10.8.1	Procedure for Changing an Existing Record	10-37
		10.8.2	Procedure for Deleting Entire Data Records	10-38
		10.8.3	Procedure for Adding Entire Data Records	10-38
		10.8.4	Procedure for Copying an Entire Record File	10-40
		10.8.5	Procedure for Copying a Selected Subset of Records onto Another Tape in any Order	10-40
	10.9	Mission	-Mode Comparison Control Cards	10-40
	10.10	Integrat	ion and Cost Control Cards	10-42
	10.11	Compute	er Input Checklists	10-44
		10.11.1	Mission Input Checklist	10-44
		10.11.2	Mode Input Checklist	10-44
		10.11.3	Mission Mode Match Checklist	10-46
		10.11.4	Exploration Program Cost Checklist	10-47
Appendix				
Α	EXAM	PLE LUN	NAR PROGRAM SUMMARY	A-1
B	DETA INPUT	ILED INS	TRUCTIONS FOR PREPARING MISSION	B-1
C	DETA INPUT	ILED INS CARDS	TRUCTIONS FOR PREPARING MODE	C-1
D	DETA MODE	ILED INS INPUT (TRUCTIONS FOR PREPARING EMERGENCY CARDS	D-1
Ε	DETA INPUT	ILED INS CARDS	TRUCTIONS FOR PREPARING SYSTEM	E-1
F	DETA SOTA	ILED INS ADVANC	TRUCTIONS FOR PREPARING SUBSYSTEM E INPUT CARDS	F-1
G	DETA INPUT	ILED INS CARDS	TRUCTIONS FOR PREPARING EXPERIMENT	G-1
Н	DETA DA TA	ILED INS UPDATE	TRUCTIONS FOR PREPARING COMPUTER	H-1
I	DETA MODE	ILED INS COMPA	TRUCTIONS FOR PREPARING MISSION- RISON CONTROL CARDS	I-1

. .

Appendix		Page
J	DETAILED INSTRUCTIONS FOR PREPARING INTEGRA- TION AND COST CONTROL CARDS	J-1
K	LOGIC FLOW FOR MISSION-MODE COMPARISON ROUTINE	K-1
$\mathbf L$	LOGIC FLOW FOR INTEGRATION AND COST ROUTINE	L-1

•

ILLUSTRATIONS

Figure		Page
1-1	General MIMOSA Approach to Lunar Program Analysis	1-2
3-1	Lunar Surface Exploration Pattern	3-8
4-1	Equipment Evolutions	4-11
5-1	Example Mission	5-4
5-2	Example Computer Printout – Missions	5-7
6-1	Sample Mode Data Sheet	6-3
6-2	Example Computer Printout – Modes	6-7
7-1	Example Computer Printout – Mission-Mode Match No. 1	7-4
7-2	Example Computer Printout – Mission-Mode Match No. 2	7-19
8-1	Example Computer Printout – Integration and Cost	8-3
10-1	Data Handling Procedure Between Man and Computer	10-4
10-2	Computer Card Stack – Input and Update Routine Only	10-10
10-3	Computer Card Stack – Mission-Mode Comparison Routine	10-11
10-4	Computer Card Stack – Integration and Cost Routine	10-12
10-5	Computer Card Stack – Input and Update Routine Following by Integration and Cost Routine	10-13
10-6	Computer Card Stack – Generation of Mission-Mode Tape	10-20
10-7	Computer Card Stack – Generation of Mission Data File	10-21
10-8	Computer Card Stack - Generation of System/Experiment Tape	10-27
10-9	Computer Card Stack – Update of Existing Fixed-Length Data Records	10-39
10-10	Computer Card Stack – Control of Mission-Mode Comparison Routine	10-41
10-11	Computer Card Stack - Control of Integration and Cost Routine	10-45

xiii

• • • • •

LMSC-A847943 Vol. I

TABLES

÷ :

•

Table		Page
3-1	Basic Scientific Questions	3-3
3-2	Elements of an Investigation Matrix for a Typical Space Science Board Question	3-5
3-3	Selected Experiments	3-7
3-4	Major Scientific Equipment	3-9
3-5	"Filler" Experiments for Locales	3-11
4-1	Exploration Equipment	4-2
4-2	Exploration Equipment Computer Inputs	4-4
6-1	Mode Computer Input	6-4
7-1	Scientific/Total-Manhours Rations	7-2
10-1	MIMOSA Computer Program Routines	10-2
10-2	Data Files Used by MIMOSA Computer Program	10-3
10-3	General Input Instruction Cards	10-14
10-4	MIMOSA Tape Assignments	10-15
10-5	Instructions for Filling Out Control Card 1-01	10-16
10-6	Instructions for Filling Out Control Card 1-02	10-17
10-7	Options for Input and Update Routine	10-18
10-8	Instructions for Filling Out Control Card 2-01	10-18
10-9	Mission Input Cards	10-22
10-10	Mode Input Cards	10-24
10-11	Emergency Mode Input Cards	10-25
10-12	System Input Cards	10-28
10-13	Subsystem SOTA Advance Input Cards	10-31
10-14	Format Used for Numbers on Experiment Input Cards	10-33
10-15	Fixed Record Update Input Cards	10-37
10-16	Mission-Mode Comparison Control Cards	10-42
10-17	Integration and Cost Control Cards	10-43

xiv

INTRODUCTION

The objectives of the MIMOSA study were twofold – to produce a methodology for generating lunar exploration programs and to generate a recommended lunar exploration plan, using the developed methodology.

The MIMOSA study was divided into the following three phases:

- Phase I compilation and generation of data for the later phases (these data are contained in the Exploration Equipment Data Book.)
- Phase II development of the planning methodology that includes a computer program for the mechanization of data handling, generation of a broad spectrum of candidate programs, and comparative analysis to answer certain planning questions
- Phase III formulation of a recommended plan of lunar exploration, generation of three selected lunar exploration programs for implementing the plan, and intensive design effort for the equipment used in these three programs

Generation of the recommended lunar exploration plan is described in the MIMOSA Technical Report. The methodology is presented in these volumes under MIMOSA Planning Methodology.

This volume contains instructions to the planner for operating the MIMOSA Planning Methodology; it covers (1) lunar program analysis approach, (2) planning question determination, (3) scientific program formulation, (4) exploration equipment and equipment evolution definition, (5) mission formulation, (6) mode formulation, (7) mission-mode comparison, (8) exploration program generation, (9) program data summary, and (10) MIMOSA computer program input instructions.

xv

Chapter 1 LUNAR PROGRAM ANALYSIS APPROACH

As part of the MIMOSA study contract, Lockheed Missiles & Space Company developed a systematic procedure for analyzing alternate approaches to lunar exploration. This procedure provides a tool for program planners to answer questions concerning scientific requirements, equipment options, operational concepts, and resource restraints. This chapter outlines the significant steps of the procedure and describes the organization of the remainder of this handbook.

1.1 GENERAL PLANNING PROCEDURE

Figure 1-1 indicates the MIMOSA procedure for lunar program analysis. As the figure shows, the procedure consists of the following eight major steps:

- Determine planning questions. Before proceeding with a lunar program analysis, a planner must know his objectives. These objectives are expressed in the form of specific questions to which well-substantiated answers are sought. The answers are obtained from a study of the characteristics of one or more candidate lunar programs. The preciseness with which the questions can be asked will determine how quickly alternate programs with the desired characteristics can be formulated.
- Formulate scientific programs. The goals and requirements of the scientific community are expressed in the form of one or more scientific programs. A scientific program consists primarily of a list of experiments to be conducted at certain locations, at specific frequencies, and with certain priorities for performing those experiments.
- Define exploration equipment and its evolution with time. The exploration equipment to perform the lunar program must be identified (with the specific planning question in mind) and described. Exploration equipment includes transportation equipment (launch vehicles and spacecraft), mission equipment



(shelters, roving vehicles, flying vehicles, base support equipment, etc.), and major science equipment (telescopes, large drills, laboratories, etc.). The desired evolution from one equipment concept to another must exhibit a smooth development of capability. An evolution is established by considering the size and scope of the scientific program and by following certain planning assumptions (e.g., launch constraints, funding rates, timing of major events).

- <u>Formulate missions</u>. With the knowledge of the equipment evolution, planners can allocate the experiments in the scientific program into "missions" of generally increasing scope.
- <u>Determine exploration modes.</u> Planners can group the candidate equipment concepts that are to perform each mission into an exploration "mode." The equipment concepts in a mode must be compatible from both the design and operations points of view.
- <u>Match missions and modes.</u> Requirements of a mission are then compared with the mission-support capability of the mode. This mission-mode comparison can then be verified in detail by a specially developed computer program using the mission, mode, equipment, and experiment data stored in the computer memory. The computer points out any areas of incompatibility between mission requirements and mode capabilities calling for data revision. Additional iterations can be made depending on the accuracy of matching desired. It is this matching process that demonstrates the capability of an equipment evolution to perform a scientific program.
- <u>Generate exploration programs.</u> When all of the missions for a candidate exploration program have been matched satisfactorily with an exploration concept mode, the mission-mode combinations are scheduled and costed. The MIMOSA computer program again performs the tedious calculations. Each scheduled set of mission-mode combinations constitutes a lunar program. Any change in the combinations used, or in the schedule of the missions, results in another program.
- <u>Summarize program data</u>. Planners answer planning questions by studying the characteristics of alternate lunar programs that are generated. To do this, the large amount of detailed program data that such tools as the MIMOSA computer program provide must be summarized.

1.2 PLANNER'S HANDBOOK ORGANIZATION

The general MIMOSA approach forms the basis for organization of the subsequent sections of the Planner's Handbook. Figure 1-1 indicates this organization.

Chapters 2, 3, and 4 discuss the preliminary steps necessary for planning exploration programs. Chapter 2 provides guidance to planners for formulating questions and determining an approach for answering these questions. Chapter 3 deals with the formulation of scientific programs from an analysis of general scientific requirements. Chapter 4 discusses the definition of exploration equipment and equipment evolutions.

Chapters 5, 6, 7, and 8 present steps for formulating candidate exploration programs. Chapter 5 discusses mission generation. Chapter 6 describes the technique for mode formulation. The procedure for comparing mission requirements and mode capabilities using the MIMOSA computer program is presented in Chapter 7. Chapter 8 outlines the method for generating exploration programs from selected mission-mode combinations.

Chapter 9 suggests a format for summarizing the data generated in the formulation of a lunar program.

Finally, Chapter 10 presents a complete set of instructions for using the MIMOSA computer program.

Chapter 2 PLANNING-QUESTION DETERMINATION

Successful use of the MIMOSA lunar program analysis procedure depends on proper formulation of planning questions. The type of question and the conciseness with which it is phrased are both important. Some types of questions require more labor than others to achieve a useful answer, and a vaguely worded question often results in wasted effort.

Planning questions can be grouped into four categories depending on whether they are related to science, equipment, operations, or resource allocation. Each of the four categories is discussed briefly in this chapter.

2.1 SCIENCE QUESTIONS

Typical science questions involve the effects on program costs or equipment evolutions of any changes in size and scope of scientific effort. The level of scientific activity, the emphasis on particular science areas, and the degree of experiment mechanization are typical topics of interest.

To answer such questions, planners derive alternative scientific programs that exhibit variations of the characteristic under study. One or more equipment evolutions are then matched to the missions obtained from these alternative scientic programs. The resulting sets of mission-mode combinations are then costed as separate candidate lunar programs. The answer to the question comes from the study of these candidate lunar programs.

2.2 EQUIPMENT QUESTIONS

Typical equipment questions concern the effects of introducing alternative equipment concepts for lunar exploration. Topics of interest include comparison of equipment

performance and efficiency, the advantages and disadvantages of a particular concept, and the effect of delaying equipment development.

Equipment questions are usually answered by the generation of several candidate lunar programs. These programs are then compared for cost and scientific achievement. Chapter 8 discusses techniques for efficiently generating such programs.

2.3 OPERATIONAL QUESTIONS

Typical operational questions involve the advisability of performing various lunar operations or the evaluation of alternative operation procedures. Sample topics from which questions could be formulated include different traversing techniques (flying or roving, long traverses or short traverses, caching or resupply), alternative techniques for hardware and crew delivery, and possible methods to reduce nonscientific time. In each case, the answers are found by deriving a sufficient number of lunar programs to exhibit the characteristics and potentialities of the various proposed operations.

The present MIMOSA planning technique was not derived to evaluate alternative approaches to details in the plan of a particular mission. For example, a question as to whether a particular experiment should be performed on the second or third day of a specific mission is beyond the scope of the MIMOSA analysis procedure. Only the influence of operations on overall program parameters (such as cost, schedule, or total scientific achievement) can be evaluated.

2.4 RESOURCE-ALLOCATION QUESTION

The allocation of resources to a lunar program is of fundamental importance in program planning. Typical questions involve total funding limits, rate of hardware procurement, launch rates, and schedule changes.

For the most part, questions on resource restraint variations are handled by separate cost calculation runs using the MIMOSA computer program. As shown in Chapter 10 of this handbook, extensive variations in input cost and schedule data for the computer are possible.

۰.

Chapter 3 SCIENTIFIC-PROGRAM FORMULATION

A scientific program is an ordered set of experiments and loci of investigation that are selected to satisfy specific scientific goals. Planners may select an existing scientific program* or formulate one of their own. The only stipulation is that program requirements be specified in the standard MIMOSA format. To formulate a new program, planners may utilize the logic developed within the MIMOSA methodology; the following steps are required to formulate a program:

- Identify basic goals.
- Identify major science disciplines to pursue the basic goals.
- Construct a matrix of scientific investigations relevent to each basic goal and scientific discipline.
- Select scientific experiments appropriate to each investigation.
- Specify major scientific equipment.
- Delineate the lunar surface exploration pattern.
- Distribute experiments in accordance with the exploration pattern.
- Complete a scientific-program specification.

A brief discussion of each of the foregoing steps is presented in this chapter.

3.1 IDENTIFICATION OF SCIENTIFIC GOALS

The first step in formulating a scientific program is to identify the goals to be achieved. Planners may have their own specific objectives, or they may rely on the desires of the scientific community.

^{*}Seven typical scientific programs representing widely different levels of scientific activity are provided as part of the MIMOSA tool. They are documented in MIMOSA Planning Methodology, Vol. III.

Basic goals for lunar exploration can be expressed in the form of a set of 15 questions promulgated by the Space Science Board of the National Academy of Sciences during their Summer 1965 Conference at Woods Hole, Massachusetts. These questions are confined to the Moon as an object of inquiry. In the MIMOSA contraced study, ten additional questions were formulated. These questions extend the scope of inquiry to include use of the Moon as a base for pursuit of extralunar goals such as planetary astronomy, stellar astronomy, and space physics. They are drawn primarily from the Space Science Board proceedings related to aspects of the national space program other than lunar exploration. Table 3-1 is a complete list of the 25 questions.

3.2 IDENTIFICATION OF MAJOR DISCIPLINES

The second step in formulating scientific programs is to identify the science disciplines involved in pursuit of the basic goals. Working groups at the NASA Conference on Lunar Exploration and Science at Falmouth, Massachusetts, following the 1965 Woods Hole Conference, established eight major scientific disciplines concerned with exploring the Moon. Later, two additional disciplines, providing for engineering assessment of the lunar environment and for exploitation of the Moon for purposes of applied science, were recognized. The ten suggested major disciplines are as follows:

- Lunar Atmosphere Studies
- Geodesy/Cartography
- Geology
- Geochemistry
- Geophysics
- Particles and Fields
- Bioscience
- Astronomy
- Mission Support
- Applied Science

Table 3-1

BASIC SCIENTIFIC QUESTIONS

Question No.*	Question
1	Is the internal structure of the Moon radially symmetrical like the Earth, and if so, is it differentiated? Specifically, does it have a core, and does it have a crust?
2	What is the geometric shape of the Moon? How does the shape depart from fluid equilibrium? Is there a fundamental difference in morphology and history between the sub-Earth and averted faces of the Moon?
3	What is the present internal energy regime of the Moon? Specifically, what is the present heat flow at the lunar surface, and what are the sources of this heat? Is the Moon seismically active, and is there active volcanism? Does the Moon have an internally produced magnetic field?
4	What is the average composition of the rocks at the surface of the Moon, and how does the composition vary from place to place? Are volcanic rocks present on the surface of the Moon?
5	What are the principal processes responsible for the present relief of the lunar surface?
6	What is the present tectonic pattern on the Moon and distribution of tectonic activity?
7	What are the dominant processes of erosion, transport, and deposition of material on the lunar surface?
8	What volatile substances are present on or near the surface of the Moon or in a transitory lunar atmosphere?
9	Is there evidence for organic or proto-organic materials on or near the lunar surface? Are living organ- isms present beneath the surface?
10	What is the age of the Moon? What is the range of the age of stratigraphic units on the lunar surface, and what is the age of the oldest exposed material? Is a primordial surface exposed?
11	What is the history of dynamical interaction between the Earth and the Moon?
12	What is the thermal history of the Moon? What has been the distribution of tectonic and possible volcanic activity in time?
13	What has been the flux of solid objects stricking the lunar surface in the past, and how has it varied with time?
14	What has been the flux of cosmic radiation and high-energy solar radiation over the history of the Moon?
15	What past magnetic fields may be recorded in the rocks at the Moon's surface?
16	What are the long-term effects of reduced gravity on various life forms. including man?
17	What lunar resources are available for exploitation?
18	What lunar environmental factors are most significant to the design of proposed lunar exploration equipment?
19	Are the basic postulates of general relativity valid?
20	What is the total inventory of stars and interstellar matter in a representative volume of our galaxy?
21	What processes account for the phenomena observed in the Sun. e.g sunspots. flares, plages, faculae, and prominences?
22	What are the structures, composition, and energy regimes of planetary atmospheres other than that of Earth?
23	What are the structures and compositions of comets?
24	What are the precise locations of discrete x-ray sources, and what is the distribution of faint x-ray sources?
25	What is the distribution of very-long wavelength radio stars?

*Questions 1 to 15, inclusive, are taken directly from proceedings of the 1965 Woods Hole conference and are concerned with the Moon per se. The remaining questions were stated or implied at the Woods Hole and Falmouth Conferences and are motivated by the potentialities of the Moon as a base for pursuit of astronomical and other extralunar scientific goals.

3.3 SCIENTIFIC-INVESTIGATION MATRIX

The third step in formulating scientific programs is to derive a scientific investigation matrix of goals versus disciplines. The 25 questions (Table 3-1) and the 10 major disciplines represent the coordinates of such a two-dimensional matrix. Each combination of a basic scientific question with a major discipline represents an element of the matrix.

Within each matrix element, a group of investigations is then identified that is relevant to the basic question and the major discipline forming the element. The total array of investigations identified in all of the matrix elements is the scientific-investigation matrix. The scientific-investigation matrix provides a generalized outline of the contents of a broad spectrum of candidate scientific programs. Table 3-2 is an example of a portion of a scientific-investigation matrix. Not all of the major disciplines contribute to every basic question. For example, with reference to question 1, there are no entries in Table 3-2 under Lunar Atmosphere, Bioscience, Mission Support, or Applied Science.

3.4 SELECTION OF SCIENTIFIC EXPERIMENTS

The fourth step in scientific program formulation is to select one or more experiments to accomplish each investigation in the scientific-investigation matrix. For the purposes of the MIMOSA analysis procedure, the principal source of experiment definitions is the catalog of lunar scientific experiments compiled by North American Aviation, Inc.* The NAA catalog comprises about 350 distinct experiments classified in accordance with the 10 major scientific disciplines listed in paragraph 3.2. Classification of the experiments in this manner speeds selection of experiments appropriate to the investigations in the scientific-investigation matrix.

Experiment selection is not a completely automatic operation. Scientific-program planners must apply selective judgement in choosing the appropriate experiments

^{*}North American Aviation, Inc., "Scientific Mission Support for Extended Lunar Exploration," NAA SID Report 66-957, NASA Contract NAS-8-20258 (in preparation).

Table 3-2

ELEMENTS OF AN INVESTIGATION MATRIX FOR A TYPICAL SPACE SCIENCE BOARD QUESTION

	Question 1
Discipline	
	Is the internal structure of the Moon radially symmetrical like the Earth, and
	if so, is it differentiated? Does it have a core and a Crust?
Lunar	
Atmosphere	
Geodesv/	(a) Establish a selenocentric coordinate system.
Cartography	(b) Establish a topographic map of the lunar surface.
	(c) Determine a reference figure.
	(d) Determine the gross elastic parameters of the Moon.
	(e) Determine the mass and moment of inertia.
	(f) Determine degree of isostatic compensation.
	(g) Determine symmetry of density distribution.
Geology	(a) Determine stratigraphic sequence of surface deposits and structure of exposed rocks.
	(b) Establish stratigraphic succession in lunar craters and rilles.
	(c) Determine petrologic variations in oldest rocks exposed on highlands and in maria
	basement rocks.
	(d) Determine composition of volcanic ejecta and plutonic xenoliths.
Geochemistry	(a) Determine the state of compositional, structural, and age differentiation of the
	lunar crust.
	(b) Evaluate with respect to genetic environment outcropping or shallowly buried
	xenoliths formed at depths beyond reach of the drill.
	(c) Determine and evaluate regional patterns of radiometric rock ages.
	(d) Obtain data on composition of exposed basement rocks.
Geophysics	(a) Determine the near surface structure (0-5 km) of selected geologic terrains by
	(b) Determine the structure of selected geologic terrain at intermediate deaths (5-100 km)
	(c) Determine deep structure (100 km to the center of the Moon)
	(d) Determine the vertical deflection of gravity over the lunar surface.
	(e) Determine rock properties by in <u>situ</u> measurements in drill holes and from drill hole cores.
Particles and	(a) Obtain structure and variations in lunar magnetic field and solar wind to determine whether
Fields	the lunar magnetic field is intrinsic or due to solar wind accretion.
Bioscience	
Astronomy	(a) Determine lunar body motions with respect to inertial space.
Mission	
support	
Applied	
Science	

implied by their investigation matrix. For example, experiments concerning oceanology and meteorology appear in the NAA catalog. These experiments find no place in a matrix defined by the 25 Space Science Board questions. Moreover, a study of the NAA catalog reveals numerous redundant experiments. Hence, the quality of the experiment-selection process depends on the degree of understanding of the basic principles and objectives of the investigations to be performed. Table 3-3 presents examples of experiments pertinent to a few of the scientific-investigation matrix elements in Table 3-2.

3.5 SPECIFICATION OF MAJOR SCIENTIFIC EQUIPMENT

The next step is to identify those items of scientific equipment that are sufficiently large to require special launch packaging considerations. These items are called major scientific equipment. Table 3-4 lists the major scientific equipment presently contained in the MIMOSA Equipment Data Book. Each new item of major scientific equipment must be assigned a MIMOSA data sheet number and treated in the same manner as other exploration equipment.

3.6 LUNAR-SURFACE EXPLORATION PATTERNS

Objects of scientific interest are distributed widely over the lunar surface. There are two approaches to the exploration of this wide distribution on lunar surface. The locale approach is to visit isolated points on the lunar surface with limited (e.g., 10- to 200-km radius) surface-traverse capability in the neighborhood of each landing site. The path approach connects points of interest on the lunar surface by extensive (e.g., up to 1,500 km) paths along which exploration traverses are performed. Exploration patterns are a combination of these two approaches. Figure 3-1 shows a typical surface exploration pattern for a MIMOSA scientific program.

The pattern that best meets exploration requirements should be derived. Later mission-mode matching will result in some modifications to the pattern. Locations of scientific activity, path lengths, and experiments required at all specific locations are then specified.

Table 3-3

.

SELECTED EXPERIMENTS

	Question 1	
Discipline	Is the internal structure of the Moon radically symmetrical like the Earth, and if so, is it differentiated? Does it have a core and a crust?	
Geodesy/ Cartography	 (a) Establish a selenodetic coordinate system: 12020606* - Selenodetic mapping with lunar orbiter 90050201 - Earth/Moon laser corner reflectors (b) Establish a topographic map of the lunar surface: 11010101 - Selenodetic astronomy on the lunar surface 12010202 - Selenodetic surveying on the lunar surface 40060565 - High-resolution radar 	
	(c) Determine a lunar reference figure etc.	
Geology	 (a) Determine the stratigraphic sequence of surface deposits: 21040101 - Geologic mapping with unmanned orbiters 21040102 - Photogeologic mapping with manned orbiters 21010203 - Fine structure mapping etc. 	

*Each experiment in the NAA catalog is identified by an eight-digit number.



Table 3-4

Data Sheet No.	Equipment	Mass (kg)
3213-01	300-meter core drill	9,243
3242-02	1-meter optical telescope	1,300
3242-03	1.3-meter optical telescope	5,025
3242-04	2-meter optical telescope	13,950
3242-05	Solar Observatory	35,410
3243-01	Stellar observatory	2 3, 750
3231-01	X-ray telescope	1,620
3224-01	Mills Cross radio telescope	12,528
3223-01	Radio telescope dishes	22,250
3132-02	Geochemical laboratory	1,130

MAJOR SCIENTIFIC EQUIPMENT

3.7 DISTRIBUTION OF SCIENTIFIC EXPERIMENTS

The next step in scientific-program formulation is to distribute the scientific experiments making up the investigation matrix over the selected exploration pattern. To speed this effort and to greatly reduce the labor required later for mission definition, many of the scientific experiments can be assembled into groups called experiment packages. Each package is characterized by a commonality of experiment requirements.

The variety of possible experiment packages reflects the variety of situations that govern the distribution of experiments. Some experiment packages are designed for assignment to areas of suspected volcanic activity. Other packages may be composed of experiments requiring a standard number of repetitions along traverses. Still another package may comprise a group of experiments associated with logging a drill hole. There are a few experiments with chronological requirements, such as particle and field measurements to be correlated with the 11-yr cycle of solar activity. These may also be grouped into packages. When the experiments and experiment packages with special requirements have been distributed to the appropriate locales and paths, there remains a number of experiments that can be performed with equal validity almost anywhere on the Moon. These experiments serve as "fillers" to round out the scope and the scientific "balance" of missions and to use mission support capability inherent in the selected equipment mode that might otherwise be wasted.

3.8 SCIENTIFIC PROGRAM SPECIFICATION

The final step is to prepare a scientific-program specification in standard MIMOSA format that comprises all of the material defining the scientific program in the form required by mission planners for mission-mode matching. The scientific-program specification consists of the following:

- One or more lunar maps representing the lunar-surface exploration pattern
- A list of locales and paths in the preferred order of accomplishment
- A list of required major scientific equipment
- A tabulation of experiments and experiment packages with instructions as to their distribution relative to the lunar-surface exploration pattern, and a specification of the required number of experiment replications in each instance of use
- Any explanatory notes concerning special requirements of experiments, such as deployment of explosives, use of photosurvey orbiters, and alternative distributions of experiments.

A typical page from a scientific-program specification is given in Table 3-5 as an illustration of the kind of material submitted by the scientific-program planner to the scientific-mission planner.

When one scientific program has been specified fully, other scientific programs can usually be generated merely by varying the magnitude of the following key factors:

• The number of lunar locales to be visited and the extent of surface paths to be traversed

Table 3-5

"FILLER" EXPERIMENTS FOR LOCALES*

.

Experiment ID No.	Name	Locales Required	Replic. per Locale	Approx. Manhours per Locale	Mass (kg)	Comments
68040102	Evidence of Existing Life	2	1	10.0	57	One locale of each type
80010302	02 Engineering Properties of Lunar Surface		1	21.0	2	
80010303	03 Engineering Properties of Lunar Subsurface		1	10.0	6	
80010305	Lunar Soil Shielding	5	1	23.0	10	
80010321	Electrical Parameters, Surface and Sub- surface	2	1	24.0	35	One Type I and one Type II
80010322	Lunar Strata E-M Properties	3	1	15.0	3	
80020101	Surface Dust Environment	3	1	4.0	2	One Type I and one Type II
80020302	Lunar RF Noise	2	1	11.0	70	
80020501	Lunar Atmospheric Contamination – Rocket	2	1	1.0	5	First locale and before sixth locale
80110101	Gas Requirements for Drill	1	1	0.1	1	Before drilling
* 80110401	Geological Chemical Sampling	3	2	8.0	20	One setup
80120301	Explosive Energy Coupling	2	1	16.0	10	Before crust-upper mantle surveys
80130101	Biocontamination of Lunar Soil	2	1	2.0	15	
80300201	Genetic Effects – Animals	1	1	53.0	100	Mass override required
80320301	RF Ground-Wave Propagation	3	1	57.0	8	
80320401	RF Subsurface Propagation	2	1	45.0	10	
80340101	Dust Removal Technique	2	1	2.0	10	
80360201	Dangerous Terrain Warning	4	2	3.0	20	Two Type I and two Type II
80370102	Visual Techniques – Land Navigation	2	1	6.0	1	
90050201	Laser-Beam Corner Reflectors	3	1	0.5	30	Place at locales 18, 5, and 3
999999909	Early ESS	4	1	22.0	150	
99999917	Particle Experiment II	1	1	5.0	29	1971-73 or 1977-79
99999920	Particle Experiment V	1	1	5.0	134	1971-73 or 1977-79

*These experiments are to be done at the Program C locales as early as possible. They can be done at either type locale except as noted.

- The number and size of major scientific equipment
- The number of experiments to be performed for each investigation in the scientific-investigation matrix
- The number of replications of each experiment at each locale and along each path

. .

Chapter 4

EXPLORATION EQUIPMENT AND EQUIPMENT-EVOLUTION DEFINITION

This chapter defines lunar exploration equipment, describes the equipment data required for MIMOSA exploration program analysis, and discusses how equipment concepts are linked together in an orderly evolution.

4.1 EQUIPMENT DESCRIPTION

Exploration equipment includes all major pieces of hardware used to conduct a lunar program. Major categories of exploration equipment are transportation systems (both launch vehicles and flight systems) and mission equipment (lunar orbiters, surface equipment, and scientific equipment).

Table 4-1 lists the various types of exploration equipment that are presently defined in the MIMOSA Exploration Equipment Data Book (MIMOSA Planning Methodology – Vol. II). Each type of equipment is subdivided into groups that embody those concepts with similar performance or design characteristics. The group identification number plays a key role in the MIMOSA computer program. Many of the group identification numbers trigger special computer calculation subroutines.

4.1.1 Transportation Systems

For purposes of MIMOSA, launch vehicles and spacecraft are called "flight systems." Generally, over one-half the equipment on any launch consists of flight systems. It has been found useful to consider all flight systems on a launch as a single entity. This entity is called a "transportation system."

Transportation systems play a vital role in the MIMOSA analysis. They are the means by which the individual launches of a mode can be identified and the performance capability of the launches can be described. As Table 4-1 indicates,

4-1

LOCKHEED MISSILES & SPACE COMPANY

Table 4-1

EXPLORATION EQUIPMENT

Equipment Type	Data Book Group ID No.				
Transportation Systems					
Personnel Transport	1111 to 1124				
Logistic Transport	1132 to 1148				
Orbiter Transport	1151 to 1152				
Probe Transport	1161				
Test Launch Transport	1171 to 1172				
Launch Vehicles	1211 to 1271				
Spacecraft	1311 to 1454				
Mission Equipment					
Lunar Probes (Unmanned)	2111 to 2132				
Lunar Orbiters	2211 to 2222				
Personnel Shelters	2321 to 2327				
Lunar Roving Vehicles (LRV)	2411 to 2423				
Roving Vehicle Trailers	2431 to 2433				
Lunar Flying Vehicles (LFV)	2512 to 2522				
Lunar Surface Support	2711 to 2722				
Major Scientific Equipment	3111 to 3243				

LOCKHEED MISSILES & SPACE COMPANY

the MIMOSA Data Book includes five types of transportation systems: personnel, logistics, orbiter, probe, and test.

4.1.2 Mission Equipment

The major items of mission support hardware used on or near the lunar surface are called "mission equipment." Mission equipment includes probes, orbiters, surface personnel shelters, roving vehicles (LRV), trailers for roving vehicles, flying-vehicles (LFV), resupply systems for expendables and for LFV propellant, fuel-regeneration systems, nuclear power supplies, and "major" scientific equipment. A piece of scientific equipment is called "major" if it exerts a pronounced influence on mission-support capability of a mode or if it requires peculiar packaging considerations.

Data on mission equipment are used directly by the MIMOSA computer program in comparing mode capabilities with mission requirements. As a result, there are special data input requirements for nearly every type of mission equipment. These are summarized in the following paragraphs.

4.2 COMPUTER INPUT PREPARATION FOR EXPLORATION EQUIPMENT

The detailed MIMOSA analysis, which utilizes a special computer program, requires specific information about each type of equipment. This information falls into the general categories of mission-support performance and resource planning (schedules and costs). All systems must have a suitable name and identification number for input to the computer. The remaining information for each type of equipment varies. This is because each type requires different parameters to describe its performance. Table 4-2 lists 33 items of required data. They are as follows:

4.2.1 Transportation Systems

 <u>Payload delivery capability</u>. Total delivered payload to the lunar surface (or orbit) in kilograms. This is not input directly to the computer, but is
Table 4-2

EXPLORATION EQUIPMENT COMPUTER INPUTS

		Required Informati	on	
System		Performance	Re	source Planning (All Systems)
Transportation System	(1)	Payload delivery capability (kg) Return to Earth payload (kg)	(26)	Nonrecurring system cost (\$M)
All Mission	(3)	Crew levels (No. of men)	(27)	Nonrecurring facili- ties cost (\$M)
Equipment	(4)	Operations expendables (kg/day)(a)	(28) .	Recurring hardware
	(5)	Experiment power expendable rate (kg/kw-hr)	(29)	Recurring operations
	(6)	Spacesuit expendable rate (kg/mhr) ^(b)		support cost (\$M)
	(7)	Unusable expendables (kg) ^(a)	(30)	Procurement cost reduction with
	(8)	Activate and deactivate time		quantity (percent of
		- Shirtsleeve (mhr) ^(a)		first cost)
		- Spacesuit (mhr) ^(a)	(31)	Research and develop- ment flight test sched-
	(9)	Non-scientific operations		ule (6-mo increments)
		- Shirtsleeve (mhr/day) ^(a) - Spacesuit (mhr/day) ^(a)	(32)	Subsystem SOTA advances:
Shelter Only	(10)	Maximum staytime (days) ^(a)		- Nonrecurring costs
LRV/Trailer	(11)	Mobility expendable rate (kg/km) ^(a)		- Required delivery
Only	(12)	Expendables per stop (kg) ^(a)		date before system development (6-mo
	(13)	Time per stop		increments)
		– Shirtsleeve (mhr) ^(a) – Spacesuit (mhr) ^(a)	(33)	Spares procurement rate
	(14)	Nom. single-excursion time (days) ^(a)		
	(15)	Nom. single-excursion range (km) ^(a)		
	(16)	Average speed (km/hr) ^(a)		
	(17)	Max. payload-carrying capability (kg) ^(a)		
	(18)	Driving in shirtsleeves or spacesuit ^(a)		
LFV Only	(19)	Time and expendables per stop(a)		
	(20)	Max. excursion time (hrs) ^(a)		
	(21)	Range vs. number of stops (km)		
	(22)	Specific impulse (m/s)		
	(23)	Flying vehicle inert mass (kg) ^(a)		
	(24)	Flying in shirtsleeves or spacesuit ^(a)		
Major Scientific Equipment Only	(25)	Identification of related experiment		

(a) At each crew level in item (3).(b) In excess of shirtsleeve expendable usage.

used to calculate the maximum mass permitted for scientific equipment by a mode (paragraph 6.1).

(2) <u>Return to Earth payload</u>. Maximum payload, in excess of the flight crew, that can be returned to Earth. This is not input directly, but is used to calculate the total sample return capability of each mode (paragraph 6.1).

4.2.2 All Mission Equipment

- (3) <u>Crew levels</u>. Various crew levels (i.e., number of men) for which the system is intended to be compatible. For LRV Trailers, the crew levels associated with the companion LRV are used.
- (4) <u>Operations expendables</u>. Total expendables in kilograms/day used at each of the crew levels in foregoing item (3) to maintain and operate the system without driving or providing power for scientific experiments. It includes power supply expendables, life support and environmental control expendables, water, and atmospheric losses. All expendables are counted (whether dumped overboard or not) except those that are required for spacesuit operation.
- (5) <u>Experiment power expendable rate</u>. Expendable rate, in kilogram/kilowatthour used to provide power for experiments supported by the system.
- (6) <u>Spacesuit expendable rate</u>. Additional expendables in kilograms/manhour, required for spacesuit operations over and above the normal shirtsleeve requirements for oxygen usage rate, water for cooling, replacement batteries, and rate of usage of lithium hydroxide cartridges. If the normal operation of the system is in spacesuit, the total spacesuit expendable rate is used.
- (7) <u>Unusable expendables</u>. Total expendables in kilograms that are supplied with the system but are not available for use. Included are boiloff (both before and during the mission), ullage, refueling losses, and reserves.
- <u>Activate and deactivate time.</u> Total manhours for both activation and deactivation of the system for each crew level in foregoing item (3).

The manhours for both spacesuit and shirtsleeve operations are specified. Deployment times, which cover offloading and preliminary work before the system is ready for setting up and activation, are not included. Deployment times are specified at the mode level (see paragraph 6.1).

(9) <u>Nonscientific operations</u>. Total manhours per day for system maintenance, repairs, stationkeeping, airlock operations, and time required by the crew for resting, eating, and personal hygiene. The time is specified for both spacesuit and shirtsleeve requirements.

4.2.3 Shelter Only

(10) <u>Maximum staytime</u>. Maximum number of days that a shelter can support each crew level specified in item (3) while depleting its load of expendables as described in items (4), (5), (6), and (7).

4.2.4 LRV/Trailer Only

- (11) <u>Mobility expendable rate.</u> Expendables in kilograms/kilometer that are required to power the mobility subsystem of an LRV or LRV plus trailer. Expendables for startup or for powering other subsystems, such as environmental control or astrionics, are not included. These expendables are included in item (4).
- (12) Expendables per stop. Total expendables in kilograms/cycle that are used by a roving vehicle for activation and deactivation at each local stop. Expendables for the crew, and for normal operations occurring during this period, are not included. These are covered by item (4).
- (13) <u>Time per stop.</u> Spacesuit and shirtsleeve manhours required for the activation and deactivation of a roving vehicle at each local stop for the crew levels given in item (3).
- (14) <u>Nominal single-excursion time</u>. Nominal time, in days, that a roving vehicle with a given load of expendables on board can support for each crew level specified in item (3).

- (15) <u>Nominal single-excursion range</u>. Nominal range in kilometers that a roving vehicle, with a rated load of expendables on board, can travel (one way) at its most efficient expendables consumption rate. Separate values must be given for each crew level specified in item (3). Normally, the nominal range is given in conjunction with nominal excursion time.
- (16) <u>Average speed</u>. Average roving vehicle speed in kilometers/hour. The input value is used by the computer to determine driving time.
- (17) <u>Maximum payload-carrying ability</u>. Total cargo and minor scientific equipment in kilograms that can be carried by the vehicle. The crew and its equipment are not included.
- (18) <u>Driving in shirtsleeve or spacesuit</u>. Specification as to whether normal operations during travel on the vehicle require spacesuit or not.

4.2.5 LFV Only

- (19) <u>Time and expendables per stop</u>. Same as items (12) and (13) as applied to flying vehicles.
- (20) <u>Maximum single excursion time</u>. Duration in hours that a given crew level can operate a flying vehicle without expendables resupply from another system. The expendables do not include propellant.
- (21) <u>Range vs. number of stops.</u> Tradeoff of kilometer range vs. number of stops (i.e., landings and takeoffs) for a single traverse by LFV.
- (22) <u>Specific impulse</u>. LFV propellant capability in terms of specific impulse in meters/second.
- (23) <u>Flying vehicle inert mass.</u> Mass of an LFV concept in kilograms derived from the sum of the total nonexpendables in the concept design plus the unusable LFV propellant. This does not include the crew members, their spacesuits and backpacks, usable LFV propellant, and all other expendables.
- (24) <u>Flying in shirtsleeves or spacesuit</u>. Specification as to whether all operations on the LFV require spacesuits or not.

4.2.6 Major Scientific Equipment Only

(25) <u>Identification of related experiment</u>. List of all the experiments that are performed either partly or wholly by a given item of major scientific equipment. Eight-digit North American Aviation study experiment identifiers* are used.

4.2.7 All Systems

- (26) <u>Nonrecurring system cost.</u> Nonrecurring funding in millions of dollars versus 6-mo intervals required for system R&D. The delivery date of the first flight-qualified system to the payload integrator is considered as the end of the R&D phase.
- (27) <u>Nonrecurring facilities cost</u>. Nonrecurring funding in millions of dollars versus 6-mo intervals for facilities construction during the system R&D phase.
- (28) <u>Recurring hardware procurement cost</u>. Required funding in millions of dollars versus 6-mo intervals for recurring hardware procurement costs of the first production unit. The required time interval runs from manufacturing go-ahead to delivery of the first unit to a payload integrator.
- (29) <u>Recurring operations support cost.</u> Required funding in millions of dollars versus 6-mo intervals for recurring operations support of the first production unit. Each 6-mo interval is measured from the delivery date of the first unit to a payload integrator.
- (30) Procurement cost reduction with quantity. Factors for determining system cost reduction associated with the "learning curve." The curve is generated by grouping the expected total number of articles to be procured into no more than ten groups and assigning to each group a decreasing percentage of first-article cost.

^{*}North American Aviation, Inc., "Scientific Mission Support for Extended Lunar Exploration," NAA SID Report 66-957, NASA Contract NAS-8-20258 (in preparation)

- (31) <u>Research and development flight test schedule</u>. The number of R&D test flights versus 6-mo intervals measured from the delivery data of the first flight-qualified system to the payload integrator. The launch vehicle used for each test flight is also identified.
- (32) Subsystem state-of-the-art (SOTA) advances. Subsystem nonrecurring R&D costs, as well as cost of subsystem facilities, by 6-mo time intervals from start of funding to delivery date of the first flight-qualified subsystem to the system integrator. These costs are figures exactly as in items (26) and (27) except at a subsystem level.
- (33) Spares procurement rate. Rate at which costs for equipment spares or backups are accumulated expressed as the number of mission articles procured before a backup is procured. One backup is automatically costed when an equipment item is used the first time.

4.3 EQUIPMENT-EVOLUTION REQUIREMENTS

Equipment evolution, as defined in the MIMOSA study, means growth of missionsupport capability from some low-performance capability level (such as Apollo) to some higher level (such as 6- to 12-man permanent bases). An equipment evolution should meet six requirements other than growth from small to large performance. These are as follows:

- <u>Endpoint Capability</u>. An equipment evolution must have some final attainment. This point is usually the first selected feature of an equipment evolution. It is expressed in terms of some level of transportation system uprating or some level of crew support. Having decided upon an endpoint capability, it is then possible to analyze alternate paths to that capability.
- <u>Compatibility</u>. At any point in equipment evolution, systems should be compatible with regard to crew support level, staytime, landed mass capability, sample return capability, dormant lunar storage time, and Earth support requirement. It makes little sense, for example, to transport low-performance surface mission equipment with high-performance transportation equipment.

- <u>Irreversibility</u>. An equipment evolution should not regress. Once a level of attainment is achieved, it must not be discarded to make room for less capable systems. To do so would raise doubts as to the necessity of the higher capability systems.
- <u>Maximum Usage of Developed Systems</u>. All systems in an equipment evolution should have maximum utility and growth potential.
- <u>Flexibility</u>. An equipment evolution, which does not take into account the possible effects of feedback from early missions or changes in national goals is extremely vulnerable. Each step in the evolution must not preclude changes in scope and level of effort or the exploitation of possible state-of-the-art advances.
- <u>Integration With Existing Plans</u>. Equipment evolutions should evolve smoothly from current planning. Evolutions formulated on an abrupt change of existing plans or operations may not be accepted readily.

4.4 SAMPLE EQUIPMENT EVOLUTIONS

Figure 4-1 shows sample equipment evolutions. It illustrates four different ways to evolve from Apollo to a large mission support endpoint capability. It steps first from Apollo to S/AA. The S/AA involves two-man, 2-wk missions with limited mobility and no Saturn V uprating. The second step is to a modest (approximately 125%) uprating of Saturn V to the level where three-man crews can be delivered and supported on the lunar surface for up to 180 days. The third step (the endpoint capability) involves substantial uprating of Saturn V to deliver and return six men with direct launches and to support those men for periods up to 2 yr.

Equipment Evolution 2 goes directly to the endpoint capability from S/AA. Evolution 3 bypasses both intermediate steps of Evolution 1 and jumps directly from Apollo to the endpoint capability. Evolution 4 bypasses the first step of Evolution 1 and goes directly from Apollo to the 125% Saturn V.



Chapter 5 MISSION FORMULATION

A mission is defined as a set of experiments performed at a definable location on or about the Moon and accomplished in a finite length of time. A mission constitutes a portion of a total scientific program. The mission may require manned effort on the lunar surface or in lunar orbit, or it may require no manned effort at all; all activity may be clustered at one point, or it may be spread over extensive traverses. This chapter presents the process of mission sizing and traverse formulation, describes a sample mission, and discusses the computer input for missions.

5.1 MISSION SIZING

The crux of mission planning is sizing the mission. Some missions are immediately delineated by virtue of their isolated location. Others are defined uniquely because of some specific required activity such as optical or radio astronomy. Whatever its major characteristics, a mission must eventually be accomplished by some grouping of mission equipment. Mission planners must have intimate knowledge of the mission equipment, and must make educated guesses as to how much scientific mass and manhours can be assigned to any specific step in equipment evolution in a given location at a given time.

To size missions, the following instructions should be observed. The entire scientific program is laid out (preferably on a map) to show exactly where each experiment is to be performed and how much scientific equipment mass is associated with each experiment. This layout is scanned to find for likely personnel and logistics landing sites. The most obvious sites are points where experiments are concentrated or where large individual mass or labor is required. These sites are then used as "centroids" of experiment arrays. The mass and manhours of experiments are summed within the

arrays and are compared with the capabilities of the mission equipment in the desired equipment evolution. If a vast disparity exists (i.e., a large overrun or underrun in one or more parameters) the number of mission experiments are trimmed or added to, whichever is appropriate. Numerous trades can be made at this point since many experiments can be performed at several sites and may be moved from one mission to another. "Filler" experiments (paragraph 3.7) can be added or removed.

5.2 TRAVERSE FORMULATION

A traverse is a scientific excursion on or over the lunar surface for the purpose of obtaining measurements, observing scientific phenomena, or accomplishing other mission-related activities. The traversing equipment has a scientific-manhour capability as well as a cargo-carrying capability. These parameters, combined with the excursion length, generally govern the number and types of traverses.

For traverse formulation, a vehicle must be identified. To define the traverse, the scientific map of the mission must be examined. Starting at the most distant experiment location on one side of the personnel landing site for the mission and working toward the landing site, the minor scientific mass of all the experiments must be summed. This process stops when the sum reaches 80 percent of the cargo-caryying capability of the traversing vehicle or when the landing site is reached. This defines a tentative traverse. If the sum of cargo-mass requirement reaches the vehicle limit before the landing site or a significant supply point is reached, then the remaining experiments should be performed on another traverse. If the sum of the cargo-mass requirement does not achieve the vehicle limit before the landing site or supply point is reached, then the possibility arises for enlarging the scope of the traverse by incorporating other traverse of the mission.

The range of a traverse is the distance traveled from a starting point (usually the landing site or supply point) to an ending point, (usually the same landing site or another supply point). With this range determined, the traverse staytime may be estimated from the range-staytime tradeoff curves found in the MIMOSA Equipment Data Book

for the traversing vehicle. This staytime must be multiplied by the vehicle crew level to obtain the gross manhour availability of the equipment. Then this availability is multiplied by some factor (20 percent has proved useful with the current estimates of man's capabilities in a spacesuit and reflects a realistic estimate of nonscientific time see Table 7-1 in Chapter 7) to estimate the scientific manhour capability of the traversing vehicle. The scientific manhour capability is then compared with the sum of the experiment scientific manhours. If the scientific manhour requirement is less than the vehicle capability, then the tentative traverse may be submitted to the computer. If the scientific manhour requirement exceeds the vehicle capability, then the list of experiments must be adjusted either by cutting the number of experiments or by cuting replications of experiments.

5.3 SAMPLE MISSION

Figure 5-1 shows a typical mission used during the MIMOSA study.* It is mission 28 of 32 required to accomplish a large scientific program. This scientific program requires traversing 20 paths on the Moon. The path shown in the inset of Fig. 5-1 involves explosives emplacement, hole drilling, general surveying, and mapping.

Actually, each particular branch of this path could have been a mission. For the branch labeled traverse R-01, the total scientific manhours required by the experiments is 601 (394 +207), and the total minor scientific equipment mass is 13,265 kg. During the early 1980's, the allowable step in the equipment evolution being analyzed has a capability of about 1,600 scientific manhours per mission and 46,650-kg reserves for minor scientific equipment. Thus, incorporation of more branches (traverse R-02 and traverse R-03) produces a better balance between mission requirements and equipment capability.

^{*}Apart from the number of experiments to be performed, the mission data given in tabular form on the figure are derived from the computer output. The degree of accuracy indicated is not required for the initial planning exercise.



LOCKHEED MISSILES & SPACE COMPANY

LMSC-A847943 Vol. I

Fig. 5-1 Example Mission

The vehicle expected to perform the traverses in the sample mission could carry an estimated 7,660 kg of cargo. If that value had been exceeded on any of the traverses, then some traverse rearrangement would be needed. If, for example, the cargo for traverse R-03 had been greater than 7,660 kg, perhaps a traverse could be added (R-04) that would go from the central point down to the 300-m drill site and back. Traverse R-03 cargo could then be reduced by whatever cargo was carried on R-04.

5.4 COMPUTER INPUT PREPARATION

After a mission has been defined to the planners' satisfaction, its pertinent characteristics data are input to the computer. Paragraph 10.5.1 of this handbook presents the detailed instruction for the following inputs.

Whether the mission is to be matched to a mode or not, all experiment and equipment data should be input to the computer for cost purposes. For example, two flying vehicles were used to implant remote geophones in traverse R-03 of the example mission. The capability of the unmanned flying vehicle was completely defined by the MIMOSA Data Book. In simple cases like these, there is no need for computer verification. In this mission, the probe experiments were grouped with R-03, and the flying vehicles were added to the cargo of the roving vehicle.

Data on a mission are input to the computer by preparing four types of cards. The first card type (No. 3-01) assigns a unique number to the mission and names the physical location on or about the Moon where the mission is to be performed. It is assigns a nominal mission start date and establishes any minimum time intervale that must elapse between that start date and the end of any earlier mission.

The second type of card (3-02) identifies each base experiment to be performed on the mission. Each card of this type specifies the number of times an experiment must be setup (Phase 1), operated (Phase 2), analyzed (Phase 3), and taken down (Phase 4). Input by experiment phase accommodates the many alternate ways in which different experiments can be performed. If the experiment uses major scientific equipment or

LMSC-A847943

shares equipment with other experiments on the mission, or if there is a desire to modify the given minor scientific experiment mass, then an opportunity is provided on this same card to override the experiment mass value used to calculate the launch mass for the mission. In a like fashion, the nonrecurring and recurring costs of the experiment may be overridden with new numbers if different scientific equipment is being used or if there is conflict with the experiment input cost data.

The next card type (3-03) identifies each traverse, the traverse range, the number of major stops and, if a flying traverse, the total velocity increment expended. Mission planners establish each of these items. They may wish to increase the straight line range by a suitable percentage (e.g. 10 percent) to account for any rough terrain. They determine the total velocity increment for a flying traverse by summing the range requirements for each separate flight of the traverse and obtaining the resulting velocity increment from the flying vehicle MIMOSA data sheet.

The last card type (3-04) identifies each experiment performed on a traverse. Replications of the experiment are given in the same manner (i.e., four phases: setup, operate, analysis, and take down) as with the 3-02 card. This card just as with the 3-02 card, provides an opportunity to override launch mass, nonrecurring cost, and recurring cost of the experiment. In addition, the value used by the computer to calculate the traversing vehicle cargo mass requirement can be overridden. The provision is used frequently as a means of achieving a realistic cargo total. Often it takes two or more traverses to transport the mass for one experiment to some specific location. This subdivision of cargo mass can be input to the computer by showing cargo overrides for that experiment on all the subject traverses. The cargo override can be used to account for the mass of a surface-to-surface probe that is used to emplace an experiment at a remote location. If an experiment is performed while on a traverse but the experiment mass is not actually delivered by the traversing vehicle, then the cargo mass for the experiment can be overridden to zero.

The computer takes all of this input data, stores it, and prints out what it has stored. Figure 5-2 shows two of 14 printed pages for the example mission 28. The figure

MISSION IDENTIFIER	A7 0128PATH 13.5,4,5 1	
NOMINAL START DATE Related Mission Number Min. No. 6 MO. Periods Betwei	1978A 1578A 16EN THE MISSIONS 0.	
		-
		• •
:		

LMSC-A847943 Vol. I

5-7

LOCKHEED MISSILES & SPACE COMPANY

TDAVEDEE THENTIETED					
TRAVERSE ADGE (KM) Traverse Adge (KM) Number of Stops	206	-01 -00 -00			
LEV VELOCITY INCHEMENT (METERS/SEC)		• 00			
<u> </u>	PHASE CODE FREQUENCY	1 A 1,00	2 A 3.00	З, 00	4 A 1•00
		PAYLOAD MASS (KG)	NON-REC COST (MILLIONS)	REC COST (MTLLTONS)	CARGO MASS (KG)
	CODE OVERRINE	0,0-	00*•	0U •-	н 00.
EXPERIMENT IDENTIFIER 9.0.05.02.01	PHASE CODE	44	~~~	Юd	4
	FREQUENCY	1.00	1.00	1.00	1.00
		PAYLOAD MASS (KG)	NON-REC COST (MILLIONS)	REC COST (MILLIONS)	CARGO MASS (KG)
	CODE OVERRINE	00*-	-•00	••00	-•00
EXPERIMENT IDENTIFIER 9.9.99.99.36	PHASE	1	2	3	
	CODE FREQUENCY	A 1.00	A 1.00	۸ 1.00	A 1.00
	and a constant of the second	PAYLOAD MASS (KG)	NON-REC COST (MILLIONS)	REC COST (MILLTONS)	CARGO MASS (KG)
	CODE OVERRIDE	00 •-	00	00	600
EXPERIMENT IDENTIFIER 2.1.01.04.12	PHASE	1	2	3	ŧ
	CODE FREQUENCY	1+00	A 1.00	A 1.00	A 1.00
		PAYLOAD MASS (xg)	NON-REC COST (MILLIONS)	REC COST (MILLIONS)	CARGO MASS (KG)
	CODE OVERRIDE	00.	• 00	00*-	в •00
EXPERIMENT IDENTIFIER 2.1.02.03.07	PHASE CODE	4 1	~~~	ñe	4
	FREQUENCY	1.00	1.00	1.00	1.00
		PAYLOAD MASS (KG)	MILLIONS)	REC COST (MILLIONS)	CARGO MASS (KG)
	CODE OVERRIDE	8 • 00	в • 00	8 • 00	8 • 00

/

LOCKHEED MISSILES & SPACE COMPANY

Fig. 5-2 (Cont.)

shows the first page of printout for the mission plus one of five pages of printout on traverse R-01. The first page is nearly blank because base experiments, which would be printed here, are not required on this mission.

١

Chapter 6 MODE FORMULATION

A mode is a physically compatible group of equipment and associated operations that can perform a mission. The concept of a mode permits identification of a complex set of equipment and operations with one identification number; this is useful for computer manipulation. The description of a mode takes into account those special operations and costs that cannot be accounted for by individual equipment items alone. The principal drawback to the mode concept is that, if two modes are identical except for one item of data, they still must be given different mode numbers and be treated as completely different entities.

This chapter describes the requirements for mode formulation, presents a sample mode, lists the required computer input for modes, and discusses the application of several types of special modes. These special types are "working" modes, "zero launch" modes, and "emergency" modes.

6.1 MODE REQUIREMENTS

A mode provides mission-support capability through (1) appropriate mission equipment, (2) sample return capability, (3) manhours for scientific operations, and (4) mass margin. Mass margin is the difference between mass delivery capability of the transportation system and mass of the mission equipment (which includes major science equipment) making up the mode.

To formulate a mode, the following steps are necessary:

 Assemble the transportation equipment, mission-support equipment (such as shelters and rovers), and major items of scientific equipment (such as laboratories and telescopes) that are required to perform a mission (leaving it up to the computer to provide verification of performance).

- (2) Identify the launch that will carry each piece of equipment.
- (3) Ensure that each launch is feasible (i.e., that mass delivery capability, payload envelope, or earth-support capability limits are not exceeded).
- (4) Calculate the mass margin and Earth-return mass capability of the mode. This is done by summing the mass margins and mass return capabilities of each launch in the mode.
- (5) Identify the mass and manhours for all special operations associated with the mode. These operations include deployment, refueling, resupply, and crew-rotation time (including on site training and orientation).
- (6) Identify mode-level nonrecurring and recurring costs for payload integration, mission control, tracking, data acquisition, and data analysis. (In the MIMOSA contract study, launch and crew recovery costs were assigned to the individual transportation systems and were not considered at the mode level.)

6.2 SAMPLE MODE

Figure 6-1 shows a typical MIMOSA mode data sheet. The mode shown (503-30114-01) was matched against the sample mission displayed in Fig. 5-1. There are four launches in the mode, three that carry mission equipment and one that delivers three personnel to the lunar surface. Each launch has been loaded with specific systems and with sufficient structure to support those systems during flight and to assist in their deployment at the landing site. All launches show a mass margin available for minor scientific equipment and mission-extension expendables.

6.3 COMPUTER INPUT PREPARATION

A mode is input to the computer by extracting certain information from the mode data sheet and other information from program ground rules, and then preparing nine types of data cards. Table 6-1 itemizes each input and shows where it is obtained. Paragraph 10.5.2 of this handbook presents the detailed instructions for filling out each type of card.

MIMOSA DA	ATA SHI	EET	No.	503-3011	•	-0	1	Pa	ige]	of
MOBILE BASE	- THREE	MAN						D Ju	ote n 67	Rev
GENERAL:	Person Logis Gurtua Total Maximu Sample Concus Numbes	nnel Del tics Del ce Créw Crew Le um Stayt P Return rrent Sh r of Pay	ivery Co ivery Co Level vol ime (Day i Capabil elters (load Lau	oncept oncept ity (Kg) Yes or No mches)		-	Dire Dire 9 9 5 N	ct 3 0 0 0 h	
OPERATIONS:	Time 1 Shirts Spaces	for Base sleeve M suit Man	Deploym Anhours hours fo	ent (Days for Specia or Special) al Ops Ops.	3.	-		6 2 3	
PAYLOAD LAUNCH IDENTIFICATION:	Mass N	Margin f	or Laune	$ \begin{array}{c} h \ \#1 \ \underline{14, 9} \\ \#2 \ \underline{16, 9} \\ \#3 \ \underline{15, 9} \\ \#^{h} \ \underline{1, 9} \end{array} $	588 132 520 400	- - 	#, #, #1 #8			
SYSTEMS DESCRIPTION:) + 11	Max	I	Delive	ry		Ē	xpenc	tub].e.
Launch # Syst	em I	Book #	Suppo	ا المراس	-rior	то . (с. м		r	or sp	v)
		<i>i i i</i>	Suppo	steu 1	Jaunen	I (O M	0.5.7	0	ps. (_ <u>Λ(',)</u>
<u>1</u> Tr <u>ansp</u> 300- m	ort I drill	147-01 321 3-02								
<u>1</u> Transp 30 <u>0-m</u> Probe	ort 1 drill	147-01 321 3-02 21 32-02				- 2 2		- - -		, ~(?,)
Transp 300-m Probe Probe	oort] drill	1147-01 321 3-02 21 32-02 21 32-02				2			ρο. ((((· · (?,) - · · · · · · · · · · · · · · · · · · ·
1 Transp 300-m Probe Probe 2 Transp Traile	ort]	147-01 321 3-02 21 32-02 21 32-02 21 32-02 132-02 147-01 24 33-04				- 2 2 - 2 - 2				
1 Transp 300-m Probe 2 Transp Traile LFV	ort] drill ort] r	1147-01 1213-02 132-02 132-02 132-02 132-02 132-02 132-02 132-02 132-02 132-02 132-02 132-02 132-02				2 2 2				· · · · · · · · · · · · · · · · · · ·
1 Transp 300 - m 300 - m Probe Probe 2 Transp Traile LFV 3 Transp	ort 1 drill cont 1 r 22 nort 1 nort 1 nort 1	1147-01 1147-01 1147-01 1147-01 1147-01 1147-01 1147-01				- 2 2 2 2 2 2 2				· · · · · · · · · · · · · · · · · · ·
1 Transp 300-m 300-m Probe Probe 2 Transp Traile LFV 3 Transp LRV	ort 1	$\frac{1117}{2213-02}$ $\frac{2132-02}{2132-02}$ $\frac{1132-02}{2132-02}$ $\frac{1147-01}{2433-04}$ $\frac{2521-02}{2521-02}$ $\frac{1147-01}{2423-01}$				2 2 2 2 2 2				
1 Transp 300 - m 300 - m Probe Probe 2 Transp Traile LFV 3 Transp LRV 4	ort 111	$\frac{1117-01}{3213-02}$ $\frac{1132-02}{2132-02}$ $\frac{1132-02}{2132-02}$ $\frac{1147-01}{2433-04}$ $\frac{1147-01}{2423-01}$ $\frac{1124-01}{124-01}$				- 2 2 2 2 2 2 2 2				· · · · · · · · · · · · · · · · · · ·
1 Transp 300 - m 300 - m Probe Probe 2 Transp 3 Transp 3 Transp LRV LRV	ort 1 ort 1 r 27 ort 1 r 27 ort 1 ort 1 or	$\frac{11147-01}{3213-02}$ $\frac{3213-02}{2132-02}$ $\frac{1147-01}{2433-04}$ $\frac{2433-04}{2521-02}$ $\frac{1147-01}{2423-01}$ $\frac{1124-01}{1124-01}$				- 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2				
1 Transp 300 - m 300 - m Probe Probe 2 Transp Traile LFV 3 Transp LRV 4	ort 111	$\frac{11}{24}, -01$ $\frac{221}{3}, -02$ $\frac{21}{32}, -02$ $\frac{21}{32}, -02$ $\frac{21}{32}, -02$ $\frac{11}{7}, -01$ $\frac{24}{33}, -04$ $\frac{25}{21}, -02$ $\frac{11}{7}, -01$				- 2 2 2 2 2 2 2 2 2 2 -				
1 Transp 300-m 300-m Probe Probe 2 Transp 2 Traile LFV 1 3 Transp LRV 1 4 Transp	ort 111	$\frac{111}{22} - 01$ $\frac{321}{3} - 02$ $\frac{321}{3} - 02$ $\frac{21}{3} - 02$ $\frac{21}{3} - 02$ $\frac{21}{3} - 02$ $\frac{21}{3} - 02$ $\frac{24}{3} - 04$ $\frac{24}{3} - 04$ $\frac{2521}{2} - 02$ $\frac{1147}{2} - 01$ $\frac{1124}{2} - 01$				-				
1 Transp 300 - m 300 - m Probe Probe 2 Transp 2 Transp 3 Transp 4 Transp 4 Transp 8 COST :	ort 111	1147-01 321 3-02 21 32-02 21 32-01 21 32-0				- 2 2 2 2 2 2 2 2				
1 Transp 300-m 300-m Probe Probe 2 Transp Traile LFV 3 Transp 4 Transp 4 Transp 2 Transp 3 Transp 4 Transp 2 Transp 3 Transp 4 Transp 2 Transp 4 Transp 2 Transp 4 Transp 2 Transp 2 Transp 4 Transp 2 Transp </td <td>ort 1 drill 7 r 1 ort 1 r 1 ort 1 or</td> <td>2 2 2 2 2 2 2 2 2 2 2 2 2 2</td> <td></td> <td>5 6</td> <td>7</td> <td>2 2 2 2 2 2 2 2 2 2</td> <td>9</td> <td></td> <td>To</td> <td></td>	ort 1 drill 7 r 1 ort 1 r 1 ort 1 or	2 2 2 2 2 2 2 2 2 2 2 2 2 2		5 6	7	2 2 2 2 2 2 2 2 2 2	9		To	
1 Transp 300-m 300-m Probe Probe 2 Transp Traile LFV 3 Transp 4 Transp 5 COST: Clime For Recurrin Cost Coperations Descriptions	ort 1 drill 7 ort 1 ort 1 ort 1 8.0 8.0	2 124-01 121-3-02 122-01 122-01 1		5 6	7	- 2 2 2 2 2 2 2 2	9	10	Tn 68	

Fig. 6-1 Sample Mode Data Sheet

LMSC-A847943 Vol. I

Table 6-1

MODE COMPUTER INPUT

			Also U	sed For
Card Type	Input Item	Where Obtained	"Working" Mode	Emergency Mode
1	 Mode identification number Active-mode crew level Total-mode crew level Mode staytime Sample return capability Minimum decimal fraction of available mode manhours that must be used during any given 	Data sheet Data sheet Data sheet Data sheet Data sheet Program analysis ground rules	$\sqrt{\frac{1}{\sqrt{1+\frac{1}{1+\frac{1}{\sqrt{1+\frac{1}{\sqrt{1+\frac{1}{\sqrt{1+\frac{1}{1}}}}}}}}}}}}}}}}}}}}}}}}}}}}}}}$	√ · √
	 Launch lead time for minor scientific equipment Flag for concurrent shelters Number of payload launches in the mode 	Data sheet Data sheet Data sheet		
2	 Total time for base deployment Shirtsleeve and spacesuit manhours for special operations Lifetime and replacement mass of spacesuits and backpacks 	Data sheet Data sheet Program analysis ground rules		
3	Nonrecurring R&D costs	Data sheet		\checkmark
4	Nonrecurring cost of facilities	Data sheet		√
5	Recurring hardware procurement	Data sheet		\checkmark
6	Recurring operations support	Data sheet		\checkmark
7	 Launch identifier Related launch identifier Timing between related launches Launch mass margin 	Data sheet Data sheet Data sheet Data sheet	\checkmark	V
8	 Launch identifier System on that launch Number of men supported by this system Launch lead time of this system (6-mo increments) Expendables required for special operations for this system 	Data sheet Data sheet Data sheet Data sheet Data sheet	√ * √	\checkmark \checkmark \checkmark
9	 Identifier of crew rescue emer- gency mode Identifier of logistics emergency mode 	Program analysis ground rules Program analysis		

*Transportation systems not required.

The first card type assigns an identification number to the mode and inputs certain general data. Two types of crew level are specified. The "active-mode crew level" is the number of men actually performing the mission experiments at any given time. If one or more additional men are in some other location (such as lunar orbit) throughout the mission, these men plus the "active" crew form the total mode crew level.

Mode staytime is generally determined from the nominal maximum staytime of the principal crew system (such as shelter or rover) supporting the "active crew level."

The second card type inputs operations data including total time for deployment, manhours for any special operations, and spacesuit and backpack replacement requirements. The total time for base deployment is the number of days required to complete deployment whether the crew spends full time at that job or not. The computer bases its calculations on the assumption that the full active crew level is at the base during this time.

The third, fourth, fifth, and sixth card types input costs for nonrecurring R&D, nonrecurring facilities construction, recurring procurement, and recurring operations, respectively. The MIMOSA contract study calculated only recurring operations costs at the mode level: hence, the standard mode data sheet format allows for only this cost.

The seventh card type identifies the various launches of the mode. Each launch requires a separate card. The card inputs any schedule dependency between that launch and any previous launch. It also inputs the launch mass margin.

The eighth card type identifies the equipment on each launch. A separate card is required for each transportation system (paragraph 4.1.1) or piece of mission equipment (paragraph 4.1.2).

The ninth card type is used only if an emergency mode is being considered as backup to the regular mode. Emergency modes are discussed in paragraph 6.6.

The computer takes all input data, stores it in predetermined locations on magnetic tape, and prints out what it has stored. Figure 6-2 shows a typical printout of mode 503-30114-01 as an example.

6.4 'WORKING' MODES

The mission-mode match routine of the computer program requires only part of the total mode input. For example, the computer works only with mission support equipment; it ignores transportation systems. In the match routine, the computer also disregards cost and schedule data. Therefore, it is often desirable, particularly in the early stages of mission-mode matching, to worry only about those items essential to proper performance of the computer. This can be done by using "working modes." Table 6-1 identifies those items that must be input for working modes. Note that for card type 8, transportation systems may be omitted.

6.5 "ZERO-LAUNCH" MODES

Occasions arise when it is desirable to use a set of equipment in a mode for missionmode matching, while ignoring several items of that equipment set in later cost calculations. For example, a mission may be accomplished by a mode in which some items of equipment are being re-used from a previous mission. Under these conditions it is desirable to match the equipment to the mission, but not calculate the cost of that equipment again. The computer has been programmed to handle such cases provided the equipments are grouped into a "zero-launch" mode.

Construct a zero-launch mode for computer input in much the same manner as a regular mode except for three things:

- (1) Begin the mode identifier with the number 9.
- (2) Group all systems that are not to be costed into one "launch" that is called launch 0 (zero).
- (3) Arrange the input cards so that launch 0 is the last one in the sequence of launches.

ŧ ł 1 EXPLANATION OF NOTES (1) NUMBER OF 6-MO. PERIODS PRIOR TO THE FIRST LAUNCH OF THE MODE THAT ALL MINOR SCIENTIFIC EQUIPWENT (1) NUMBER OF 6ELIVERED FOR PAYLOBD INFERTION. INPUT IS 1 FOR 0-6 MO.. 2 FOR 7-12 MO.. ETC. (2) 0 INSTEE DELIVERENT SHELTERS, 1 INDICATES TWO OR MORE CONCURRENT PERSONNEL SHELTERS. (3) MAXIMUM ALLOWED IS 8. (4) NUMBER OF FULL OR PARTIAL 6-MO. TIME PERIODS THAT OCCUR FROM START OF FUNDING UP TO AND INCLUDING THE 6-MO. TIME PERIOD OF A SPECIFIC DEADLINE DATE. i 0000 12.2000 0000 0000 0000 Example Computer Printout - Modes 0000 14.1000 -.0000 0000 0000 ł į (NOTE 1) (NOTE 2) (NOTE 3) 0000 0000. 17.6000 .0000 0000. •0000 • • • • • • • MODE 503-30114-01 COST CURVES 6,0000 2,0000 3,0000 240,0000 33,0000 10,0000 90,0000 50,0000 .75 503-30114-01 n0 ± 13.8000 -.0000 .0000 0000 0000. 0000 --0000 ACTIVE MODE CREW LEVEL TOTAL MODE CREW LEVEL MODE STATTIME (DAYS) SAMPLE RETURN CAPABILITY (KG) MIN. DECIMAL FRACTION LAUNCH LEAD TIME-MINOR SCIEN. EQUIP. CONCURRENT SHELFERS FLAG NUMBER OF PAYLOADS 0000 A.0000 2.5000 Fig. 6-2 0000 0000 TOTAL TIME FOR BASE DEPLOYMENT (DAYS) UEPLOYMENT MAN-HOURS (SHIRTSLEEVE) DEPLOYMENT MAN-HOURS (SHIRTSLEEVE) DEPLOYMENT MAN-HOURS (SPACESUIT) MAN-HOURS OF SUIT LIFETIME MAN-HOURS OF BACKPACK LIFETIME MASS OF BACKPACK REPLACEMENT (KG) MASS OF BACKPACK REPLACEMENT (KG) .0000 -.0000 ••• ì ÷ ទុ - RECURRING ZERO POINT (NOTE 4) VALUES (MILLIONS) OF FACILITIES ZERO POINT (NOTE 4) VALUES (MILLIONS) HECURRING PROCUREMENT ZERO POINT (NOTE 4) VALUES (MILLIONS) RECURRING OPERATIONS ZERO POINT (NOTE 4) VALUES (MILLIONS) WODE IDENTIFIER COST NoN 1 ł ; Į

6-7

	TED LAUNCHES Max, NU, PIASS Montal Mondato	PERIODS AVAILAPLE (KG)	-0. 143A8.0	-0. 15520.0	-0° 1+00°0	EXPENDABLES S SPECIAL OPERATIONS (Κὐ)		c c	0.1	EXPENDABLES S SPECIAL OPERALIONS (KG)	C O O O	EXPENDABLES S SPECIAL OPERATIONS (KG)	2.0 	EXPENDABLES S. SPECIAL OPERATIONS (KG)	0•-		
	TIME BETWEEN RELAT MIN, NO. 4 Montu	PERIODS	.0.1	••••	•01	LAUNCH LEAD TIME (FULL OR PARTIAL 6-MO. PERIODS	BEFORE LAUNCH DATE)		-0-	(EULL_OR_PARTIAL_6-MO. FERIOD <u>S</u>	BEFORE LAUNCH DATE) 2. 2.	(FULL OR PARTIAL 6-MO. PERIODS	BEFORE LAUNCH DATE) 3. -0.	LAUNCH LEAD TIME LAUNCH LEAD TIME (FULL OR PARTIAL 6-M0. PERIODS REFORE 1 ANNEL DATE 1			
	VYLOAD RELATED	MBER NUMBER	1.		+• • •	OAD 1 NUMBER OF MEN SUPPORTED	C			OAD 2 NUMBER OF MEN SUPPORTED		OAD . 3 NUMBER OF MEN SUPPORTED	ман	OAD 4 NUMBER OF MEN SUPPORTED	-0. CREW RESCUE EMERGENCY MODE	LOGISTICS EMERGENCY MODE	
PAYLOADS	7d -	N		4		SYSTEMS ON PAYL SYSTEM IDENTIFIER	201 105	213202	102411	SYSTEMS ON PAYL System Identifier	243304 252102 114701	SYSTEMS ON PAYL SYSTEM IDENTIFIER	242303 114701	SYSTEMS ON PAYL System Identifier	112401 Identifier for	IDENTIFIER FOR	

LOCKHEED MISSILES & SPACE COMPANY

6.6 EMERGENCY MODES

It may be desirable to buy special modes that can be used as an emergency backup to a regular mode. The MIMOSA computer program costing routine has been set up to handle this possibility. As presently programmed, two types of emergency modes can be considered: (1) crew rescue and (2) emergency logistics. Each type of mode is limited to one launch and is procured only once whether actually used or not per program. The computer program requires input of the mode costs and the list of systems on the one launch. Table 6-1 indicates the required data inputs for emergency modes.

LOCKHEED MISSILES & SPACE COMPANY

Chapter 7 MISSION-MODE COMPARISON

The object for comparing mission requirements and mode capabilities is to demonstrate the ability of an equipment evolution to perform a scientific program. This chapter explains the operations required for comparing mission modes using the MIMOSA computer program. These operations include a preliminary hand analysis and computer input preparation. A sample match is presented for indicating certain problems that might be encountered by planners.

7.1 HAND ANALYSIS

After the mission has been formulated tentatively (paragraph 5.1) and a mode has been selected for matching, the following simple hand analysis can save considerable computer time:

- Calculate the available manhours by multiplying the maximum mode staytime by the active (or surface) mode crew level.
- (2) Calculate the approximate scientific manhour capability by multiplying the available manhours by the appropriate factor from Table 7-1.
- (3) Total the scientific manhours associated with the mission by summing all experiment phase 2 (operate) manhours.
- (4) Check to see that this sum is less than 75 percent of the approximate mode scientific manhours capability. (The 75 percent commitment represents an adequate "fit" at this point in the analysis; other factors might tend to increase the requiment as the computer checks in depth, or the percentage requirement may be changed arbitrarily).
- (5) Sum the mass of minor scientific equipment for the experiments making up the mission.
- (6) Make sure this required equipment mass is less than the total mode mass margin.

After the foregoing six steps have been completed, the mission and mode are input to the computer (paragraphs 5.4 and 6.3).

Table 7-1

·	Ratio of Scientific	to Total Manhours
Mode Crew Level	Fixed-Base Mode	Mobile-Base Mode
2	0.25	0.20
3	0.33	0.25
6	0.33	0,25

SCIENTIFIC/TOTAL-MANHOURS RATIOS

7.2 COMPUTER INPUT PREPARATION

After the mission and mode are input to the computer, control cards must be generated that cause the computer to perform the necessary calculations. Only three types of control card are required for a mission-mode match. Paragraph 10.9 gives detailed instructions for preparing these cards.

The first card is required at the beginning of each computer run. Punching a zero in the first column of this card results in the standard printout for match calculations. Punching the number 1 in the first column results in a detailed diagnostic printout of each computer calculation.

The second type of card is required for each mission-mode match being attempted. This card calls out the mission number and mode identification number. Space is also provided on this card to input a multiplication factor to automatically increase the shirtsleeve and spacesuit manhour numbers called for in the experiment data cards. This factor can be used to account for performance degradation in spacesuits and under reduced gravity. It should be noted that the experiment times called out in the North American Aviation data cards are based on man's capability in shirtsleeves, at normal gravity, in Earth environment.

The third card type, also required for every mission-mode match, identifies all flying and roving vehicles performing traverses and establishes the traverses that each vehicle performs. If the vehicle performing the traverse is a rover and trailer combination, special care is required for computer input.

7.3 SAMPLE MISSION-MODE COMPARISON

Figure 7-1 shows the computer printout resulting from an example mission-mode match involving mission 28 (Fig. 5-1) and mode 503-30114-01 (Fig. 6-1). Most computer runs point out definite deficiencies in the mission-mode match. The example shown here illustrates typical printouts caused by sloppy input and demonstrates proper corrective action. The following paragraphs discuss these printouts in detail.

7.3.1 Computer Printout

The computer printout for mission-mode matching (Fig. 7-1) contains one or two pages of comments followed by various calculation summaries. The comments are verbal messages describing faults or other items of interest concerning the match. The comments shown in the example printout are typical of the many possible messages. Pages 7-6 and 7-7 of the numerical summaries present results of the most important calculations. The calculated mission requirements and the input mode capabilities are displayed side-by-side. The ratios of the two values for various parameters are also shown. A perfect mission-mode match occurs if all these ratios are unity.

Pages 7-8 and 7-9 present the traverse calculations traverse by traverse. Page 7-10 presents traverse calculations vehicle by vehicle (in this case, only one vehicle).

Pages 7-11, 7-12, and 7-13 summarize mission requirements traverse by traverse. These calculations are particularly helpful in "trouble-shooting" mission-mode faults. Page 7-14 summarizes mission requirements at the base. In the example shown here, there were no base experiments.

,

-

• . • •

																Vo	ol. I		
· <	1	ł	1		:	1	ı	;			1	ı	1		1	,	•	1	
								;						1	i				
															!				
•			i . I .			1						1 -	:	1					
						;							;	Į	:				
				-		Ĩ			، ز			*							
						ļ	(i				:				· i		
							1						ĺ						
											i								
					1									j i	Í		1	ł	
							1												
																		İ	
																		i.	
														1		1		-	
1														i					
			ļ ,	1	:														
	:			ĺ															
	•			1	:	1				I									
1			:							:					ł		:		
:	! . •		1			:			1		1	ļ	2		i		· · ·		
:	,			ļ				:				1	-				1.1		
l	:		į ·	ł	I				1					i.	Ì				
:				:					į						ĺ		:		
				-	,												÷		
4				+ 1	-				1						İ				
1				!								1				1	;		Ŀ.
				1			ļ		i			1							on
: :				1				i	1	!						:	2		0
			:	:	;	1 . :									1		ł		Ţ
						· . ·			i	ļ		1		:					-2
т	: 			i										1			:		à
							i		!	,				1	ł				Ē
į	·				÷	:	:			:					;				
-				1	:	:	·							•	ł		;		
ATA	ATA ATA	AT A	1		:				į	;				r	I	:	•		
<u> </u>						ł			1							:	:		
EN EN	EN EN	U	:		1		1		i t	1				:	ĺ		÷		
L I	RI.	RI.		1	:	1	į			;					Ì	:	i		
КРЕ	XPE XPE	ХРЕ		ł	; .	;					Ì			-			i		
ш́ т	ա ա։	ω z						•		1			1	,		:	1		
Ä	5 5	ĥ			1				i	j				•		:	:		
40T	107	NOT			i											:	1		
N	2 2	20			ł		Í	1						!		i	i	:	
5°•6	5 6	6°6		1				1					!	1			ł		
6.6	5°6	6°6				i			1	:									
-6•E	6 6	6.6		1	,	1	!		1						; ,				
5.6	6	5. 6							1	. :					1		:	·	
5	12 E	۲.					1								1		•		
IMEI		IME		•		i	;							ł			i		
ĒKJ	ER.	ER]					i		!		i			ı			1 :		
БХР	EXP	ЕXР		:					;		1						:		
_		_		1		i	:		!	1							i		
·	l	;		:		į · ,	1		:				. 1		2	•	.:	~	

LMSC-A847943 Vol. I

5

·				RATIO			· · ·		•766	*****	· ·			•	•
PAGE I	503-30114-01 101AL= 3	EKIMENISE 3 E (DAYS)= 90.	-	PERMITTED VALUE	•06•	6480.			9438.	• 0		520.	•0		
LION	CREW LEVEL	PERFORMING EXP	ÉD"BY THE COMPUTER.	CALCULATED VALUE	89.6 44.9 38.8 38.8 6.0	6454.5	1717.0 1.8.8 1.2 1.2	4737.5 • 2 • 4 • 1	7226-8	2181.0 .U		5045.8 3.8 2.0	0.		
SUMMARY OF MISSION-MODE MATCH CALCULAT	SCIENTIFIC PROGRAM A7 MISSION MIX 01 MISSION NUMBER 28	NOMINAL MISSION START DATE= 1978A MISSION LOCATION- PATH 13.5.4.5 Number of Times at This Location. Counting this Mission- 1	THE LISTING LELUW SUMMARIZES THE MISSION-MODE CALCULATION PERFORME	ІТЕМ ІТЕМ	TOTAL STATIME (DAYS) TIME FOR ALL ROVING TRAVERSES (DAYS) TIME FOR ALL FLYING TRAVERSES (DAYS) TIME FOR BASE ACTIVITIES (DAYS) TIME FOR UEPLOYMENT (DAYS)	TOTAL MANYJOUKS UTILIZED	TOTAL SPACESUIT MANHUURS SCIENTIFIC - ON ALL TRAVERSES (INPUT MULTIPLIER=2.00) 676 SCLENTIFIC - AT BASE (INPUT MULTIPLIER=2.00) 746 NONSCIENTIFIC - ON ALL TRAVENSES NONSCIENTIFIC - AT BASE NONSCIENTIFIC - AT BASE	TOTAL SHIRTSLEEVE MANHOURS SCIENTIFIC - ON ALL TAAVERSES (INPUT MULTIPLIER=1.00) 155 SCIENTIFIC - AT BASE (INPUT MULTIPLIER=1.00) 689 RONSCIENTIFIC - ON ALL TRAVERSES NONSCIENTIFIC - ON ALL TRAVERSES 1598	TOTAL EXPENDABLES UTILIZED (KG)	TOTAL EXPENDAGLES USED AT THÉ BASE (KG) USED FON SYSTEM SUMPORT DURING ALL TRAVERSES	USEL FOR SYSTEM SUPPORT DURING ALL NON-TRAVERSE PERIODS 1397 USED FOR POWER ON BASE EXPERIMENTS EXTRA AMOUNT FOR SPACESUIT USE AT BASE EXTRA AMOUNT FOR SPACESUIT USE ON TRAVERSES USED FOR SPECTAL OPERATIONS UNINGRAPE (1011 AND 1005	TOTAL EXPENDALES USED BY ROVING AND FLYING VEHICLES (KG) USED DUKING TRAVERSES (SEE FOLLOWING PAGES FOR DETAILS) 4643 USED FOR SPECIAL OPERATIONS UNUSABLE (BUILOFF, ULLAGE, ETC.) 400	TOTAL RESUPPLY EXPENDALES REQUIRED (KG) For FLYING TRAVERSES For Roving Traverses For Base Use		

•

Fig. 7-1 (Cont.)

Clent England Mission Martin Mission Martin Mission Martin Mission Locations Part Internet Mission Locations Control Mission Mission Locations Control Mission Mission Locations Part Internet Mission Locations Control Mission Mission Locations Control Mission Mission Locations Control Mission Mission Mission Mission Locations Control Mission Mission Mission Mission Tits Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission Mission	SUMMARY OF MISSION-W	ODE MATCH CALCULATI	NO	PAGE 2		
Multiply updations and multiply and and and and and and and and and and	SCIENTIFIC PROGRAM A7 MISSION MIX 01 MISSION NU	MRER 28	CREW LEVEL	1 503-30114-01 7074L= 3		
Model of Taris A Taris Controle ruls HISEONe 1 1 FE LISTIA ULE UN SUMMERS THE HISEONHOLE CALCULATION REPORTED UNTER FROMMERS THE GALOLATION REPORTED WALL 601 1011. LEV SUMMERS THE HISEONHOLE CALCULATION REPORTED UT THE CONTROL WALL 610 1011. LEV SUMMERS THE HISEONHOLE CALCULATION REPORTED UT THE CONTROL WALL 610 1011. LEV SUMMERS THE HISEONHOLE CALCULATION REPORTED UNIT 9 1011. LEV SUMMERS THE HISEONHOLE CALCULATION REPORTED HISEONHOLE 9 1011. LEV SUMMERS THE HISEONHOLE 9 1012. LEV SUMMERS THE HISEONHOLE 9 1013. LEV SUMMERS 9 1014. MASS SUMMERS 9 1015. STATE SUMMERS 9 1016. STATE SUMMERS 9 1017. WASS SUMMERS 9 1018. STATE 9 1019. STATE 9 1010. STATE 9 1011. STATE 9 1010. STATE 9 1011. STATE 9 1011. STATE 9 1011. STATE 9 1011. STATE 9 <t< th=""><th>MISSION LOCATION- PATH TA-544.5</th><th>New and the second states of the second s</th><th>PERFORMING E) MAX MODE STAYLIN</th><th>(PERIMENTS= 3 HE (DAYS)= 90.</th><th></th><th></th></t<>	MISSION LOCATION- PATH TA-544.5	New and the second states of the second s	PERFORMING E) MAX MODE STAYLIN	(PERIMENTS= 3 HE (DAYS)= 90.		
$\label{eq:linear} \begin{tabular}{ c c c c c c c c c c c c c c c c c c c$	NUMBER OF TIMES AT THIS LOCATION COUNTING THIS M	ISSION- 1				
1014. Uv Under Ling Curcurate way Parto 1014. Uv Vinou-Lung 1014. West short with Ling 1010. 1010. 1014. Vev Vinou-Lung 1010. 1010. 1010. 1010. 1014. Vev Vinou-Lung 1010. 1010. 1010. 1010. 1010. 1014. Mass short with 20 (ci) 11270 (ci) 11270 (ci) 11270 (ci) 1010. 1010. 1010. 1014. Mass sterring Sterring 11270 (ci) 11270 (ci) 1130.0 1010. 1010. 1010. 1014. Mass sterring Sterring 11700.0 11700.0 11700.0 1010. <td>THE LISTING BELOW SUMMAHIZES THE MISSION-MODE CA</td> <td>LCULATION PERFORMED</td> <td>BY THE COMPUTER.</td> <td>* * *</td> <td></td> <td></td>	THE LISTING BELOW SUMMAHIZES THE MISSION-MODE CA	LCULATION PERFORMED	BY THE COMPUTER.	* * *		
101A Fright Fright Mediation Mediation 0 1000 000 10AS 05 should UTLIZED GUIPHER 11700.0 000 000 10AS 05 should UTLIZED GUIPHER 11700.0 50 1000 10AS 05 should UTLIZED GUIPHER 11700.0 50.0 1000 10AS 05 should UTLIZED GUIPHER 11700.0 50.0 1000 10AS 05 should UTLIZED Web 11700.0 50.0 50.0 10AA 05 should UTLIZED Web 11700.0 50.0 50.00 10AA 05 should UTLIZED Web 11700.0 50.0 50.00 10AA 05 should UTLIZED Web 11700.0 50.0 50.00 10AA 10AA 10AA 10AA 10AA 10AA <t< td=""><td></td><td>C1</td><td>ALCULATED VALUE</td><td></td><td>RATIO</td><td></td></t<>		C1	ALCULATED VALUE		RATIO	
Total solution 1199-9 2210+9 0740-0 651 Mass of syntax solution (scientific culterent mass of syntax solution (scientific culterent solution (scientific culterents) 1,199-9 1199-9 1199-9 151-0 <td>TOTAL LEV PROPELLANT UILLIED (KG)</td> <td>a da ana ang ang ang ang ang ang ang ang an</td> <td>0.</td> <td>1030. 0.</td> <td>000.</td> <td>,</td>	TOTAL LEV PROPELLANT UILLIED (KG)	a da ana ang ang ang ang ang ang ang ang an	0.	1030. 0.	000.	,
Tita and Taylor of Taylor	TOTAL MASS MARGIN UTILIZED (KG) Mass of Minok Scientific Equipment Mass of Spare Spacesuits Mass of Spare Backpacks	5. REGU 15. REGD	31789-9 31789-9 165-0 150-0	47640.	•674	
Fig. 7-1 (Cont.)	TOTAL EANTH RETURN MASS UTILIZED (KG) FOR BASE EXPENIMENTS FOR TRAVENSE EXPERIMENTS		1790.0 1790.0	50+	35.800	
Fig. 7-1 (Cont.)						
Fig. 7-1 (Cont.)				ander a rich o warden under an and a feinement under a gent		
Fig. 7-1 (Cont.)		and a second commence of the second se	· · · · ·			
Fig. 7-1 (Cont.)	and a summary we have a many many many many many many many a sum and and a sum of a subject of the sum of the s	a second and a second second second second second second second second second second second second second second				1
Fig. 7-1 (Cont.)					:	
Fig. 7-1 (Cont.)		ter Mandala Managero o A. Allo - avec V. La L. Mandala -	an han ann an Anna a mar ann ann ann ann ann ann ann ann ann a			
Fig. 7-1 (Cont.)						
Fig. 7-1 (Cont.)						i
Fig. 7-1 (Cont.)			a mananan mananan manan a ka ana a ka ka		ŧ	•
Fig. 7-1 (Cont.)		 An example of the second se Second second li>	Anna annaise a' a' a' anna anna an a' a' a' a' a' a' a' a' a' a' a' a' a'		• • •	
Fig. 7-1 (Cont.)				and a second second second second second second second second second second second second second second second		
Fig. 7-1 (Cont.)		And advantuments of the state o				
rlg. 7-1 (Cont.)	Ĭ			a marine a summer a summer a sum a su	-	
	F1g.	7-1 (Cont.)				

LMSC-A847943 Vol. I

7-7

;

	SUMMARY OF MISSI	ON-MODE MAT	CH CALCUL	VTION	PAGE 3		- 5
S(LIENTIFIC PROGRAM A7 MISSION MIX 01 MISSIO	N_NUMBER2	39	MODE I.	D. NUMBER 503-30114-01		-
JZ	DMINAL MISSION START DATE= 1976A		•	PERF MAX MOD	ORMING EXPERIMENTS= 3 DE STAYTIME (DAYS)= 90.	:	i
-	THE LISTING BELOW SUMMARIZES THE TRAVERSE C	ALCULATIONS	PERFORMED	D BY THE COMPU	JTER.		
	KAVERSE IDENTIFIER /EHICLE IDENTIFIER	R-01 2433-04	R-02 2433-04	R-03 2433-04	· · · · · · · · · · · · · · · · · · ·		ł
	IIVIMUM CREW REQUIRED CEHICLE CREW LEVEL CREW AT GASE DUKING TRAVERSE	N M D 1	-1 10 O	د بن <mark>و</mark> ا			. 1
⊢ <u>−</u> .	KRAVERSE RANGE (KM) Vaximum Veniclé Capability (Km)	0.0007	350.0 7000.0	1150.0			
	REGUIRED EXCURSION TIME (DAYS) Aaximum vehlicle capability (days)	19.4 400.0	4•5 400•0	20•9 400•0			1
с	NUMBER UF STOPS	10.0	10.0	10.0	-		
	(ASS OF MINUR SCIENTIFIC EQUIPMENT (KG) (ARTH RETURN MASS REGUIRED (KG) (ASS OF SPACESUIT REPLACEMENTS (KG) (ASS OF BACKPACK REPLACEMENTS (KG)	12764.5 452.0 10.0	6227.5 145.0 .0	6842.5 1193.0 .0 10.0			r
<u>,-</u> €	TOTAL CARGO MASS REJUTRED (KG) Aaximum vehicle Capability (kg)	13226•5 7660•0	6372.5 7660.0	8045.5 7660.0	· . :	÷	
	SCLENTIFIC WAN-HOURS ON TRAVERSE Shirtsleeve Spacesuit	402.8 90.8 312.0	50.4 3.6 46.8	378•8 60•8 318•0		• • • • • • • • • • • • • • • • • • •	1
	JONSCIENTIFIC WAN-HOURS ON TRAVERSE Shirtsleeve Spacesuit	997.2 952.5 44.7	273•0 262•6 10•3	1127.4 1079.3 48.1		·····	• •
	SCIENTIFIC MAN-HOUMS REGUIRED AT BASE Shirtsleeve Spacesuit	197.8 115.8 82.0	30.5 24.3 6.2	1207.7 549.7 658.0		- 1	
	ONSCIENTIFIC, MAN-HOURS PEQUIRED AT BASE		000	•••		-	t :
	rraverse expendagles required (kg) Aaximum expendables permitted (kg)	1480.7 7520.0	532.9 7520.0	2630.2 7520.0	· · · · · · · · · · · · · · · · · · ·	:	
]	JASE EXPENUABLES FOR SYSTEM SUPPORT	0.	•	0•			ł
	EELTA V HEGUIRED OF LEV (M/S) Specific impulse (M/S) Inert Mass of Lev (KG)	ۄؘڡ۪؋	000				ĩ
	ATTROPELEANT PERMITTEU (KG)	•••				· · · · · ·	ı
	-	Fig. 7-1	(Cont.)	: : : .			•

LOCKHEED MISSILES & SPACE COMPANY

≺-	-	1	!	!					t	! · ·		1	
PAGE 4	I.D. NUMBER 503-30114-01 LEVEL TOTALE 3 RECOMING EXPERIMENTS 3	QUE SIATIME (DAIS) - 90.	R THE TRAVERSES REGUIRED ON				LE EQUALS OF ANY CALCULATED EXPENDABLES Ement of any other expendables indicates Roving vehicle <u>Acting As</u> a shelter, and To 0.						
VIION	MODE CREW	M. A.A. M.	ULATIONS FU	R-03 2433-04	2630.2 7520.0	652.7 943.4 851.0 183.1	COVING VEHIC (THE REQUIR HELTER, OR TO BE EQUAL		•				
TCH CALCULA	28		DABLES CALC	R-02 2433-04	532.9 7520.0	140.1 105.2 259.0 28.6	FLYING OR R T MANHOURS AINST THE S CONSIDERED	ann ha chuir a' ann ann ann an an a	•				1 (Cont.)
SION-MODE MA	ION NUMBER	•	OF THE EXPEN	R-01 2433-04	1480.7 7520.0	606.7 29.6 666.0 178.4	ITTED FOR A FOR SPACESUI E CHARGED AG RAVERSE ARE						Fig. 7-
SUMMARY OF MISS	SCIENTIFIC PROGRAM A7 MISSION MIX 01 MOMINAL MOSCION FIANT DATE 107.4	NUMITIAL MISSION START DALE- 1970A	THE FOLLOWING LISTING SUMMARIZES DETAILS	TRAVERSE IDENTIFIER Vehicle identifier	TRAVERSE EXPENUABLES REQUIRED (KG)	EXPENDAULES USAGE BREAKDOWN (KG) VEHICLE SUPPONT BASED ON DAILY USE EXPERIFENT EWERGY FOR TRAVELINU EOR STGPS EXTRA FOR SPACESUIT MANHOURS	NOTE WHEN THE MAXIMUM EXPENDABLES PERM REGUIRED ON A THAVENSE SHOULD BE AN ERROR IN INPUT DATA). THESE AR THE EXPENDABLES REGUIRED ON THE T						

															•0	/1. 1			
-	— ,	;	·		•	!	r	1	. 8	i	2	ĩ	•		•	ı	!	ı	
									:									•	
									IAT DATA.						-				
				:) RY TH VDABLES INPUT	:		:				:		:	
:	1 1 1 1 1 1 1 1	06	VIIONS	:			:	-	EQUIRE EQUIRE L EXPER			:		-	and a second second second second second second second second second second second second second second second	4		!	
AGE 5	-30114 TOTAL=	MENTS= AYS)=	CALCUI				:	:	BLES RI E TOTAI ERROR	•	l	۱ ۰				•			
¢.	1ER 503	EXPERI IME (D	ELLANT		1 1	· · ·		:	XPENDA AND TH IS AN	:	:	-		;				:	
	0. NUMF VEL	DRMING E STAYI	JD PROP		•		·		- THE E CLTER. THERE	•		•	•	:				•	
	REW LE	AX MODI	BLES A		:				ALL OL S A SHI F NOT.		:	·			:				
IION	20	Σ	XPENDA	!	!	••••		:	0. THEN Sting A VL 0. I	•	•		1			:		:	
ALCULAT			LS OF E	;				•	EQUAL (ICLE AC	:				:	-			i	**
ATCH C	28		DETAI					:	HICLE NG VEH S SHOU	:		1							
-MODE N	JUMBER		OF THE	2433-04	5043.8 7520.0	1399.5 1078.3 1776.0	390.00 0.004	000	ON A VE DR ROVI ENDABLE		: :		•		1			1	- L 2
NOISSI	SSION		UMMARY		KG) KG)			(9)	LY EXPL		;	1					i		Ğ
₹¥ OF M	IW.		HICLE S		LTTED (JIRED (USE	2) (KG) FTED (K IRED (K	ES PERM FHE SHE RESUPP		•	-	- - -				:		-
SUMMAF	:	19764	åY VEH UN PA(S PERM.	ע (גה) UAILY	OURS ONS E+ ETC	EQUINEL PERMI T REGU	ENDAULI ED BY ASS GF	 		:					•	•	
		DATE=	(EHICLE PORTED		COLIRED CNDABLE FINDABL	REAKDOW NSED ON	T MANH DPERATI ULLAG	LANT R ELLANT PELLAN	NUM EXP SUPPLI O THE M	! : :		i			1		ł		
	GKAM AT	N STARI	IS A V SES RE	IFIER	JLES RE DF EXPE	SAGE BF PORT B/ ENERGY	PACESUI	PHOPEL OF PROF	E MAXIA LE ARE Red And	:	1			ı			:		
	IC PHO	OTSSIM	LOWING	IDENT	XPENUAI MASS	LE SUPI LE SUPI LACELT RAVELTI	FOR SPI	ASS OF MASS HESUPI	LF THI VEHICI REGUI	i		2	1	1	1	1	i	•	
	LENTIF M	DMINAL	THE FOL	/EHICLE	FOTAL E MAXIMUM MASS OF	EXPENDA VEHIC EXPER	FOK S EXTRA USED UNUSA	OTAL M AXIMUM ASS OF	40TE	•		:		ı				,	
	50	ž		-		u .			-	·								·	
	-	: 		•				* :	, t										
:	l		ł	I	i	· 1	:		•	·	l		ł	•	. :	•	ei -	•	

LOCKHEED MISSILES & SPACE COMPANY

Fig. 7-1 (Cont.)

LMSC-A847943 Vol. I

1
Mark Mode Statistics Mark Mode Statistics Statistics	NOMINAL MISSIO MISSION LOCATI NUMBER OF TIME		-	Ĕ	N MOTOST	UMISEN 20			PERFORMIN	IG EXPERIME	NTS= 3	
The LISTING JELUM SUMERAIZES THE STENTIFIC PROGRAM REGULIREMENTS FOR THIS MISSION. THE LISTING JELUM SUMERAIZES THE STENTIFIC PROGRAM REGULIREMENTS FOR THE MISSION. THE ARL IO FLUIDE EXCERTENTIS KENE PERFORMED ON TANKERE POOL THE PLALLOLING EXCERTENTIS KENE PERFORMED ON TANKERE POOL THE PLALLOLING EXCERTENTIS KENE PERFORMED ON TANKERE POOL THE PLALLOLING EXCERTENTS KENE PERFORMED ON TANKERE POOL THE PLALLOLING EXCENTION STAL COUPPORT COUPPORT COUPPORT NUMBER REVLIC STAL CPAN JULY STAL COUPPORT COUPPORT COUPPORT NUMBER REVLIC STAL CPAN JULY STAL CPAN JULY STAL COUPPORT COUPPORT NUMBER REVLIC STAL CPAN JULY STAL CPAN JULY STAL COUPPORT CO	1	N START ON- PAT S AT TH	DATE= 19 H 13.5. IS LOCATI	76A 4.5 On COUNTI	NG THIS I	-NOISSIM-	1	MAX	. MODE STA	YIIME, (DAY)	S)≝. 90•	
The Fact Load In Contract Control of Transformer Control of Contro of Contro of Control of Control of Control of Contro of	THE LISTING UE	MUS WU	MERIZES T	HE SCIENT VOERIMENT	IFIC PRO	GRAM REGUI	IREMENTS F	OR THIS	MISSION.	:	· · · · ·	
ZPREMIRTATI PHAGE Contraction PhAGE Contraction Contr	THE FOLLOWING	EXPERIM	IENTS WERE	PERFORME	D ON THA	VERSE P-01						
Y Y Y Y Y Y Y Y Y Y Y Y Y Y Y Y Y Y Y	EXPERIMENT P	HASE 2 TEPLIC	(TRAVE SPSU	M A N H RSE) SHSL	0 U R S (AT SPSU	REOU BASE) SHSL	I K E D (10T SPSU	AL)	EQUIPMENT ENERGY (KW-HR)	EARTH RETURN MASS (KG)	EOUIPMENT MASS CODE KG	VEHICLE Cargo Cone kg
7:10:10:10:10:10:10:10:10:10:10:10:10:10:	9.9.99.99.12 4.0.05.02.01	n -	0•4	0.	14.6	0.6	18•6 .8	0.6	1.2	0,0	24.0	
2::10:01:4:12 1: 4:0 2::0 1:0	5.9.99.99.36	• •	17.6	t•t	2.0	2.9	19.6	7.3	2.1	7.0	61.0	
59999918 1 22.0 7.4 22.0 7.4 290.0 150.0 1 5441010101 1	2.1.01.04.12 2.1.02.04.67		0.4	22.0	0.4	٩	0.1	22.0	22.0	10.0	110.0	00
11.1.11.11.11.11 1 0 7.4 2.0 7.4 2.0 7.4 2.0 1.1 1.0 1.1	01.69.99.99.9	• •	22.0	••	,		22.0			0	150.0	. œ
59,99,99,21 1 600 2 0 <	1.1.01.01.01		•	ò	4° -	2°0	, t 1	0°0 7°0		0°0	150.0	e. (
3.1.010.04.02 1 0 <			0.00 0.00	25.0			4•0	25.0		0.05	0.065 7.1	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	3.4.01.04.62	1.	0.	ມ ສ	•	•	0.	4•5	1.0	0	20.0	e e
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	4.1.01.04.14		10.0	•	0	- • •	10.0	•	с, с ,		6000.0	6000
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	5.9.99.99.12	•••	1.0	•••		2.8	1 N	2.8	0 4 N			- +
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	9.0.05.02.01			•	i nj	0	3	0.		0	30.0	0.0
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	4.1.01.04.14	. .	10.0	•	• •	• •	10.0	0.0	Ģ	0.	6000.0	6000.0
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	9.9.99.99.10 2.1.02.03.07		0.55		 		0.12	••		•••	0.0c1 B	150.0
9,9,99,99,36 1. 1.2 2.7 1.2 2.7 1.2 2.1 7.0 8.4 1.2 0 24.0 4,00,09,044 1 3 3.0 0 12.6 8.4 15.6 8.4 1.2 3.5 3.6 0 1.0	2.1.01.04.12	1.	3.0	20.0	c.	0	3.0	20.0	22.0	10.0	8	110.0
4.0.08.04.54 1 .0 <td>9.9.99.99.36 0.0.00.00.12</td> <td></td> <td>11.0</td> <td>2.5</td> <td>1.2</td> <td>2.7</td> <td>12.2</td> <td>6° 4</td> <td>2.1</td> <td>7.0</td> <td>с. с.</td> <td>61•(24.4</td>	9.9.99.99.36 0.0.00.00.12		11.0	2.5	1.2	2.7	12.2	6° 4	2.1	7.0	с. с.	61•(24.4
4.1.05.03.41 1. .0 1.0 21.0 23.0 23.0 23.0 23.0 23.0 25.0 57.0	4.0.08.04.54	1.	0.		0	0				0.	3.5	n
h, h, 0/10h, 21 1 $h, 0$	4.1.05.03.41	-	•	•	•	0.	•	•	.	•	0.4 4	3
9.9.99.99.33 46: 0 11.2 15.8 14.7 15.8 25.9 3.9 230.0 86.0 66.0 9.9.99.99.34 5. 51.2 0 10 57.0 100.0 57.0 57.0 9.9.99.99.35 3. 49.6 0 3.8 8.5 53.4 8.5 6.3 45.0 75.0 75.0 75.0 75.0 101ALS 312.0 90.8 82.0 115.8 394.0 206.6 80.1 452.0 13264.2 12764.5	4.4.0/.04.51 4.4.04.44.32		• N.	-		0.0	0°.	0.0			1.0	
9.9.99.34 5. 51.2 .0 1.8 44.6 53.0 44.6 1.0 100.0 57.0 57.0 9.9.99.35 3. 49.6 .0 3.8 8.5 53.4 8.5 6.3 45.0 75.0 75.0 ToTALS 312.0 90.8 82.0 115.8 394.0 206.6 80.1 452.0 13264.2 12764.5	9.9.99.99.33	9		11.2	15.8	14.7	15.8	25.9	0 0	230.0	86.0	96.1
1.0.4.5.0 5.0 45.0 5.0 75.0 75.0 75.0 75.0 75.0 75.0 75	9.9.99.99.24		51.2	0.	1.8	44.6	53.0	9.14	1.0	100.0	57.0	57.(
. ToTALS	GC•66•66•6•	'n	4.64	•	9•9	e.8	53.4	8•5	6•3	45.0	75.0	75.(
	TOTALS		312.0	90.8	82.0.	115.8	394 • D	206.6	80.1	452.0	13264.2	
						Ann - Annya Propaga a Anna a Anna a					** ***	1
									1			
	and the same of th											

LOCKHEED MISSILES & SPACE COMPANY

MOMIAL MISSION THE: 1794 MARKED OF THES AT THIS LOCATION CONTINE THIS MISSION- IF E LISTING GELVA SUMMERIZES THE SCIENTINE HIS MISSION- THE FOLLANDE EXPERIMENT FOR THIS MISSION- THE POLLANDE EXPERIMENT FOR THIS MISSION- MINERER MINISSION- MINISSION	OMINAL MISSION START DATE= 1976A ISSION LOCATION FATH I3.55445 UWBER OF TIMES AT THIS LOCATION COUNTIN HE LISTING JELUW SUMMERIZES THE SCIENTI HE FULLOWING EXPERIMENTS WERE PERFORMED XPERIMENT PHASE 2 ***********************************	NG THIS MISS IFIC PROGRAM D ON TRAVERSI O U R S BASE SPSU SPSU 0 5.44 5.44 6.2 6.2	ION- 1 REQUIREMENTS F E R-02 E 0 U I R E D -0 00 00 00 00 00 00 00 00 00 00 00 00	PA PA PA PA PA PA PA PA PA PA PA PA PA P	ENFORMING MODE STAN MODE STAN EUEPHENT ENERGY (KW-HR)	5 EXPERIMEN FIIME (DAYS RETURN MASS (KG) 0 0 15.0 15.0	TTS= 3 EQUTPMENT EQUTPMENT CODE KG B 6000.0 B 00 B 00 B 00 B 00 B 00 B 00 B 00	VEHICLE CARGC CARGC CODE KG 6000.0 1.0 1.0 1.0 75.0 57.0 57.0
The LISTING UELOW SUMERIZES THE SCIENTIFIC PROGRAM REQUIREMENTS FOR THIS MISSION. The Fullouling Extrements were regressing on traverses reads ExpErturent PMSK 2 Experuent PMSK 2	HE LISTING BELOW SUMMERIZES THE SCIENTI HE FOLLOWING EXPERIMENT WE FOLLOWING EXPERIMENT NP XPERIMENT PHASE XPERIMENT PHASE TRAVERSE) NUMBER REPLIC SPSU 1.011 0.0 0 1.015 0.155 0 0.09504451 1 0 1.0503.41 1 0 1.0503.41 1 0 1.0503.41 1 0 1.0503.41 1 0 1.0503.41 1 0 1.0503.41 1 0 1.0503.41 1 0 1.0503.41 1 0 1.0503.41 1 0 1.0503.41 1 0 1.0503.41 1 0 1.0503.41 1 0 1.0503.41 1 0 0.09.999.33 1 0 0.0101.51 2 0 0.0101.51 2 0 0.0101	IFIC PROGRAM D ON TRAVERSI O U R S BAR (SPSU BASE (SPSU BASE 5.44 5.44 6.2 6.2	E R-02 E R-02 E C U I R E D HSL SPSU 0 0 0 0 0 0 0 0 0 0 105 8 15.8 55.4 17.0 20.8 1.5 16.8 1.5 20.8 24.3 53.0	-OK THIS M IAL EC SHSL -0 -0 -0 -0 -0 	ISSION. GUIPHENT ENERGY (KW-HR) (KW-HR) (KW-HR) 280.0 280.0 280.0 280.0 281.6 1.5 2.1 2.1 2.4 2.1	EARTH RETURN MASS (KG) 0 0 0 15.0 145.0	EQUIPMENT EQUIPMENT MASS MASS F B 6000.0 B B 0.0 B B 0.0 B B 0.0 0 B 0.0 0 0 0	VEHICLE CARGO CARGO CODE KG 6000.0 1.0 1.0 1.0 75.0 75.0
LAFELINERIN (MUNERI ALL) LAFELINE (Composition) Composition (Composition) Composition (Composition) Composition (Composition) Composition (Composition) Composition (Composition) Composition (Composition) Composition (Composition) Composition (Composition) Composition) Composition (Composition) Composition) Composition Composition) Composition) <thcomposition)< th=""> Composition)</thcomposition)<>	XPEKIMENT PHASE M. A. N. H. G. NUMBER REPLIC (TRAVERSE) SPSU SFSU SHSL 1.01.04.14 1. SPSU SHSL 1.01.05.03.41 1. 10.0 0 0 1.05.03.41 1. 1. 0 0 0 1.05.03.41 1. 1. 0 0 0 0 9.99.9933 18. 2. 20.4 0 3.6 0 0 9.99.9933 2. 20.4 2. 20.4 0 0 0 9.99.9933 2. 20.4 3.6 46.4 0 0 0 9.99.9935 1. 16.4 0 3.6 0 0 0 9.1ALS 46.6 3.6 1 16.4 0 0 0	0 1 2 2 2 2 2 2 2 2 2 2 2 2 2	E © U I R E D HSL SPSU 10-0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	IAL) EC SHSL) C 	GUTPHENT ENERGY (KW-HR) (KW-HR	EARTH MASS (KG) MASS (KG) 0 0 0 15.0 145.0	ZOUIPWENT MASS MASS MASS MASS MASS MASS MASS MAS	VEHICLE CARGC CARGC CARGC CARGC 6000-0 3-5 4-0 1-0 1-0 75-0 75-0 57-0
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	1.011.04.14 1. 10.0 0 0.004.544 1. 0 0 0 1.05.04.544 1. 0 0 0 4.07.04.51 1. 0 0 0 9.99.99.33 18. 0 0 0 9.99.99.35 1. 16.4 0 9.99.99.35 1. 16.4 0 101ALS 46.8 3.6	0 Q 0 Q 0 Q	•0 10.0 •0 •0 •0 •0 •0 •0 •0 •0 •17.0 5.4 17.0 20.8 1.5 16.8	00 9.4 17.0 27.9	.0 .0 .0 .0 .0 .0 .0 .0 .0 .0 .0 .0 .0	• 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0	60000 6000 6000 6000 6000 6000 6000	6000.0 3.5 4.0 4.0 1.0 1.0 7.0 75.0 6227.5
	1.000-04-04-04 1. 00 00 4.07-04-51 1. 00 00 4.07-04-51 1. 00 00 9-99-99-33 18. 0 9-99-99-34 2. 20.4 00 9-99-99-35 1. 16.4 00 9-99-99-35 3.6	0 0 0 0 0	24.3 53.0	11.5 27.9	.0 280.0 1.5 2.1 2.1 2.1 2.1	00 00 15.0	8 8 8 8 8 9 0 0 0 0 0 0 0 0 0 0 0 0 0 0	4.5 44.0 757.0 577.0 6227.5
	101ALS 11. 00 00 00 00 00 00 00 00 00 00 00 00 00	0 0 0 0 0 0	-0 -0 5.8 5.4 17.0 20.8 1.5 16.8 24.3 53.0	00 9.4 17.0 27.9	280 • 0 280 • 0 2 • 1 284 • 4	00000000000000000000000000000000000000	8 8 60000 60000	1.0 1.0 86.0 57.0 75.0 6227.5
9.9-99-99-331 10. 2010 10. 201	9.99.99.33 IR0 3.6 9.99.99.35 IR. 20.4 .0 9.99.99.35 I. 16.4 .0 101ALS 46.8 3.6	0 Q	5.8 5.4 17.0 20.8 1.5 16.8 24.3 53.0	27.9	2.1	90.0 15.0 145.0	0.000 0.0000 9	6227.5
70199195155 - 11: 10:0 - 14 11:0 10:0 - 15 10:0 - 10 - 15 10:0 - 10 - 10 - 10 - 10 - 10 - 10 - 10	107ALS 1. 16.4 3.6	. 0 . 1	24.3 53.0 24.3 53.0	27.9	2.1 2.84.4	15.0 145.0	8 8 6000.0	57.0 75.0 6227.5
10TALS 46-16 3-16 6-2 24-3 53-10 284-4 145-0 6000.0 6227	101ALS 46.6	6•2	243	27.9	284•4	145.0	6000.0	6227.5
								, ,
	ı							
			•	:			!	
	· · · · · · · · · · · · · · · · · · ·				and an example of the same	an a color of another sector as the body		
	· · · · · · · · · · · · · · · · · · ·	-						
				· · ·				· ·
			·					

LOCKHEED MISSILES & SPACE COMPANY

Schlick (London 2013) Mission (London 2013) Mission (Link (London 2013) Mission (Link (Missi Anti) Mission (Link (Missi Anti) Mission (Missi Anti) Mission (Link (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Mission (Missi Anti) Missin (Missi Anti) Mission (Missi	SCIENTIFIC											
Figure 10:10:10:10:10:10:10:10:10:10:10:10:10:1	7 4	C PROGRAM AT SSTUN MIX U	. 7	ž	ISSION N	UMBER 20		CRE	E I.D. NU W LEVFL	MBER 503-30 TO:	0114-01 TAL= 3	
Iff C L13T110 ELA SUMPRITES THE SCIENTFIC POCOMM REQUIRENENTS FON THIS WISSION COUPERAT VERTURE THE FCLUENTURE EXPERIENCIALS FOR COMM REQUIRENENTS FON THAT THE FOULTER POLY ENDINENT FOULTER'N VERTURE UNCENTRATION AFFLIC SFSU SFSU <th>NOMINAL WI MISSION LO NUMBER OF</th> <th>ISSLON START Ocallon- Pat Times at th</th> <th>DATE= 19 H 13.5. IIS LOCATI</th> <th>7,84 4,5 ON COUNTI</th> <th>NG THIS</th> <th>-HOISSIM</th> <th></th> <th>MAX</th> <th>PEKFURMIN MODE STA</th> <th>G EXPENIME YTIME (DA<u>Y</u>9</th> <th>5)=</th> <th>·</th>	NOMINAL WI MISSION LO NUMBER OF	ISSLON START Ocallon- Pat Times at th	DATE= 19 H 13.5. IIS LOCATI	7,84 4,5 ON COUNTI	NG THIS	-HOISSIM		MAX	PEKFURMIN MODE STA	G EXPENIME YTIME (DA <u>Y</u> 9	5)=	·
THE FOLLOWING EPERIMENTS MEET OUT FOR TANK CUPRENT TO COLPUENT WITH OUT OUT SET OUT FOR TANK CUPRENT TO COLPUENT WITH OUT OUT SET OUT FOR TANK CUPRENT TO COLPUENT WITH OUT OUT SET OUT FOR TANK CUPRENT TO COLPUENT WITH OUT OUT SET OUT FOR TANK CUPRENT TO COLPUENT WITH OUT OUT SET OUT FOR TANK CUPRENT TO COLPUENT WITH OUT OUT OUT OUT OUT OUT OUT OUT OUT OUT	115LJ BHI	NG PELON SUN	MERIZES T	HE SCIENT	IFIC PRO	GRAM REQUI	REMENTS FO	R THIS	•NOISSIW	:		
Transmission Confliction An II Ho U IS R E O U I R E O Confliction Con	THE FULLON	WING EXPERIM	ENTS WERE	PERFORME	D ON TRA	VERSE R-03						
$\label{eq:relation} F_{1}^{(1,1,1,1,1,1,1)} = \left\{ \begin{array}{cccccccccccccccccccccccccccccccccccc$	EXPERIMENT 	T PHASE 2 - BEPLIC		M A N H RSE) SHSL	0 U R S (AT SPSU	REQU BASE) SHSL	I R E D 	ארך) י ור)	EQUIPMENT ENERGY (KW-HR)	EARTH RETURN MASS (KG)	EQUIPMENT MASS CODE KG	VEHICLE CARGO CODE KG
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	1.1.01.01.0	01 . 1.	0.	0•	7.4	2.0	7.4	2.0	.1	5.0	150.0	8)
Fig. 7-1 Fig. 7-1 <th< td=""><td>1.2.01.02.0</td><td>02 100.</td><td>0.0</td><td>0.10</td><td>400.6</td><td>200.0</td><td>400.6</td><td>200.0</td><td>400.0</td><td>500.0</td><td>290.0</td><td></td></th<>	1.2.01.02.0	02 100.	0.0	0.10	400.6	200.0	400.6	200.0	400.0	500.0	290.0	
F(2,7) = (2,	4 • 1 • 01 • 01 • 04 • 1	51 1.	10.0				66.U	25.0	1.6	30.0	1.7	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	2.1.02.03.0	08 1.	0.	0.	170.0	213.0	170.0	213.0	2100.0	200.0	0 • 0 • 0 • 0	 . <
$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$	6*6*66*6	12 10.	10.0	a.	42.0	28.0	52.0	28.0	4.1	0.	24.0	с.
Fig. 7-1 (Cont.) = 100 + 100	4.2.02.02.4	1. 1.	•	-	• ;	•••	•	.	ç	•	0.°	B 185.
Fig. 7-1 (Cont.) Fig.	2.1.02.03.(0.7 1.	• •		4.8	•••	. 1		• • •		۵ م	H IRD.
Fig. 7-1 (Cont.) = 100 = 0 = 100 = 0 = 100 = 0 = 100	1.94.99.99.9	12 1.	1.0	0	4.2	2.8	5.2	2 •8		0		
$\label{eq:relation} F_1 = 100 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0$	4.1.01.04.1	13 1.	16.0	•	0.	6.	10.0	•	0.	0.	2500.0	2500.
Fig. 7-1 (Cont.) = 1100 = 100 = 100 = 100 = 100 = 100 = 100 = 100 = 10000 = 1000 = 10000 = 1000 = 1000 = 10000 = 10000 = 1000 = 1000	4°1°01°04°1	12	10.0	•	•••	•	10.0	•	<u> </u>	•	2100.0	2100.
Fig. 7-1 (Cont.) = 1100 = 10	4 • 1 • 1 • 0 • 1 • 1 • 1 • 1		0.01	•		•	0.01	•		•••	1300.0	. 1300.
Fig. 7-1 (Cont.) = 1 = 100 = 0 = 0 = 0 = 0 = 0 = 0 = 0 =	4.1.01.04.(1. 09	10.0		•••		10.01	•••	2		80°0 1970	80. 10
$Fig. 7-1 \ (Cont.) \ Fig. 7-1 \ (Cont.) \ Fig. 7-1 \ (Cont.) \ Fig. 7-1 \ (Cont.) \ $	4.1.01.04.0	0H1.	10.0	•	•	•	10.0	•		•	13.0	
710144-115 1: 1: 2: 1:	4.1.01.04.(07 1.	14.0	0.	0.	0.	14.0	•	Ģ		19.0	19.
Fig. 7-1 Cite	2.1.01.04.1 0.0.00.04	12 1.	0 0 7 •	22.0	•••	• • •	0 (1 (22.0	22.0	10.0	E .	110.
Fig. 7-1 (Cont.)	9 9 99 99	1.2			1.1.1	- v • u	7 • V	י ע די ע די ע	1 a 1			
Fig. 7-1 (Cont.)1 1.0 0 0 0 0 0 0 10 4440700.441 1.0 0 0 0 0 0 10 <t< td=""><td>4.0.05.04.5</td><td>54 1.</td><td>0.</td><td>•</td><td>•</td><td>0</td><td>0</td><td>•</td><td>=</td><td>•</td><td></td><td>•••</td></t<>	4.0.05.04.5	54 1.	0.	•	•	0	0	•	=	•		•••
9,4,0,0,1,21 11 0 0 $2,0$ $46,0$ 0.0 $21,0$ $9,0,9,1,0,12$ $11,1$ $11,0$ $17,4$ $30,2$ $46,0$ 10 10 10 $21,0$ 10 $21,0$ 10 $21,0$ 10 $21,0$ <		411.	•	•	•••	0.	0.		0.	0.	B	=
Fig. 7-1 Cont. 1	4°4°01°04°1	51 1.	0.34	•	•	0,0	0.	•• ;	•	c .	8	1.
9:9:99:93:34 6: 0 1.2 51.0 52.4 51.0 52.4 51.0 75.0 97.0 9:9:99:95:35 2: 32:8 .0 1.2 51.0 52.4 51.0 75.0 75.0 9:9:99:95:35 2: 32:8 .0 1.2 51.0 63.4 51.0 75.0 101ALS 318:0 60:8 658:0 549.7 976:0 610:5 2549:8 1193:0 12525.7 6842:5 701ALS 318:0 60:8 658:0 549:7 976:0 610:5 2549:8 1193:0 12525.7 6842:5	0.0.00.00.0	- - - - - - - - - - - - - - - - - - -		•••		0.02 7 0 1			- C	0.000	о . в с	21.
9.9.99.35 2. 32.8 .0 .8 3.0 33.6 3.0 4.2 30.0 B .0 75.0 TOTALS 318.0 60.8 658.0 549.7 976.0 610.5 2549.8 1193.0 12525.7 6842.5 Fig. 7-1 (Cont.)	66 66 6 6	14		0.77	1.2	51.0	62.4	51.0	7, e + -	220.0	B C	-98-
ToTALS J18.0 60.8 658.0 549.7 976.0 610.5 2549.8 1193.0 12525.7 6842.5 Fig. 7-1 (Cont.)	· 66 • 66 • 6	35 2.	32.8	•	8	0.5	33.6	0.5	4.2	30.0		75.
Fig. 7-1 (Cont.)	TOTALS		318.0	60.8	658.0	549.7	976.0	610.5	2549.8	1193.0	12525.7	6842.
Fig. 7-1 (Cont.)		-										
Fig. 7-1 (Cont.)		1	;			1						
Fig. 7-1 (Cont.)												
Fig. 7–1 (Cont.)				-				1				
Fig. 7-1 (Cont.)							•	***				
Fig. 7-1 (Cont.)												
						7-1 T-1	Cont V	•		-		

.

7-13

,	, .							Vol.	Ι
		:							• •
		•	1	2	1		;	: :	
	: -			1	i	2			
1	,	,			•	· 1	. 1	•	
	•		1		•	2 1			
			÷						
				1					
									1
			-						i
			i i						:
TOT									
A PE A									÷
MAE 6 E YTI									
STA STA	z			T					1
	2210		;				,		
	2								
MA CRO	. HIS		÷						5
			·						
	2 2		:						1
	len 1								
-	IKE		1						nt.
28 1 1	EGU								CG :
JER SSI0	Σ Σ								, 1
SIM SI M	100K								-2
NO	ž j		;				:		្តំស្
ISSI 197	-1-1		•		:				म
M TIN			1	· · · · ·					
cou	241								
978A	Ę		:						
1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 -	NTS.								
ATE 1 5 LO			:						
ATH ATH THIS									
AT P	ກີພິ ເພື່								
N M MES	BAS		:						:
C PI SSIC UCAL UCAL									
L S L S L	ARE								
IENT AINA SSIO ABER									
NON NON	Ë Ë								
	:								;
	:			1		:			
		1		1					

1

LMSC-A847943

7.3.2 The Mayday Message

The Mayday message tells the mission planner that the computer has been instructed to use an experiment, system, or mode identifier that is not in its data storage. Under such conditions, the computer is programmed to seek out some other identifier for the missing experiment, system, or mode so as to terminate the run and insert the substitute set of input values into any subsequent computations. At the same time, the computer prints the Mayday message and identifies the mission experiment, system, or mode.

Usually such trouble is traced to a keypunch error in data input. Such is the case for the example in Fig. 7-1. The computer has announced itself incapable of finding a record of experiment 9.9.99.99.92 (see printout page following Mayday message). It then selected experiment 0.1.03.01.59 and substituted data from this experiment in the places required for 9.9.99.99.92. This substitute experiment appears in the R-02 traverse list (page 7-12, Fig. 7-1). A check of the list of experiments intended for traverse R-02 revealed that 9.9.99.99.92 should really have been 9.9.99.99.32, an experiment that is in the computer data storage.

Mayday messages are simple to diagnose. The important point is that all resulting calculations are subject to doubt. In this case, the computer inserted data for experiment 0.1.03.01.59 instead of 9.9.99.99.92 in traverse R-02 and continued computing. The resulting computation errors in this instance were small. The solution to this error is to correct the experiment identification on the appropriate mission input card.

7.3.3 Excessive Cargo Message

The next three comments (page 7-4, Fig. 7-1) say the cargo capability of LRV 2433-04 on traverse R-01 is excessive. These comments are printed out after successive computation subroutines are completed. An examination of the cargo

figures on page 7-8 shows that the required cargo exceeds the permitted cargo by almost 6,000 kg, with the bulk of the excess coming from minor scientific equipment. The breakdown of the vehicle cargo mass (page 7-11) lists two 6,000-kg masses as required. A check of the experiment layout for this mission revealed that the 6,000-kg mass required for experiment 4.1.01.04.14 (see page 7-11) was never intended to be transported as cargo and that the cargo override had been forgotten. The solution to this problem is to input a cargo override of 0.0 kg for this experiment on the appropriate computer input card for mission 28.

7.3.4 Decimal Point Errors

Two printout comments (page 7-4) indicate excessive cargo on traverse R-03. Referring to page 7-8 for a clue to the excessive cargo, it can be seen that the Earthreturn mass for traverse R-03 is quite high (1,193 kg) in comparison with the other traverses. Several large Earth-return masses are noted in the experiment summary on page 7-13. The largest one (500 kg) is associated with experiment 1.2.0.1.02.02. The same experiment on traverse R-01 (page 7-11) requires only 5 kg of Earth-return mass, a discrepancy of two ordersof magnitude between the two entries. Furthermore, the Phase 2 replications show the same discrepancy ratio. Checking the mission input card for the experiment showed that the decimal point was omitted in the input of replications for operation (Phase 2) of the experiment on traverse R-03. When this happened, the computer adjusted the input number to the far left of the field that was available for that number (three spaces). So, instead of an input of 1.0, the computer read 100.0. The solution to the problem is to punch in the decimal point as required.

7.3.5 Routine Statements

Whenever a mode is used that has a rover and a trailer (but no shelter), the computer recognizes two possibilities for supporting the crew – the rover alone or the rover-plus-trailer combination. The performance data on the rover-plus-trailer combina-tion is input to the computer under the trailer identification number (see paragraph

10.6 and 10.9). If there is any activity at a central landing point, the computer must decide which of the two entities is supporting the crew. To overcome this problem, the computer is programmed to accept the first trailer or rover identifier that it encounters in the mode input, and use the data from that system for any base-type calculations, and print out what it has done on the comments page output. It repeats the written notice during subsequent subroutines as it has done in the example in Fig. 7-1. No corrective action is required in these cases.

7.3.6 Excessive Earth-Return Mass Statement

Whenever there is insufficient capability in a mode for returning required mass to Earth, a statement is printed out as shown at the bottom of the comments page of Fig. 7-1. The example mission requires 1,790 kg; the mode supplies only 50 kg (see page 7-7). Some of the required mass is attributable to the missing decimal point discussed in paragraph 7.3.4. An examination of each element of Earth-return mass shows no other mistakes in information content.

The personnel transportation system (1124-01) of the mode is capable of transportating six men, but is being used at a three-man level; this allows an additional 426 kg of Earth-return capability. Furthermore, the available mass margin of that system can be traded off for an additional 310 kg of Earth-return capability. Thus, the maximum Earth-return mass can be as high as 786 kg (310+426+50).

These augmentations are not usually input with the mode data, since other mode usage may not require the maximum Earth-return capability. In any event, the required Earth-return payload for this example exceeds capability, and experiment redefinition or system redesign is in order.

7.3.7 Other Messages

If the mode staytime is exceeded, or more expendables are required, messages are printed in the comments section. In any case, it is advisable to check into every comment printed.

7.3.8 Second Iteration

All the foregoing changes in input were made to the sample mission-mode comparison. The comparison was rerun and the resulting computer printout is presented in Fig. 7-2. All problem messages, except Earth-return mass, have disappeared. The correction of mission input has reduced the amount of required Earth-return mass to 1,295 kg. The possible augmentation (paragraph 7.3.6) permits a total of 786 kg. Thus, the actual ratio is 1,295/786 = 1.65. However, because the accuracy of the Earth-return requirement given with the experiments was questionable, no further input changes were undertaken.

LOCKHEED MISSILES & SPACE COMPANY

MISSION-MODE MATCH COMMENTS PROGRAM AT MISSION-WODE MATCH COMMENTS CATION- PATH 13.5.4.5 TIME FOLLUWING ABOUT THIS MISSION-WODE MATCH COMMENTS THE FOLLUWING ABOUT THIS MISSION-WODE MATCH COMMENTS MORE THAN ONE ROVING VEHICLE CAN SUPPORT THE MODE CALCULATIONS HAVE REEN MADE USING THE PERFORMANCE EXCOUNTERED IN THE MODE LAUNCH DATA. MORE THAN ONE ROVING VEHICLE CAN SUPPORT THE MODE CALCULATIONS HAVE REEN MADE USING THE PERFORMANCE EXCOUNTERED IN THE MODE LAUNCH DATA. MORE THAN ONE ROVING VEHICLE CAN SUPPORT THE MODE CALCULATIONS HAVE REEN MADE USING THE PERFORMANCE EXCOUNTERED IN THE MODE LAUNCH DATA. MORE THAN ONE ROVING VEHICLE CAN SUPPORT THE MODE CALCULATIONS HAVE REEN MADE USING THE PERFORMANCE EXCOUNTERED IN THE MODE LAUNCH DATA. MORE THAN ONE ROVING VEHICLE CAN SUPPORT THE MODE CALCULATIONS HAVE REEN MADE USING THE PERFORMANCE EACCOUNTERED IN THE MODE LAUNCH DATA. MORE THAN ONE ROVING VEHICLE CAN SUPPORT THE MODE CALCULATIONS HAVE REEN MADE USING THE PERFORMANCE EACCOUNTERED IN THE MODE LAUNCH DATA. MORE THAN ONE ROVING VEHICLE CAN SUPPORT THE MODE CALCULATIONS HAVE REEN MADE USING THE PERFORMANCE EACCUNTERED IN THE MODE LAUNCH DATA. MORE THAN ONE ROVING VEHICLE CAN SUPPORT THE MODE CALCULATIONS HAVE REEN MADE USING THE PERFORMANCE EACCUNTERED IN THE MODE LAUNCH DATA. MORE THAN ONE ROVING VEHICLE CAN SUPPORT THE MODE CALCULATIONS HAVE REEN MADE USING THE PERFORMANCE EACCUNTERED IN THE MODE LAUNCH DATA. CALUATIONS HAVE REEN MADE USING THE PERFORMANCE	g. 7-2 Example Computer Printout – Mission-Mode Match No. 2
SCIENTIFIC PH MISSION LOCA NUMBER OF TIA MOULD YOU BELIEVE TH MOULE 503-30114-01	Fig.

LMSC-A847943 Vol. I

SUMMART OF MISSION-MODE MATCH CALCUL		PAGE 1	
IC PRUGHAM A7 ISSIUN MIX 02 MISSION NUMBER 28 MISSION START DATE= 1976A LOCATION- PATH 13.574.5 F TIMES AT THIS LOCATION COUNTING THIS MISSION- 1	MODE I.D. NUMB CREW LEVEL PERFORMING E MAX MODE STATT	CR_503-30144-01 TOTAL= 3 CXPERIMENTS= 3 LME_(DAYS)= 90.	
ING JELOW SUMMARIZES THE MISSION-MODE CALCULATION PERFOR	MED BY THE COMPUTER.		
ITEM -	CALCULATED VALUE	PERMITTED VALUE	RATIO
IAYTIME (DAYS) COR ALL ROVING TRAVERSES (DAYS) FOR ALL FLYING TRAVERSES (DAYS) FOR LEFLATING TRAVERSES (DAYS) COR LEPLOYMENT (DAYS)	72.6 45.3 0 21.3 6.0	•06	•806
WIOURS UTILIZED	5225.47	6480.	•806
SPACESUIT MATHOURS LATIFIC - ON ALL TRAVERSES (INPUT MULTIPLIER=2.00) LATIFIC - AT HASE (INPU) MULTIPLIER=2.00) Scientific - ON ALL TRAVERSES Scientific - AT UASE	1278.3 90.0 500.2 04.1 33.2		
SHLMTSLEEVE MANHOURS ENTIFIC - ON ALL TRAVERSES (INPUT MULTIPLIER=1.00) 1 ENTIFIC - AT MASE (INPUT MULTIPLIER=1.00) 4 Scientific - ON ALL TRAVERSES SCIENTFIC - AT BASE	3947.5 55.2 98.7 84.9		
(1) ENVANLES UFILIZED (KG)	6056.4	9438•	.642
EXPENDAULES USED AT THE BASE (KG) J FUN STSTEM SUPPORT DURING ALL TRAVERSES PEUN SUPPORT DURING ALL NON-TRAVERSE PERIODS - B PEUN PAULTER SUPPORT DURING ALL NON-TRAVERSE PERIODS - B	1240.0 •U 52.4	0.	
D FOR POWER ON BASE EXPERIMENTS A AMOUNT FOR SPACESUIT USE AT BASE (A AMOUNT FOR SPACESUIT USE ON TRAVERSES (A FACIAL OPERATIONS AND AND AND AGE FOR AT AND A AND AND AND AND AND AND AND AND A	11 0000 0000		
JUNING TRAVENSES USE USE USE USE USE USE USE USE USE	4816.4 14.4 2.0 00.0	7520.	•640
ESUPPLY EXPENDALES REQUIRED (KG)	0.00	0.	000.
	2		

LOCKHEED MISSILES & SPACE COMPANY

SUMMARY OF MI	ISSION-MODE MATCH CALCULATI	NO	PAGE 2		ج
SCIENTIFIC PROGRAM A7	SSION NUMBER 28	RODE I.D. NUMBER CREW LEVEL	R 503-30114-01 TOTAL= 3		-
NOMINAL MISSION START DATE= 1976A MISSION LOCATIOH- PATH 13.5.4.5 NUMBER OF TIMES AT THIS LOCATION COUNTING	I -MISSION- I	MAX MODE STAYTIN	VFEKIMENIS= 3 VE (DAYS)= 90.		:
THE LISTING DELOW SUMMARIZES THE MISSION-	-MODË CALCULATION PERFORMED	BY THE COMPUTER.	:		I
ITEM	J	ALCULATED VALUE	PERMITTED VALUE	RATIO	
TOTAL LEV PROPELLANT UTILIZED (KG) TOTAL LEV PROPELLANT RESUPPLIED (KG)	:	00	1030. 0.	000	:
TOTAL MASS WARGIN UTILIZED (KG) MASS OF MINON SCIENTIFIC EQUIPMENT MASS OF SPARE BACKPACKS MASS OF SPARE BACKPACKS	3. REQD 10. REQD	31749.9 31749.9 99.0 100.0	47640.	.671	
TOTAL EARTH RETURN MASS UTILIZED (KG) For UASE EXPERIMENTS For TRAVENSE EXPERIMENTS		1295+0 1295+0	50 .	25 . 90N	1
		· ·			•
	• • • •				
					•
:	· · · ·				·
A 19 A 19 A 19 A 19 A 19 A 19 A 19 A 19			-		1
					•
			a - mar - and - and - and - and - and - and - and - and - and -		
		· · · · · · · · · · · · · · · · · · ·			
				f	
		r vonneka din er a terre an en er a terre a men er er er er er er er er er er er er er			
	Fig. 7-1 (Cont.)				

۰,

-

LMSC-A847943 Vol. I

SCIENTIFIC PHOGHAM A7 MISSION HIX U2 MISSION		9	MODE I.D.	NUMBER 503-30114-01 Total = 3	
NOMINAL MISSION START DATE= 1978A			PERFORM MAX MODE S	IING EXPERIMENTS= 3 TAYTIME (DAYS)= 90.	
THE LISTING BELOW SUMMARIZES THE TRAVERSE CAL	LCULATIONS	PERFORMED	D BY THE COMPUTER		
TRAVERSE LUGNTIFIER Vehicle Joentifier	R-01 2433-04	R-02 2433-04	R-03 2433-04		
MINIMUM CREW REQUIRED VEHICLE CREM LEVEL CREW AT EASE DUMING TRAVERSE	NMO	400	N FC O I		
THAVERSE RANGE (KM) Maximum vehicle capability (km)	0°0002 0°0002	350.0 7000.0	1150.0 7000.0	:	
PEQUIRED EXCURSION TIME (DAYS) MAXIMUM VEHICLE CAPAGILITY (DAYS)	19.4	4.0	20.9 400.0		
NUMBER OF STOPS	10.0	10.0	10.0		
MASS OF MINOR SCIENTIFIC FOULPMENT (KG) EANTH RETURN MASS REOULHED (KG) MASS OF SPALESUIT REPLACEMENTS (KG) MASS OF BACKPACK REPLACEMENTS (KG)	6764.5 452.0 10.0	6247.5 145.0 .0	6842.5 698.0 .0 10.0		
TOTAL CARGO MASS REQUINED (KG) Maximum vehicle capability (KG)	7226.5 7660.0	6392.5 7660.0	7550.5		
SCIENTIFIC MAN-HOURS ON TRAVERSE SHIRTSLEEVE SPACESUIT	402.8 90.8 312.0	64•4 3•6 60•8	378.8 60.8 318.0		
NOMSCIENTIFIC MAN-HOURS ON TRAVERSE SHIRTSLEEVE SPACESUIT	997.2 952.5 44.7	268.1 276.8 11.3	1127.4 1079.3 48.1		
SCIENTIFIC MAN-HOURS REQUIRED AT BASE Shirtsleeve Spacesuit	197.8 115.8 82.0	37.5 31.3 6.2	613.7 351.7 262.0		:
NONSCIENTIFIC_MAN_HOURS_REGUIRED_AI_BASE SHIRTSLEEVE SPACESUIT	0.00	••••	0		
TRAVERSE EXPENUARLES REQUIRED (KG) MAXIMUM EXPENDAULES PERMITTED (KG)	1480.7 7520.0	450.1 7520.0	2483.6 7520.0		
BASE EXPENDABLES FOR SYSTEM SUPPORT	•0	0•	• 0		:
DELIA, V. REQUIRED. OF LEV (MZS) SPECIFIC IMPULSE (MZS)	0.	0.	0.		
LEV PROPELLANT REGULARY	•••		•••		
MAXIMUM PROPELLANT PERMITTED (KG)	•	•••	0.		

Fig. 7-2 (Cont.)

*	_	I	!	•	1	. 1			1	!	!		•			
DN	MODE 1.D. NUMBER 503-30114-01 CREW LEVEL DITAL= 3	PERFORMING EXPERIMENTS= 3 Max mode staytime (days)= 90.	ATIONS FOR THE TRAVERSES REQUIRED ON	R-03 +33-04	9483.6 1520.0	652.7 796.9 851.0 183.1	ING VEHICLE EQUALS 0' ANY CALCULATED EXPENDABLES 16 Requirement of any other expendables indicates 168' or koving vehicle acting as a shelter, and	BE EQUAL TO 0.		a na mana mana mpinanananan para mananan a na ana ana ana ana ana ana a						•
H-MODE MATCH CALCULATIC	NUMBER 28		HE EXPENDABLES CALCULA	R-01 R-02 2433-04 2433-04 24	1480.7 450.1 2 7520.0 7520.0	606.7 152.8 29.6 2.3 666.0 259.0 178.4 36.0	ED FOR A FLYING OR ROVI Spacesuit Manhours (Ti Arged Against the Shel	ERSE ARE CONSIDERED TO				ny ang nama na manang a la panga ang mang mang mang mang mang mang m				Fig. 7–2 (Cont.)
SUMMARY OF MISSION	SCIENTIFIC PROGRAM A7 MISSIUN MIX 02	NOMINAL MISSION START DATE= 1978A	THE FOLLOWING LISTING SUMMARIZES DETAILS OF TI PAGE 1.	TRAVENSE IULNTIFIER VEHICLE IDENTIFIER	TRAVERSE EAPENUARLES REGUIRED (KG) MAXIMUM EXPENDABLES PERMITTED (KG)	EXPENDAULES USAGE BREARDCWA (KG) VEHICLE SUPPORT JASED ON DAILY USE EXPERIMENT ENERGY FOR TRAVELING FOR STOPS EXTRA FOR SPACESUIT MANHOURS	NOTE WHEN THE MAXIMUM EXPENDABLES PERMITTE REGUIRED ON A TRAVERSE SHOULD BE FOR AN ERROR IN INPUT DATA). THESE ARE CH	THE EXPENDABLES REGUIRED ON THE TRAVE	· · ·		•			· · · · · · · · · · · · · · · · · · ·		I

7-23

	SUMMARY	OF MISSION-MODE MATCH	CALCULATION PAGE 5	<
	SCIENTIFIC PROGRAM A7 MISSIUN MIX 02	MISSION NUMBER 28	MODE I.D. NUMBER 503_30114-01 CREW LEVEL	- ;
	NOMINAL MISSION START DATE≑ 1978A	· · · · · · · · · · · · · · · · · · ·	PERFORMING EXPERIMENTS= 3 MAX_MODE STATTIME (DAYS)= 90.	
	THE FOLLOWING IS A VEHICLE UV VEHIC For the traverses reported on page	LE SUMMARY OF THE DETA 1.	VILS OF EXPENDABLES AND PROPELLANT CALCULATIONS	
	VEHICLE IDENTIFIER	2433-04		
	TOTAL EXPENUABLES REGUIRED MAXINUM MASS OF EXPENDALES PERMITT MASS OF RESUPPLY EXPENUABLES REQUIR	4814.4 ED_(KG) 7520.0 ED_(KG) .0		F
	EXPENDAULES USAGE BHEAKDOWN (KG) VEHICLE SUPPORT BASED ON DAILY US EXPERIMENT ENERGY FOR TRAVELING	E 1412•1 828•8 1776•0		
	FUR STUPS EXTRA FOR SPACESUIT WANHGURS USEU FOR SPECIAL OPERATIONS UNUSAULE (BOILOFF, ULLAGE' ETC.)	397.4 4.00.0 4.004		, -
	TOTAL MASS OF PROPELLANT REAUTRED (MAXIMUM MASS OF PROPELLANT PERMITTE MASS OF RESUPPLY PROPELLANT REGUIRE	KG) •0 · 0 · 0 · 0 · 0 · 0 · 0 · 0 · 0 · 0		•
: : : :	NOTE IF THE MAXIMUM EXPENDABLES VEPLICLE ARE SUPPLIED BY THE Regulated and the Mass of Re	PERMITTED ON A VEHICLE SHELTER, ON ROVING VE SUPPLY EXPENDABLES SHO	E EQUAL OF THEN ALL OF THE EXPENDABLES REQUIRED AY THAT CHICLE ACTING AS A SHELTER, AND THE TOTAL EXPENDABLES OULD EQUAL OF IF NOT, THERE IS AN ERROR IN THE INPUT DATA.	

				-
				ļ
An Andre		•	· · · · · · · · · · · · · · · · · · ·	:
			· · · · · · · · · · · · · · · · · · ·	i
		Fig. 7–2 (C	Cont.)	

LOCKHEED MISSILES & SPACE COMPANY

Montial registions furth registions for this mission MR mode STATTRE (During this mission) The Lusting definits of this mission The Lusting definits for this mission The Lusting definits of this mission The Lusting definits of the surveyer extention The Lusting definits of this mission The Lusting definits of this mission The Lusting definits markets greenee The Lusting definition of this mission The Lusting definition of this mission The Lusting definition Extending and the markets greenee The Lusting definition The Lusting definition The Lusting definition Extending and the lust set of this mission The Lusting definition The Lusting definition The Lusting definition Extending definition Extending definition The Lusting definition The Lusting definition Extending definition Extending definition The Lusting definition The Lusting definition Extending definition Extending definition The Lusting definition The Lusting definition Extending definition Extending definition The Lusting definition The Lusting definition Extending definition Extending definition The Lusting definition The Lusting definition Extending definit	International distribution International distribution <th< th=""><th>Modeline Tission Similaries Providential Figures Manual Model Similaries Providential Providenti Providentia Providential Providential Providential Providential</th><th>SCIENTIFIC PH</th><th>NUCHAM A</th><th>02</th><th>!</th><th>N NOISSIN</th><th>UMBER 28</th><th></th><th>CRE</th><th>W LEVEL</th><th>T0</th><th>TALE 3</th><th></th><th></th></th<>	Modeline Tission Similaries Providential Figures Manual Model Similaries Providential Providenti Providentia Providential Providential Providential Providential	SCIENTIFIC PH	NUCHAM A	02	!	N NOISSIN	UMBER 28		CRE	W LEVEL	T0	TALE 3		
Iffet List Jie Jelluk Suwerkieze Frie Scientifier Promotion Resultations Theie, with on Euriphican Expensions Theie, with on Euriphican Towards Proprint Their Ectuation Expensions Proprint Their Ectuation	Iffe. L151166 active submetalizes the SciEntific Prodental Reformance in the L151166 active submetalizes. Exercisions Inter L151166 active submetalizes. The SciEntific Prodental Reformance in the SciEntific Product Reference in the SciEntific Product Produ	Iff. L151166 actus SumeralZES ML. SCIENTFIC Prodenk #FOUNEHENTS For fils MISS101. TUER. ANG. NO FL716 TANNESE. EVERTINGS NEE ENTITIE FUNCTION TO THE CONTRACT TANNESE FOR MUSS. TUER. ANG. NO FL716 TANNESE. EVERTINGS NEE ENTITIE FUNCTION TO THE CONTRACT TANNESE FOR MUSS. TUER. ANG. NO FL716 TANNESE. EVERTINGS NEE ENTITIE FUNCTION TO THE CONTRACT TANNESE FOR MUSS. TUER. ANG. NO FL716 TANNESE. A set of the contract of the contract tank tank to the contract of the contrac	NOMINAL MISS. Mission Loca NUMBER OF TIA	ION STAR Lion- Pa Ges at t	T DATE= 1 TH 13.5 HIS LOCAT	976A •4•5 Ion count	I SIHI DUI	MISSION-	, -	MAX	PERFORMIN MODE STA	G EXPERIME YTIME (DAY	s)= 90.	:	
THEM. AND EXPERIMENT EXPERIMENT THEN THEN THEN THEN THEN THEN THEN T	Theile, Alle for Experiments Theile, Alle for Experiments Explore a for the Expension on that were a formation where the point of the point	Triel. All: No ELTID TANGESE EXPERIMENTS Triel. All: No ELTID TANGESE EXPERIMENTS Trie Following EVENTMENT EVENTMENT<	The LISTING	SELOW SU	MMENIZES	THE SCIEN	TIFIC PRO	GRAM REQU	IREMENTS	OR THIS	.NOISSIM	And a second of the second second second second second second second second second second second second second		i	ı
The Following Extensioned on Taxterse P-01 The Following Carton Werese P-01 The Following Extensioned on Taxterse P-01 Contract Carton Taxterse P-01 C	The Following Exteriments were privated on Traverse F-01 The Following Exteriments Following Exterimats Following Exteriments	The Following Experiments were performed on Traverse F-01 The Following Experiments Entropy for the formed on Traverse F-01 Lowelds RePLIC Experiments Experiments Experiments Experiments Experiments Numbers RePLIC Experiments Experimates Experiments Experiments	THERE ARE NO	FLYING	TRAVERSE	EXPERIMEN	TS			:		:			
WHELTRE: FARL NELT FMASS Early and the second system EXPLIMENT FERIOR EXPLICATION FERIOR EXPLICATION FERIOR EXPLICATION FERIOR EXPLICATION FERIOR EXPLICATION FERIOR EXPLICATION FERIOR <	District Trans Final Contract Final C	EXERTING FIRE Contract Exercise Contract Exercise <th< td=""><td>THE FOLLUWING</td><td>5 EXPERI</td><td>MENTS WERE</td><td>E PERFORM</td><td>ED ON TRA</td><td>VERSE R-0</td><td>1</td><td></td><td></td><td></td><td></td><td></td><td></td></th<>	THE FOLLUWING	5 EXPERI	MENTS WERE	E PERFORM	ED ON TRA	VERSE R-0	1						
9,0,0,0,0,0,12 1,0 0,0 1,0 0,0 1,0 0,0	9,0,0,0,0,0,0,0,0 1,1 1,0 <th>$\begin{array}{cccccccccccccccccccccccccccccccccccc$</th> <th>EXPERIMENT</th> <th>PHASE 2 REPLIC</th> <th>(TRAVE SPSU</th> <th>M A N H ERSE) SHSL</th> <th>0 U R S (AT E SPSU</th> <th>R E © Ú 3ASE) SHSL</th> <th>I R E D (10 SPSU</th> <th>TAL)</th> <th>EQUIPMENT ENERGY (KW-HR)</th> <th>EAPTH RETURN MASS (KG)</th> <th>EQUIPMENT MASS CODE KG</th> <th>VEI CUE</th> <th>HICLE Argo Kg</th>	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	EXPERIMENT	PHASE 2 REPLIC	(TRAVE SPSU	M A N H ERSE) SHSL	0 U R S (AT E SPSU	R E © Ú 3ASE) SHSL	I R E D (10 SPSU	TAL)	EQUIPMENT ENERGY (KW-HR)	EAPTH RETURN MASS (KG)	EQUIPMENT MASS CODE KG	VEI CUE	HICLE Argo Kg
2:10:10:10:10 1: 17:6 2: 2: 1: 1:0:0 1:0:	Zirungrout III III IIII IIII IIIII IIIII IIIIII IIIIIII IIIIIIII IIIIIIII IIIIIIII IIIIIIIII IIIIIIIII IIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII	29,99,99,39 11 17,6 22,0 22,0 10,0	9.9.99.99.12 9.0.05.02.01	ы.	0.4 0.9	•••	14.6 •2	0.6 0.	18.6 •8	0•6 •	1.2	•••	24.	е С О	0.05
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	11.01.01.01.01 11.01.01.01 10.01.01	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	9.9.99.99.36 2.1.01.04.12		17.6	4.4 22.0	2.0	5.0	19.6	7.3	2.1	7.0	61.		
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	2.1.02.03.07		0	••	4 •8	?	8	-	5.0	1-0	6		••
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	9.9.99.99.10 1.1.01.01.01		22.0	•••	7.4	5.0	22.0	2•0	•-	0.0	150.	е a с с	•
7:4:49:49:21 1: 6:0 25:0 1: 6:0 25:0 1: 6:0 25:0 1: 6:0 1: 0:0 1: 0:0 1: 0:0 1: 0:0 1: 0:0 1: 0:0 1: 0:0 0:0 0:0 0:0 0:0 0:0 0:0 0:0 0:0 0:0 0:0 0:0 0:0 0:0 0:0	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y y <t< td=""><td>1.2.01.02.02</td><td></td><td>0</td><td>•</td><td>4•€</td><td>2.0</td><td>9 1</td><td>0</td><td>t. 0</td><td>0.0</td><td>290.</td><td></td><td>•••</td></t<>	1.2.01.02.02		0	•	4 •€	2.0	9 1	0	t. 0	0.0	290.		•••
21101.01.011 100 0	21101.00010 0 <td< td=""><td>2110104014 11 100 <</td><td>9.9.94.99.31 3.4.01.04.02</td><td></td><td>66.0 Ú</td><td>25.0 4.5</td><td>•••</td><td>•••</td><td>66.0</td><td>25.0 4.5</td><td>1.6</td><td>30•U</td><td>1.</td><td>80 : :</td><td>••</td></td<>	2110104014 11 100 <	9.9.94.99.31 3.4.01.04.02		66.0 Ú	25.0 4.5	•••	•••	66.0	25.0 4.5	1.6	30•U	1.	80 : :	• •
\$\$1,00000000000000000000000000000000000	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	4.1.01.04.14		10.0	0	0	-	10.0	•	•		6000	• •	
9:0:9:0:201 1 0 <td< td=""><td>9:0:0:0:2:01 1 0 <t< td=""><td>W10.05.02.01 I 0.0</td><td>2.1.02.03.07 9.9.99.99.12</td><td></td><td>•••</td><td>••</td><td>0 t</td><td></td><td>3 U</td><td>0.0</td><td>2°0</td><td>C - 1</td><td>a: a</td><td>c c</td><td>0°77</td></t<></td></td<>	9:0:0:0:2:01 1 0 <t< td=""><td>W10.05.02.01 I 0.0</td><td>2.1.02.03.07 9.9.99.99.12</td><td></td><td>•••</td><td>••</td><td>0 t</td><td></td><td>3 U</td><td>0.0</td><td>2°0</td><td>C - 1</td><td>a: a</td><td>c c</td><td>0°77</td></t<>	W10.05.02.01 I 0.0	2.1.02.03.07 9.9.99.99.12		•••	••	0 t		3 U	0.0	2°0	C - 1	a: a	c c	0°77
441.00.01 0 0 10.0 0 <t< td=""><td>94:191.04114 1: 22:0 0</td><td>9:1:01:01:01 1: 1:0:0 0 1:0:0 0 1:0:0 0 1:0:0<!--</td--><td>9.0.05.02.01</td><td></td><td></td><td>•</td><td>~</td><td></td><td>2</td><td>0</td><td>••</td><td></td><td>00</td><td>- 0</td><td></td></td></t<>	94:191.04114 1: 22:0 0	9:1:01:01:01 1: 1:0:0 0 1:0:0 0 1:0:0 0 1:0:0 </td <td>9.0.05.02.01</td> <td></td> <td></td> <td>•</td> <td>~</td> <td></td> <td>2</td> <td>0</td> <td>••</td> <td></td> <td>00</td> <td>- 0</td> <td></td>	9.0.05.02.01			•	~		2	0	••		00	- 0	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	2::::::::::::::::::::::::::::::::::::	4.1.01.04.14 9.9.99.99.10		10.0	0.0	••	•	10.0	•		•	6000.		6000.0
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	21000012 10000 2200 2000 $21000000000000000000000000000000000000$	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	2.1.02.03.07	ב. יי	0	20	t • t	•		•	2.0	1.0			
9, 9, 9, 9, 9, 12 $3, 3, 10$ $1, 10$ <t< td=""><td>9.9.99-99:12 3. 3.0 12.6 8.4 15.6 8.4 15.6 8.4 15.6 8.4 15.6 8.4 15.6 8.4 15.6 8.4 15.6 8.4 15.6 8.4 15.6 8.4 15.6 8.4 15.6 14.0 1</td><td>9.9.99.99.12 3. 5.0 10<td>2.1.01.04.12</td><td>-</td><td>0°£</td><td>20.0</td><td>0 °</td><td>0 r (</td><td>0.0</td><td>20.0</td><td>22.0</td><td>10.0</td><td>6 (</td><td>0</td><td>110.0</td></td></t<>	9.9.99-99:12 3. 3.0 12.6 8.4 15.6 8.4 15.6 8.4 15.6 8.4 15.6 8.4 15.6 8.4 15.6 8.4 15.6 8.4 15.6 8.4 15.6 8.4 15.6 8.4 15.6 14.0 1	9.9.99.99.12 3. 5.0 10 <td>2.1.01.04.12</td> <td>-</td> <td>0°£</td> <td>20.0</td> <td>0 °</td> <td>0 r (</td> <td>0.0</td> <td>20.0</td> <td>22.0</td> <td>10.0</td> <td>6 (</td> <td>0</td> <td>110.0</td>	2.1.01.04.12	-	0°£	20.0	0 °	0 r (0.0	20.0	22.0	10.0	6 (0	110.0
4,0,00,04,04 1 .0 .0 .0 .0 .0 .0 .0 3.5 3.5 $4,4,00,03,441$ 1 .0 .0 .0 .0 .0 .0 4.0 4.0 $4,4,00,03,441$ 1 .0 .0 .0 .0 .0 1.0 4.0	4,0,00,04,54 1. .0 .0 .0 .0 .0 .0 3.5 3.5 $4,10,05,03,41$ 1 .0 .0 .0 .0 .0 1.0 4.0 $4,10,05,03,41$ 1 .0 .0 .0 .0 .0 1.0 4.0 $4,4,4,05,03,41$ 1 .0 .0 .0 .0 .0 1.0 4.0 $4,4,4,05,03,41$ 1 .0 .0 .0 .0 .0 1.0	$\mu_10.08.04.54$ 1 .0 </td <td>9.9.99.99.12</td> <td>- m</td> <td>3.0</td> <td>v ⊃</td> <td>12.6</td> <td>- + • 0 •</td> <td>15.6</td> <td>9.4 9.4</td> <td>1.2</td> <td></td> <td></td> <td></td> <td>61.0</td>	9.9.99.99.12	- m	3.0	v ⊃	12.6	- + • 0 •	15.6	9.4 9.4	1.2				61.0
444.07.00.011 1.0 0.0 <td>4.4.07.0451 $1.$ $0.$ $0.$ $0.$ $0.$ $1.$ lt;</td> <td>44440704511 10 00 00 00 00 10 lt;</td> <td>4.0.08.04.54</td> <td>.</td> <td>.</td> <td>0.0</td> <td>e 6</td> <td>0,0</td> <td>•</td> <td>•</td> <td></td> <td>0.</td> <td>•е</td> <td>5</td> <td>3.5</td>	4.4.07.0451 $1.$ $0.$ $0.$ $0.$ $0.$ $1.$ <	44440704511 10 00 00 00 00 10 <	4.0.08.04.54	.	.	0.0	e 6	0,0	•	•		0.	•е	5	3.5
9.9.99.99.32 91. 36.4 1.5 .0 10.2 36.4 19.7 4.7 .0 21.0 21.0 9.9.99.33 40. 0 11.2 15.8 14.7 15.6 25.9 3.9 230.0 86.0 86.0 86.0 9.9.99.99.34 5. 11.2 15.8 14.7 15.6 53.0 86.0 86.0 86.0 86.0 86.0 86.0 75.0 9.9.99.99.35 3.1 49.6 5.3 4.5.0 75.0 75.0 75.0 75.0 75.0 75.0 75.0	9.9.99.99.32 91. 36.4 1.5 .0 18.2 36.4 19.7 4.7 .0 210 210 9.9.99.33 45. 0 11.2 15.8 14.7 15.6 25.9 3.9 230.0 86.0 86.0 90.9 9.99.99.34 5. 0 11.2 1.8 14.7 15.6 53.4 8.5 6.3 45.0 75.0 75.0 75.0 10.0 57.0 75.0 75.0 75.0 75.0 75.0 75.0 7	9.9.99.99.32 91. 36.4 1.5 .0 18.2 36.4 19.7 4.7 200 21.0 21.0 21.0 21.0 20.0 86.0 86.0 9.99.39 4.5 1.0 10.0 57.0 86.0 86.0 75.0 75.0 75.0 75.0 75.0 75.0 75.0 1.0 10.0 1.0 10.0 10.0 10.0 10.0 10.	4.4.07.04.51	• •			•••		•				- t		0 0 7 -
9.9.99.93 45 . 0 14.7 15.6 14.7 15.6 25.9 3.9 230.0 86.0 87.0 77.0	9.9.99.33 45 . 0 14.7 15.6 14.7 15.6 25.9 3.9 230.0 86.0 86.0 86.0 86.0 86.0 86.0 86.0 86.0 86.0 86.0 86.0 86.0 86.0 86.0 86.0 86.0 86.0 86.0 87.0 87.0 87.0 87.0 75.0 75.0 75.0 75.0 75.0 75.0 75.0 75.0 75.0 75.0 75.0 75.0 75.0 77.0 75.0	9-999933 4b. 0 11.2 15.8 14.7 15.6 25.9 3.9 230.0 86.0 86.0 9.999934 5. 51.2 .0 1.0 1.0 57.0 57.0 57.0 75.0 9.999934 5. 1.0 1.0 1.0 57.0 57.0 75.0 75.0 9.999934 5. 1.0 1.0 1.0 1.0 75.0 75.0 9.999934 5. 1.4.6 5.3.4 0.5 5.3.4 0.5 57.0 75.0 9.999934 5. 1.0 1.0 1.0 1.0 75.0 75.0 101AL5 312.0 90.8 82.0 115.8 394.0 206.6 80.1 452.0 13264.2 6764.5 101AL5 312.0 90.8 82.0 115.8 394.0 206.6 80.1 452.0 13264.2 6764.5	9.9.99.99.32	91.	36.4	1.5	0.	18.2	36.4	19.7	4.7	•	21.1	. c	21.0
7:7:77:77:00 57:0 57:0 57:0 57:0 9:9:99:99:35 3: 49:6 .0 3.8 8.5 53:4 8.5 6.3 45:0 75:0 75:0 312:0 90.8 82:0 115.8 394.0 206.6 80.1 452.0 13264.2 6764.5	7:7:7:7:7:7:0 5:0 5:0 5:0 57:0 57:0 9:9:99:99:35 3: 49:6 .0 3.8 8:5 53:4 8.5 6.3 45:0 75:0 101ALS 312.0 90.8 82.0 115.8 394.0 206.6 80.1 452.0 13264.2 6764.5	ToTALS 312.0 90.8 82.0 115.8 394.0 206.6 80.1 452.0 75.0 75.0 ToTALS 312.0 90.8 82.0 115.8 394.0 206.6 80.1 452.0 75.0 75.0	9.9.99.99.33	• 0 -	•	11.2	15.8	14.7	15.8	25.9	6°2	230.n	96.	0	86.0
T01ALS 312.0 90.8 82.0 115.8 394.0 206.6 80.1 452.0 13264.2 6764.5	T01ALS 312.0 90.8 82.0 115.8 394.0 206.6 80.1 452.0 13264.2 6764.5	T01ALS 312.0 90.8 82.0 115.8 394.0 206.6 80.1 452.0 13264.2 6764.5	9°°66°66°6°6	n n	9.64	•••	3•1 3•8	8.5	53.4	8•5	6.3	100.0	57. 75.		. 57.0 75.0
			TOTALS		312.0	8•06	82.0	115.8	394.0	206.6	80.1	452.0	13264 .	N	6764.5
										-					
										1		;		•	
				:	:	•	÷	:	¥	-				•	

LMSC-A847943 Vol. I

*	1	1	I ;	,	1	r	:	r	
	s	·			VEHICLE CARGO ODE KG	6000.0 3.5 4.0	21.0 21.0 57.0 75.0	6247.5	,
7	14-01	SH 3 90.	- k	:	EQUIPMENT MASS CODE KG C	6000.0 B	8 8 8 8	6000+0	
PAGE	413ER 503-301 TOTA	G EXPERIMENT YTIME (DAYS)			EARTH RETURN MASS (KG)		90.0 90.0 40.0	145.0	
	E 1.D. NU	PERFORMIN MODE STA	•NOISSIM		EGUIPMENT ENERGY (KW-HR)	0 4 0	2. F	6.2	
NOI	MOD	MAX	DR THIS	1	اللہ اللہ اللہ اللہ اللہ اللہ اللہ اللہ	000	1.5	34 • 9.	
CALCULAT	:	1	REMENTS F	:	I R E D _(TOT, SPSU	0000	14.0 5.4 20.8 16.8	67.0	(Cont.)
ODE MATCH	4RER 28	-ivolssi	AM REQUIN	ERSE P-U2	REOU ASE) SHSL	000	17.0 5.8 17.0	31+3	ig. 7-2
W-NOISSIN	ISSION NUI	NG THIS M	IFIC PROG	D. ON, TRAVI	OURS 	c o o o	20444	6•2	Æ
MARY OF	ž	784	HE SCIENT	PERFORME	M A N H RSETT		0.000	3.6	
SU	2	DATE= 19 H I3.51 IS LOCATI	MERIZES TI	ENTS WERE	<u> </u>	1000	14.0 20.4 16.4		
•	HOGHAM A7	ION START TION- PAT WES AT TH	SELOW SUM	G_EXPERIM	PHASE 2 REPLIC		35. 35. 18. 2.		
	SCIENTIFIC P	"ISSION LOCAT	THE LISTING E	THE FOLLOWING	EXPERIMENT	4.1.01.04.14 4.0.09.04.54 4.1.05.03.41	52.99.99.99.92 52.99.99.99.9 45.99.99.99.9 45.99.99.99.9 55.99.99.99.99	TOLALS	
•						- - -			

Michaeling Mission Status Mission Status Mission Status Mission Status Mission Mission Status Mission Status </th <th>Michaeling Mission Status Mission Status Mission Status Mission Status Mission Status Status Mission Status Mission Status Mission Status Mission Status Mission Status Mission Statu</th> <th>THE FOLLOWING AT AT ALSEION WITH AT ALSEION STATT DATE NOMINAL MISSION STATT DATE ALSEION LOCATION- PATH I3. NUMBER OF TIMES AT THIS LOCA THE LISTING BELOW SUMMERIZES THE LISTING BELOW SUMMERIZES THE FOLLOWING EAPERIMENT PHASE 2 ANNUMER REPLIC (TRA</th> <th>ι <u>ν</u></th> <th>IN NOISSI</th> <th>UMRER 28</th> <th></th> <th></th> <th><u>й</u> - 0 - 1 - 1</th> <th>MBER 503-3</th> <th>0114-01</th> <th></th> <th></th>	Michaeling Mission Status Mission Status Mission Status Mission Status Mission Status Status Mission Status Mission Status Mission Status Mission Status Mission Status Mission Statu	THE FOLLOWING AT AT ALSEION WITH AT ALSEION STATT DATE NOMINAL MISSION STATT DATE ALSEION LOCATION- PATH I3. NUMBER OF TIMES AT THIS LOCA THE LISTING BELOW SUMMERIZES THE LISTING BELOW SUMMERIZES THE FOLLOWING EAPERIMENT PHASE 2 ANNUMER REPLIC (TRA	ι <u>ν</u>	IN NOISSI	UMRER 28			<u>й</u> - 0 - 1 - 1	MBER 503-3	0114-01		
Instantion for the first intermentation of the first intermentatinterect intermentation of the first intermenta	111.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.	NOMINAL MISSION START DATE MISSION LOCATION- PATH 13. NUMBER OF TIMES AT THIS LOCA THE LISTING BELOW SUMMERIZES THE FOLLOWING EAPERIMENTS WE EXPERIMENT PHASE 2 (TRA					CKE	W LEVEL Devendant	10 10 10	TAL= 3		
THE LISTING LEVENTERS THE SCENTERS PROME REQUIREENTS FOR THIS MISSION. EQUIPMENT MEET FREE Provided on TAXAFINS FOR THIS MISSION. EQUIPMENT MEET FREE Provided on TAXAFINS FOR THIS MISSION. EQUIPMENT MEET FREE Provided on TAXAFINS FOR THIS MISSION. EQUIPMENT MEET FREE Provided on TAXAFINS FOR THIS MISSION. EQUIPMENT MEET FREE Provided on TAXAFINS FOR THIS MISSION. EQUIPMENT MEET FREE PROVIDED on TAXAFINS FOR THIS MISSION. EQUIPMENT MEET FREE PROVIDED on TAXAFINS FOR THIS MISSION. EQUIPMENT MEET FREE PROVIDED on TAXAFINS FOR THIS MISSION. EQUIPMENT MEET FREE PROVIDED on TAXAFINS FOR THIS MISSION. EQUIPMENT MEET FREE PROVIDED on TAXAFINS FOR THIS MISSION. EQUIPMENT MEET FREE PROVIDED on TAXAFINS FOR THIS MISSION. EQUIPMENT MEET FREE PROVIDED on TAXAFINS FOR THIS MISSION. EQUIPMENT MEET FREE PROVIDED on TAXAFINS FOR THIS MISSION. EQUIPMENT FREE PROVIDED ON TAXAFINA FREE PROVIDED ON TAXAFINA FREE PROVIDED ON TAXAFINA FREE PROVIDED FROVED FREE PROVIDED FREE PROVIDED FREE PROVID	THE LISTING JELME STREET THE SCENTERE PROGRAM REQUIREERTS FOR THIS MISSION. ENERGY MEANING WEEK PERFORMED ON TAXAFERS F-03 THE FELLONTING EXPERIMENTS WEEK PERFORMED ON TAXAFERS F-03 ENERGY MEANING WEEK PERFORMED ON TAXAFERS F-03 ENERGY MEANING WEEK PERFORMED ON TAXAFERS F-03 THE FELLONTING EXPERIMENTS WEEK PERFORMED ON TAXAFERS F-03 ENERGY MEANING WEEK PERFORMED ON TAXAFERS F-03 ENERGY MEANING WEEK PERFORMED ON TAXAFERS F-03 ANDER F SPINEL C	THE LISTING JELOW SUMMEKIZES The Following Eaperiments We Experiment Phase 2 Mighter Replic (Tra	1976A 5+4+5 TION COUNTIN	IG THIS N	-NOISSIM	: -	MAX	MODE STA	VTIME (DAY	• 06 = (S		
THE FALLONING EXCEPTIONED ON TAVERSE P-03 THE FALLONING EXC	The Fallowing Extendient on TANETSE Frequencies Control (Control (Contro) (Contro) (Control (Control (Control (Control (Contro)	THE FOLLOWING EAPERIMENTS WE EXPERIMENT PHASE 2 MINNHER REPLIC (TRA	THE SCIENT	IFIC PROG	GRAM REQUI	REMENTS F	OR THIS	•NOISSIM				
Christiking Energy Contract Contract <thcontract< th=""></thcontract<>	Christikar Effect Construct Effect Construct Con	EXPERIMENT PHASE 2 NUMMAR REPLIC (TRA	RE PERFORME	ON TRA	VERSE P-03					*		
SPU SISL SISL <ths< td=""><td>SPU SISL SPU SPU SISL SPU SISL SPU SISL SPU SPU SISL SPU <th< td=""><td></td><td>N A N H (</td><td>0 U R S (AT I</td><td>REGU BASE)</td><td>I R E U (70]</td><td>()</td><td>EQUIPMENT ENERGY</td><td>EARTH RETURN</td><td>EQUIPMENT MASS</td><td>CAL</td><td>ICLE 860</td></th<></td></ths<>	SPU SISL SPU SPU SISL SPU SISL SPU SISL SPU SPU SISL SPU SPU <th< td=""><td></td><td>N A N H (</td><td>0 U R S (AT I</td><td>REGU BASE)</td><td>I R E U (70]</td><td>()</td><td>EQUIPMENT ENERGY</td><td>EARTH RETURN</td><td>EQUIPMENT MASS</td><td>CAL</td><td>ICLE 860</td></th<>		N A N H (0 U R S (AT I	REGU BASE)	I R E U (70]	()	EQUIPMENT ENERGY	EARTH RETURN	EQUIPMENT MASS	CAL	ICLE 860
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	SPSU SPSU	SHSL	SPSU	SHSL	SPSU	SHSL	(KW-HIR)	MASS (KG)	CODE KG	CODE	¥G
		1.1.01.01.01	0.	7.4	2.0	7.4	2.0	•	2.0	150.	بنا د	••
		9.9.99.99.31 I. 66.0	25.0	4 C	0.5	4•6 66•0	25.0	4•0 1•0	0.05	- 062 - 1	с с 0 р	•
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	4.1.01.04.14		•••		10.0		ç		6000-I	- e	•••
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		2.1.02.03.08 10 9.9.99.99.12 10.11	0.0	170.0	213.0	170.0	213.0	2100.0	200.0	4	< 0	••
	21:0:0:0:0:0:0:0 1 0	4.2.02.07.84 10	0	0	0.	0	0.03		•••	-•M	. 0	185.0
041101101101 1100 0	559999912 1		••	0		0.	•		•	3.(E .	185.0
441101104112 1100 00 100 00 2500.0 2500.0 44110110410 11 100 0 100 0 1000 2500.0 44110110410 11 100 0 100 0 1000 2500.0 44110110410 11 100 0 100 0 1000 2500.0 44110110410 11 1100 0 0 1000 1000 24110110410 11 1100 0 0 1000 241101104112 11 1440 0 1100 241101104112 11 1440 0 1100 241101104112 11 1440 0 1100 241101104112 11 1440 0 1100 241101104112 11 1100 0 0 1100 241101104112 11 11 11 0 1100 241101104112 11 11 10 0 0 0 241101104112 11 11 10 0 0 0 241101104112 11 11 10 0 0 0 24110111111 11 11 10	41101 1010 0<			 -		ສຸດ • •	0.8.0	.0•2	1.0			••
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	4.1.01.04.13 1. 1. 10.0	•	0	•	10.0	•	-	•••	2500.0		2500.0
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		•	•	°.	10.0	•	Ģ	•	2100.0		2100.0
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$			a e	••	10.0			.	1300.		1300.0
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	4.1.01.04.09 1. 10.0	•••	••	•••	10.01	••	••		142°1		0.04
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	4.1.01.04.08 1. 1. 10.0	•••	0.	c •	10.0	•	0.	0.	13.(0	13.0
9.9.99.99.35 1 11.0 2.2 1.2 2.1 7.0 9 $4.1.09.049.12$ 2. 2.0 0	999999956 1. 11.0 2.2 1.2 2.7 12.2 1.9 2.1 7.0 8 0.0 1.0 4.1004.94.1 1. 0.0 10 4.10 1.0 2.0 1.0 1.0 1.0 1.0 1.0 1.0 1.0 1.0 1.0 1	4.1.01.04.07 1. 14.0 2.1.01.04.12 1. 4.0		••	•••	14.0	0.00	0.00	0.01	19•(1	00	19.0
9,9,99,12 2 2 2 0	9, 9, 99, 99, 12 $2, 2, 0$ 0 $8, 4$ $5, 6$ $10, 4$ $5, 6$ $10, 4$ $5, 6$ $10, 4$ $0, 0$ $10, 4, 50$ $4, 0, 0, 0, 0, 12$ 11 $0, 0$ $0, 0$ $0, 0$ $0, 0$ $10, 4, 50$ $4, 1, 05, 03, 41$ 11 $1, 0$ $0, 0$ $0, 0$ $0, 0$ $10, 4, 50$ $9, 4, 97, 99, 33$ $56, 0$ $11, 6$ $17, 4$ $18, 6$ $17, 4$ $30, 2$ $4, 9$ $290, 0$ $10, 4, 50$ $9, 9, 999, 993, 33$ $56, 0$ $11, 6$ $17, 4$ $18, 6$ $17, 4$ $30, 2$ $4, 9$ $0, 0$ $10, 4, 50$ $9, 9, 999, 993, 35$ $50, 23, 0$ $51, 0$ $32, 6$ $31, 0$ $41, 2, 5$ $10, 0$ $10, 0$ $9, 9, 999, 993, 35$ $50, 99, 993, 35$ $51, 0$ $32, 6$ $31, 0$ $41, 2, 5$ $2153, 6$ $694, 2, 5$ $701ALS$ $310, 0$ $412, 5$ $2153, 6$ $698, 0$ $12555, 7$ $6942, 5$ $701ALS$ $310, 0$ $412, 5$ $2153, 6$ $698, 0$ $125255, 7$ $6942, 5$	9.9.99.99.36 1. 11.0	2.2	1.2	2.7	12.2	4 •9	2.1	7.0			61.0
$4_{110}(0), (0, 1), ($	1,1,0,0,0,1,0,1 $1,1$ $0,0$ </td <td>9.9.99.99.12 2. 2.0</td> <td>•</td> <td>8.4</td> <td>5.6</td> <td>10.4</td> <td>2.0</td> <td>£.</td> <td>•</td> <td>B.</td> <td>1 0</td> <td>•</td>	9.9.99.99.12 2. 2.0	•	8.4	5.6	10.4	2.0	£.	•	B.	1 0	•
9.99.9933 115. 46.0 .0 <td>9:99:99:33 11 .0 .0 .0 .0 .0 .0 10</td> <td>4.0.05.04.54 I. I. V</td> <td></td> <td></td> <td></td> <td>•</td> <td></td> <td>÷ C</td> <td>•</td> <td>. .</td> <td>0 0</td> <td>ດ ຄ</td>	9:99:99:33 11 .0 .0 .0 .0 .0 .0 10	4.0.05.04.54 I. I. V				•		÷ C	•	. .	0 0	ດ ຄ
99.99.99.32 115. 46.0 10 23.0 46.0 0 0 8.0 12.4 10 21.0 12.1	9.99.99.33 115. 46.0 .0 .0 23.0 46.0 23.0 6.0 .0 8 .0 21.0 9.99.99.33 58. 10 21.0 9.99.99.33 58. 0 11.6 17.4 13.6 31.0 1.2 120.0 8 0.0 9.99.99.33 2. 32.6 0 1.2 13.0 33.6 3.0 4.2 120.0 8 .0 75.0 9.99.99.35 2. 32.8 .0 351.7 580.0 4.12.5 2153.8 698.0 12525.7 6842.5 7074LS 7074LS 318.0 60.8 262.0 351.7 580.0 4.12.5 2153.8 698.0 12555.7 6842.5	4.4.07.04.51 1.			0.	0	•	•				1.0
9.9.99.99.33 580 11.6 17.4 18.6 17.4 30.2 4.9 290.0 R .0 66.0 9.9.99.99.34 5. 01.2 51.0 62.4 51.0 1.2 120.0 R .0 75.0 9.9.99.99.35 2. 32.8 .0 1.2 51.7 580.0 412.5 2153.8 698.0 12525.7 6642.5 10TALS 318.0 60.8 262.0 351.7 580.0 412.5 2153.8 698.0 12525.7 6642.5	9.9.99.9933 580 11.6 17.4 18.6 17.4 51.0 62.4 51.0 1.2 20.0 8 .0 75.0 9.9.99.99 33 6. 32.6 .0 1.2 31.0 6. 33.6 5.0 4.2 20.0 8 .0 75.0 9.9.99.99 35 23.6 .0 1.2 33.6 5.0 4.2 55.7 6. 42.5 30.0 1.2 55.7 6. 42.5 31.0 1.2 55.7 6. 42.5 51.5 1.5 1.5 1.5 1.5 1.5 1.5 1.5 1.5	9.9.99.99.32 115. 46.0	-	•	23.0	46.0	23.0	6.0	•	. œ		21.0
9.999-99.35 2: 51.0 1.2 51.0 1.2 51.0 57.0 9.999-99.35 2: 32.8 .0 .8 3.0 4.2 30.0 8 .0 57.0 101ALS 318.0 60.8 262.0 351.7 580.0 412.5 2153.4 698.0 12555.7 6842.5	9.9.99.99.35 2. 01.2 5.0 62.4 51.0 1.2 120.0 8 .0 57.0 9.9.99.99.35 2. 32.8 .0 .8 3.0 33.6 3.0 4.2 30.0 8 .0 57.0 707ALS 318.0 60.8 262.0 351.7 580.0 412.5 2153.8 698.0 12555.7 6842.5	9.9.99.99.33 58. 0	11.6	17.4	18.6	17.4	30.2	6.4	290.0	н.	0	86.0
TOTALS J18.0 60.8 262.0 351.7 580.0 412.5 2153.8 698.0 12525.7 6842.5	T0TALS 318.0 60.8 262.0 351.7 580.0 412.5 2153.8 698.0 12525.7 6842.5	9.9.99.99.34 b. 61.2 9.9.99.99.35 2. 32.8	•••	1.2	51.0	62.4 33.6	51.0	4.2	120•0 30•0		00	57.0 75.0
		TOTALS 318.0	60.8	262.0	351.7	580.0	412.5	2153.8	698.0	12525.	7	5842.5
				:				•				
			: - - 8			- - -	-	5 7				
								:	•	•••••••••••••••••••••••••••••••••••••••	:	
											÷	
				:	:							
		•			:	!						

LMSC-A847943 Vol. I

,<	- <u>i</u>		1		1		i	•	,		ı	i	1	:	1		!	
							i				•		;	•	1			
	•				:		1										•	
	:						:						1		i			
							1				•			i		•		
					:				:		÷			:		,		
	1				i		1			•				;		i	1	
		:								: { -	1			1	ĺ			
:			•	•					1					! 1 †	1	ł		
í		i	i	t	l	!		i			Ì	1						1
		0	ł	i		:		:				:	{		1			
'n		σ	1	:		1		;	1	1	ļ					ĺ		1
, ш	1 ALL	5)=	:	;				:	1		1	:						'
AGI		AY	: ;	,		·		1										
	503 ER1	5	i					i ,			ļ							
	ER	IME	:			į	Ì	:	!			1	ł		ł			
	ы Ма	TX .			1			1	i	1	}	i 1						
	2 - F	ST.	z		ļ	:		•	1	;	:							:
	0 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	ы О	S1 0			:	: 					:			}			
	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	MO	MIS		i		1		f	•	i	:	1	2	1	1		
		1AX	្រុ				l	;	; ;		! ;				1	•		
۱	Ĩ	1	TH	1	I	į		1								İ		ł
101			ę					1	1		i	•						l
LAT		i	,щ С	:				-			i	•						
-CL	Ì		INT	:				1										nt
CAL			EME			1		:		1		1	ŀ					ີຍີ
E	8		UIF	}				ì		i		1	:					
MAT	2	- NO	REG						Ì			•						2-2
÷щ	3ER	ISS	W	·		;	1	1	!		1					!		
ЮŴ-	Ĩ,	IW	GRI			1	1		:		1	÷	1					ig.
ż	z	IS	PRC				l	•		f !	:		•					H
SSI	SIO	Ŧ	1C					:		1	:	1	*	1	1			
IW	SI	9N I	H	i.				•			•							
Ъ	-	INT.	ENI			-		:	Ì	:								
RY		CoL	SCI	1				:				•						
MMA	:	29 - 20 - 20 - 20	¥	:			1	:			•	-		l				
SU	i	11	ie Ivo	S			1	:				:						
:	:	En S	ZE	N L				•				:						:
1	i	N D	ĒR]	R IV					1	1	1	:	l					:
i	A7. 02	THI	MMO	H X	Í				1				Í					•
	N X		N N	ы ш						1								•
4	N NO	N N	LO.	IÄÄ			İ	1	Ì	1								:
1	10N	AT S I S I S I S I S I S I S I S I S I S	3	0				•			1							•
:	ISS I	N N N N N N N N N N N N N N N N N N N	ING	2		+			İ	•	1							
	ΞTE	N N N	IST	AR-		:			-	l I	1				ĺ			
	EN	AIR SSIE		<u>n</u>		:			1	1								
	201	2 11 2 2 11 2 2 11 2	ΤHE	Ë	1	t			1	•	1							
	ļ	+	:	1			1				: -	;						
•		i	i		l						I I		ļ					
	ł	1	i I						í		:	- 1						
		i I	i i				1	1	Ì	1		: :		; !			l	
1	l	•	i		i	i	F	1	i		;		l	,		• •		

ι

7-28

١

Chapter 8 EXPLORATION PROGRAM GENERATION

A candidate lunar exploration program results from scheduling a specific set of missionmode combinations. Each such scheduled set exhibits unique values for various parameters. These parameters include total cost, annual costs, launch rate, and equipment start dates. In the MIMOSA analysis procedure, the planner establishes a tentative mission schedule, and the MIMOSA computer program adjusts that schedule to fit given launch rates and calculates the resulting values for the various lunar program parameters. This chapter discusses the two special aspects of lunar program generation – mission scheduling and computer usage.

8.1 MISSION SCHEDULING

In lunar program generation, the program planner establishes the schedule of missions by specifying the particular 6-mo time period of each mission start. The mission start date is input with the original mission data, but this date can be altered easily on any subsequent computer run.

Generally, study ground rules will dictate the mission start dates for any particular program. Most often, these rules involve financial constraints imposed in the form of launch-rate limits and maximum funding levels. Sometimes specific dates are dictated for new R&D, or a requirement is established as to the timing for introduction of new hardware. Integration with current or projected NASA plans will also affect scheduling.

8.2 COMPUTER USAGE

The MIMOSA computer program performs nearly all the cost and funding calculations for a candidate lunar program. If the data for all missions and modes are input

properly (paragraph 5.4 and 6.3), only four types of control cards are needed to initiate the costing routine. Paragraph 10.10 presents detailed instructions for preparing these cards.

A first card is required for each candidate program. This card identifies the program (i.e., titles the printout), establishes the maximum Saturn launch rate before funds must be expended to improve facilities, and sets "flags" for the computer as to whether certain cost calculation options are desired. There are three permitted options: to consider emergency modes (paragraph 6.6), to procure backup or spare equipment (item 33, Table 4-2), and to exceed a predetermined Saturn launch rate.

A second type of card is required to input the necessary Saturn launch-rate limit.

The third type of card inputs the mission-mode combinations to be costed. The combinations can be input in any order except where the start of one mission depends on completion of another mission. Space is provided with each input to override any start date previously given for a mission.

The fourth set of cards inputs the cost for facility launch-rate improvements. The effects of up to ten incremental step-ups in launch rate may be costed.

Figure 8-1 displays a portion of the computer printout for an example exploration program. This program performs a scientific effort of about 54,000 scientific manhours. The program has three major steps in its post-Apollo equipment evolution. The first step involves use of a LM shelter and a personnel taxi for 14-day missions. The second step, beginning in 1976, calls for a Saturn uprating to 125 percent of present payload for three-man LOR delivery, a cryogenic logistics vehicle for direct cargo delivery, and use of a three-man rover and trailer. The third step, beginning in 1981, involves a 175 percent Saturn for eventual six-man direct delivery. Mission 28 is the same example mission described in paragraphs 5.3 and 7.3.

SCIENTIFIC PROBAM NISSION WIX 00 NUMBER OF NISSION WIX 00 WONLD TOU BELIEVE THE FOLLOWING CONVENTS FROM THE COSTING SECTION NUMBER OF NISSION WIX 00 WONLD TOU BELIEVE THE FOLLOWING CONVENTS FROM THE COSTING SECTION NUMBER OF NISSION WIX 00 WONLD TOU BELIEVE THE FOLLOWING CONVENTS FROM THE COSTING SECTION NUMBER OF NISSION-SECTION SYSTEM 11700 HAS NO HISSION-ORIENTED JEES BUT IS USED FOR A TEST LUNCH OF ANOTHER SYSTEM. SYSTEM 117100 HAS NO HISSION-ORIENTED JEES BUT IS USED FOR A TEST LUNCH OF ANOTHER SYSTEM. SYSTEM 111700 HAS NO HISSION-ORIENTED JEES BUT IS USED FOR A TEST LUNCH OF ANOTHER SYSTEM. SYSTEM 111700 HAS NO HISSION-ORIENTED JEES BUT IS USED FOR A TEST LUNCH OF ANOTHER SYSTEM. SYSTEM 111700 HAS NO HISSION-ORIENTED JEES BUT IS USED FOR A TEST LUNCH OF ANOTHER SYSTEM. SYSTEM 111700 HAS NO HISSION-ORIENTED JEES BUT IS USED FOR A TEST LUNCH OF ANOTHER SYSTEM. SYSTEM 111700 HAS NO HISSION-ORIENTED JEES BUT IS USED FOR A TEST LUNCH OF ANOTHER SYSTEM. SYSTEM 11300 HAS NO HISSION-ORIENTED JEES BUT IS USED FOR A TEST LUNCH OF ANOTHER SYSTEM. SYSTEM 111700 HAS NO HISSION-ORIENTED JEES BUT IS USED FOR A TEST LUNCH OF ANOTHER SYSTEM. SYSTEM 11300 HAS NO HISSION-ORIENTED JEES BUT IS USED FOR A TEST LUNCH OF ANOTHER SYSTEM. SYSTEM 111700 HAS NO HISSION-ORIENTED JEES BUT IS USED FOR A TEST LUNCH OF ANOTHER SYSTEM. SYSTEM 11300 HAS NO HISSION-ORIENTED JEES BUT IS USED FOR A TEST LUNCH OF ANOTHER SYSTEM. THE ANOMECURARING COST CUNCE FOR A TEST LUNCH OF ANOTHER SYSTEM. </th <th>Summary of Lumar De Lumar De Louan E Contration Procedam A7 Mission Raulet Scientific Procedam A7 Mission Raulet Unualet To Locate A Boosted For Transportation System 121101 Unalet To Cacific A Boosted For Transportation System 121101 System System System<th>COSTING SECTION COMMENTS</th></th>	Summary of Lumar De Lumar De Louan E Contration Procedam A7 Mission Raulet Scientific Procedam A7 Mission Raulet Unualet To Locate A Boosted For Transportation System 121101 Unalet To Cacific A Boosted For Transportation System 121101 System System System <th>COSTING SECTION COMMENTS</th>	COSTING SECTION COMMENTS
Scientific Phodama A7 Mission wix 03 WoldD You allefer THE FOLLONING COMENIS FROM THE COSTING SECTION Uvakule To Locatte A BOGTER FOR TRANSPORTATION SYSTEM 121101 WHICH IS BELIEG USED FOR TEST LLUNCH OF SYSTEM. Uvakule To Locatte A BOGTER FOR TRANSPORTATION SYSTEM 121101 WHICH IS BELIEG USED FOR TEST LLUNCH OF SYSTEM. SYSTEM 117101_MAS_NO VISSION-ORIENTED_USES_BUT 15, USED FOR A TEST LLUNCH OF ANDTHER SYSTEM. SYSTEM 117101_MAS_NO VISSION-ORIENTED_USES_BUT 15, USED FOR A TEST LLUNCH OF ANDTHER SYSTEM. SYSTEM 117101_MAS_NO VISSION-ORIENTED_USES_BUT 15, USED FOR A TEST LLUNCH OF ANDTHER SYSTEM. SYSTEM 117101_MAS_NO VISSION-ORIENTED_USES_BUT 15, USED FOR A TEST LLUNCH OF ANDTHER SYSTEM. SYSTEM 117101_MAS_NO VISSION-ORIENTED_USES_BUT 15, USED FOR A TEST LLUNCH OF ANDTHER SYSTEM. SYSTEM 12101_MAS_NO VISSION-ORIENTED_USES_BUT 15, USED FOR A TEST LLUNCH OF ANDTHER SYSTEM. SYSTEM 12101_MAS_NO VISSION-ORIENTED_USES_BUT 15, USED FOR A TEST LLUNCH OF ANDTHER SYSTEM. SYSTEM 12101_MAS_NO VISSION-ORIENTED_USES_BUT 15, USED FOR A TEST LLUNCH OF ANDTHER SYSTEM. SYSTEM 12101_MAS_NO VISSION-ORIENTED_USES_BUT 15, USED FOR A TEST LLUNCH OF ANDTHER SYSTEM. SYSTEM 12101_MAS_NO VISSION-ORIENTED_USES_BUT 15, USED FOR A TEST LLUNCH OF ANDTHER SYSTEM. THE NOWEGURARING COST CUNCE FOR THE TANN 196/A THE NOWEGURARING COST CUNCE FOULP_BEGINS_ELALLER THAN 196/A THE NOWEGURARING COST CUNCE FOULP_BEGINS_ELALLER THAN 196/A THE NOWEGURARING COST C	Scientric Process A7 Mission WX 03 Mission WISSION STATE WOLU YU BLLEVE THE FOLLOWING COMENTS FEMA THE COSTING SECTION Windle TO LOGATE A BOSTER FOR TRANSPORTATION SYSTEM 121101 MILEM 15 BELID USED FOR TEST LUNDEM OF SYSTEM 213101 UNABLE TO LOGATE A BOSTER FOR TRANSPORTATION SYSTEM 121101 MILEM 15 BELID USED FOR A TEST LUNDEM OF SYSTEM. SYSTEM 117101 MIS NO VISSION-ONLINED USES BUT IS USED FOR A TEST LUNDEM OF ANDTHER SYSTEM. SYSTEM 117101 MIS NO VISSION-ONLINED USES BUT IS USED FOR A TEST LUNDEM OF ANDTHER SYSTEM. SYSTEM 117101 MIS NO VISSION-ONLINED USES BUT IS USED FOR A TEST LUNDEM OF ANDTHER SYSTEM. SYSTEM 117101 MIS NO VISSION-ONLINED USES BUT IS USED FOR A TEST LUNDEM OF ANDTHER SYSTEM. SYSTEM 12101 MIS NO VISSION-ONLINED USES BUT IS USED FOR A TEST LUNDEM OF ANDTHER SYSTEM. SYSTEM 12101 MIS NO VISSION-ONLINED USES BUT IS USED FOR A TEST LUNDEM OF ANDTHER SYSTEM. FIE MONRECURATING COST CUNVE FOR THIS USED FOR A TEST LUNDEM OF ANDTHER SYSTEM. FIE MONRECURATING COST CUNVE FOR TILL USED FOR A TEST LUNDEM OF ANDTHER SYSTEM. FIE MONRECURATING COST CUNVE FOR THIS USED FOR A TEST LUNDEM OF ANDTHER SYSTEM. FIE MONRECURATING COST CUNVE FOR TENT FOR USED FOR A TEST LUNDEM OF ANDTHER SYSTEM. FIE MONRECURATING COST CUNVE FOR THIS USED FOR A TEST LUNDEM OF ANDTHER SYSTEM. FIE MONRECURATING COST CUNVE FOR THER THAN 196/A FIE MONRECURATING COST CUNVE FOR MISS ESCHARTER FUNAL 196/A FIE MONRECURATING COST CUNVE F	SUMMARY OF LUNAR EXPLORATION PROGRAM EXAMPLE
VOILD YOU BELIEVE THE FOLLONING COMENTS FROM THE COSTING SECTION UNABLE TO LOCATE À BODSTER FOM THANSPORTATION SYSTEM 121101, WHICH IS BEITIG USED FOR TEST LAUNCH OF SYSTEM SYSTEM 117202 MAS NO 4155109-ORTENTED USES BUT IS USED FOM A TEST LAUNCH OF ANDTHER SYSTEM. SYSTEM 117202 MAS NO 4155109-ORTENTED USES BUT IS USED FOM A TEST LAUNCH OF ANDTHER SYSTEM. SYSTEM 117202 MAS NO 4155109-ORTENTED USES BUT IS USED FOM A TEST LAUNCH OF ANDTHER SYSTEM. SYSTEM 117202 MAS NO 4155100-ORTENTED USES BUT IS USED FOM A TEST LAUNCH OF ANDTHER SYSTEM. SYSTEM 121101 MAS NO 4155100-ORTENTED USES BUT IS USED FOM A TEST LAUNCH OF ANDTHER SYSTEM. THE NOMBELURATINE CONSTITUTED USES BUT IS USED FOM A TEST LAUNCH OF ANDTHER SYSTEM. THE NOMBELURATINE CONTENTED USES BUT IS USED FOM A TEST LAUNCH OF ANDTHER SYSTEM. THE NOMBELURATINE COST CUNVE FOM VIONE SCIENTIFIC EQUIP. BEGINS EALLER THAN 1967.A. THE NOMBELURATINE COST CUNVE FOM VIONE SCIENTIFIC EQUIP. BEGINS EALLER THAN 1967.A. THE NOMBELURATING COST CUNVE FOM VIONE SCIENTIFIC EQUIP. BEGINS EALLER THAN 1967.A. THE NOMBELURATING COST CUNVE FOM VIONE SCIENTIFIC EQUIP. BEGINS EALLER THAN 1967.A. THE NOMBELURATING COST CUNVE FOM VIONE SCIENTIFIC EQUIP. BEGINS EALLER THAN 1967.A. THE NOMBELURATING COST CUNVE FOM VIONE SCIENTIFIC EQUIP. BEGINS EALLER THAN 1967.A. THE NOMBELURATING COST CUNVE FOM VIONE SCIENTIFIC EQUIP. BEGINS EALLER THAN 1967.A. THE NOMBELURATING COST CUNVE FOM VIONE SCIENTIFIC EQUIP. BEGINS EALLER THAN 1967.A. THE NOMBELURATING COST CUNVE FOM VIONE SCIENTIFIC EQUIP. BEGINS EALLER THAN 1967.A. THE NOMBELURATING COST CUNVE FOM VIONE SCIENTIFIC EQUIP. BEGINS EALLER THAN 1967.A. THE NOMBELURATING COST CUNVE FOM VIONE SCIENTIFIC EQUIP. BEGINS EALLER THAN 1967.A. THE NOMBELURATING COST CUNVE FOM VIONE SCIENTIFIC EQUIP. BEGINS EALLER THAN 1967.A. THE NOMBELURATING COST CUNVE FOM VIONE SCIENTIFIC EQUIP. BEGINS EALLER THAN 1967.A. THE NOMBELURATING COST CUNVE FOM VIONE SCIENTIFIC EQUIP. BEGINS EALLER THAN 1967.A. THE NOMBELURATING COST CUNVE FOM VIONE SCIENTIFIC EQUIP. BEG	 WOLD YOU BELIEVE THE FOLLOWING COMENTS FROM THE COSTING SECTION UNABLE TO LOCATE À BODSTER FON TRANSPORTATION SYSTEM 12101 WHICH IS RELIGA USED FOR TEST LAUNCH OF ANDTHER SYSTEM. SYSTEM 117201 MAS JOU 4155109-ORIENTED USES BUT IS USED FOR A TEST LAUNCH OF ANDTHER SYSTEM. SYSTEM 117202 MAS JOU 4155109-ORIENTED USES BUT IS USED FOR A TEST LAUNCH OF ANDTHER SYSTEM. SYSTEM 117202 MAS JOU 4155109-ORIENTED USES BUT IS USED FOR A TEST LAUNCH OF ANDTHER SYSTEM. SYSTEM 117202 MAS JOU 4155109-ORIENTED USES BUT IS USED FOR A TEST LAUNCH OF ANDTHER SYSTEM. SYSTEM 111201 MAS JOU 4155109-ORIENTED USES BUT IS USED FOR A TEST LAUNCH OF ANDTHER SYSTEM. SYSTEM 111201 MAS JOU 4155109-ORIENTED USES BUT IS USED FOR A TEST LAUNCH OF ANDTHER SYSTEM. SYSTEM 112101 MAS JOU 4155109-ORIENTED USES BUT IS USED FOR A TEST LAUNCH OF ANDTHER SYSTEM. SYSTEM 112101 MAS JOU 4155109-ORIENTED USES BUT IS USED FOR A TEST LAUNCH OF ANDTHER SYSTEM. THE NOMBEURATING COST CUNVE FOR HINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A. THE NOMBEURATING COST CUNVE FOR HINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A. THE NOMBEURATING COST CUNVE FOR HINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A. THE NOMBEURATING COST CUNVE FOR HINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A. THE NOMBEURATING COST CUNVE FOR HINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A. THE NOMBEURATING COST CUNVE FOR HINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A. THE NOMBEURATING COST CUNVE FOR HINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A. THE NOMBEURATING COST CUNVE FOR HINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A. THE NOMBEURATING COST CUNVE FOR HINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A. THE NOMBEURATING COST CUNVE FOR HINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A. THE NOMBEURATING COST CUNVE FOR HINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A. THE NO	SCIENTIFIC PROGRAM A7 MISSION MIX 03 NUMBER OF MISSIONS = 26
UWARLE TO LOCATE A BOOSTER FOR TRANSPORTATION SYSTEM 121101 WICH IS BELIIG USED FOR TEST LAUGEH OF SYSTEM. SYSTEM 117202 HAS, DO MISSION-ORTENTED USES BUT 15 USED FOR A TEST LAUGEH OF ANDTHER SYSTEM. SYSTEM 117203 HAS NO MISSION-ORTENTED USES BUT 15 USED FOR A TEST LAUGEH OF ANDTHER SYSTEM. SYSTEM 111204 HAS NO MISSION-ORTENTED USES BUT 15 USED FOR A TEST LAUGEH OF ANDTHER SYSTEM. SYSTEM 121101 HAS NO MISSION-ORTENTED USES BUT 15 USED FOR A TEST LAUGEH OF ANDTHER SYSTEM. SYSTEM 12101 HAS NO MISSION-ORTENTED USES BUT 15 USED FOR A TEST LAUGEH OF ANDTHER SYSTEM. SYSTEM 12101 HAS NO MISSION-ORTENTED USES BUT 15 USED FOR A TEST LAUGEH OF ANDTHER SYSTEM. SYSTEM 12101 HAS NO MISSION-ORTENTED USES BUT 15 USED FOR A TEST LAUGEH OF ANDTHER SYSTEM. THE NOMECURATING COST CURVE FOR HINOR SCIENTIFIC EQUIP. BEGINS EARLER THAN 1967A THE NOMECURATING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLER THAN 1967A THE NOMECURATING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLER THAN 1967A THE NOMECURATING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLER THAN 1967A THE NOMECURATING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLER THAN 1967A THE NOMECURATING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLER THAN 1967A THE NOMECURATING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLER THAN 1967A THE NOMECURATING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLER THAN 1967A THE NOMECURATING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLER THAN 1967A THE NOMECURATING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLER THAN 1967A THE NOMECURATING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLER THAN 1967A THE NOMECURATING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLER THAN 1967A THE NOMECURATING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLER THAN 1967A THE NOMECURATING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLER THAN 1967A THE NOMECURATING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLER THAN 1967A THE NOMECURATING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEG	walle fo Locate A Booster for transportation system 12101 which its define used for test tauget of system 21201 system 11201, max we used on the system 12101 which of a first tauget of an on the system. system 11202 max we used on the set of a first tauget of a more system. system 11202 max we used on the set of a first tauget of a more system. system 11202 max we used on the system. system 11202 max we used on the system. system 11204 max we used on the system. system 12101 max we used on the system. system 12101 max we used on the system. system 12101 max we used on the system. system 12101 max we used on the system. system 12101 max we used on the system. system 12101 max we used on the system. system 12101 max we used on the system. system 12101 max we used on the system. system 12101 max we used on the system. system 12101 max we used on the system. system 12101 max we used on the system. the wonecommute cost curve for unue sectors called sature term 135/a the wonecommation cost curve for unue sectors called sature than 135/a the wonecommation cost curve for unue sectors called sature than 135/a the wonecommatus cost curve for unue sectors called sature than 135/a the wonecommatus cost curve for unue sectors called sature to the sof at 136/a<	WOULD YOU BELIEVE THE FOLLOWING COMMENTS FROM THE COSTING SECTION
SYSTEM 11710. MAG NO WISSION-ORTENTED JOSES BUT IS USED FOR A TEST LAUNCH OF ANOTHER SYSTEM. SYSTEM 117202 MAG NO WISSION-ORTENTED USES BUT IS USED FOR A TEST LAUNCH OF ANOTHER SYSTEM. SYSTEM 121101 MAG NO WISSION-ORTENTED USES BUT IS USED FOR A TEST LAUNCH OF ANOTHER SYSTEM. SYSTEM 121101 HAG NO WISSION-ORTENTED USES BUT IS USED FOR A TEST LAUNCH OF ANOTHER SYSTEM. SYSTEM 121101 HAG NO WISSION-ORTENTED USES BUT IS USED FOR A TEST LAUNCH OF ANOTHER SYSTEM. SYSTEM 121101 HAG NO WISSION-ORTENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NOWRECURRING COST CURVE FOR HINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NOWRECURRING COST CURVE FOR HINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NOWRECURRING COST CURVE FOR HINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NOWRECURRING COST CURVE FOR HINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NOWRECURRING COST CURVE FOR HINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NOWRECURRING COST CURVE FOR HINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NOWRECURRING COST CURVE FOR HINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NOWRECURRING COST CURVE FOR HINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NOWRECURRING COST CURVE FOR HINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NOWRECURRING COST CURVE FOR HINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NOWRECURRING COST CURVE FOR HINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NOWRECURRING COST CURVE FOR HINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NOWRECURRING COST CURVE FOR HINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NOWRECURRING COST CURVE FOR HINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NOWRECURRING COST CURVE FOR HINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NOWRECURRING COST CURVE FOR HINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NOWRECURRING COST CURVE FOR HINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NOMECURRING COST CURVE FOR HINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NOMECURRI	SYSTEM 117101. MAS, NO WISSION-ORIENTED JJSES, BUT IS USED FOR A TEST LAUNCH OF ANOTHER SYSTEM. SYSTEM 117202. HAS, NO WISSION-ORIENTED JJSES BUT IS USED FOR A TEST LAUNCH OF ANOTHER SYSTEM. SYSTEM 117200. HAS NO WISSION-ORIENTED JJSES BUT IS USED FOR A TEST LAUNCH OF ANOTHER SYSTEM. SYSTEM 12101 HAS NO WISSION-ORIENTED JJSES BUT IS USED FOR A TEST LAUNCH OF ANOTHER SYSTEM. SYSTEM 12101 HAS NO WISSION-ORIENTED JJSES BUT IS USED FOR A TEST LAUNCH OF ANOTHER SYSTEM. SYSTEM 12101 HAS NO WISSION-ORIENTED JJSES BUT IS USED FOR A TEST LAUNCH OF ANOTHER SYSTEM. THE NOMECURATING COST CUNVE FOR MINOR SCIENTIFIC EQUID. BEGINS EALLER THAN 1967.A THE NOMECURATING COST CUNVE FOR MINOR SCIENTIFIC EQUID. BEGINS EALLER THAN 1967.A THE NOMECURATING COST CUNVE FOR MINOR SCIENTIFIC EQUID. BEGINS EALLER THAN 1967.A THE NOMECURATING COST CUNVE FOR MINOR SCIENTIFIC EQUID. BEGINS EALLER THAN 1967.A THE NOMECURATING COST CUNVE FOR MINOR SCIENTIFIC EQUID. BEGINS EALLER THAN 1967.A THE NOMECURATING COST CUNVE FOR MINOR SCIENTIFIC EQUID. BEGINS EALLER THAN 1967.A THE NOMECURATING COST CUNVE FOR MINOR SCIENTIFIC EQUID. BEGINS EALLER THAN 1967.A THE NOMECURATING COST CUNVE FOR MINOR SCIENTIFIC EQUID. BEGINS EALLER THAN 1967.A THE NOMECURATING COST CUNVE FOR MINOR SCIENTIFIC EQUID. BEGINS EALLER THAN 1967.A THE NOMECURATING COST CUNVE FOR MINOR SCIENTIFIC EQUID. BEGINS EALLER THAN 1967.A THE NOMECURATING COST CUNVE FOR MINOR SCIENTIFIC EQUID. BEGINS EALLER THAN 1967.A	UMABLE TO LOCATE A BOOSTER FOR TRANSPORTATION SYSTEM 121101 WHICH IS BEING USED FOR TEST LAUNCH OF SYSTEM 212101
SYSTEM 117202 MAS NO MISSION-ORTENTED USES BUT IS USED FOR A TEST LAUNCH OF ANDTHER SYSTEM. SYSTEM 117204 MAS NO WISSION-ORTENTED USES BUT IS USED FOR A TEST LAUNCH OF ANDTHER SYSTEM. SYSTEM 121401 HAS NO WISSION-ORTENTED USES BUT IS USED FOR A TEST LAUNCH OF ANDTHER SYSTEM. SYSTEM 121401 HAS NO WISSION-ORTENTED USES BUT IS USED FOR A TEST LAUNCH OF ANDTHER SYSTEM. SYSTEM 121401 HAS NO WISSION-ORTENTED USES BUT IS USED FOR A TEST LAUNCH OF ANDTHER SYSTEM. THE NONRECURRING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A	SYSTEM 117202 HAS NO MISSION-ORIENTED USES BUT IS USED FOR A TEST LAUNCH OF ANDTHER SYSTEM. SYSTEM 1117204 HAS NO MISSION-ORIENTED USES BUT IS USED FOR A TEST LAUNCH OF ANOTHER SYSTEM. SYSTEM 121101 HAS NO MISSION-ORIENTED USES BUT IS USED FOR A TEST LAUNCH OF ANOTHER SYSTEM. SYSTEM 121101 HAS NO MISSION-ORIENTED USES BUT IS USED FOR A TEST LAUNCH OF ANOTHER SYSTEM. SYSTEM 121101 HAS NO MISSION-ORIENTED USES BUT IS USED FOR A TEST LAUNCH OF ANOTHER SYSTEM. HE NONRECURRING COST CHUVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAM 1967A THE NONRECURRING COST CHUVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAM 1967A THE NONRECURRING COST CHIVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAM 1967A THE NONRECURRING COST CHIVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAM 1967A THE NONRECURRING COST CHIVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAM 1967A THE NONRECURRING COST CHIVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAM 1967A THE NONRECURRING COST CHIVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAM 1967A THE NONRECURRING COST CHIVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAM 1967A THE NONRECURRING COST CHIVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAM 1967A THE NONRECURRING COST CHIVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAM 1967A THE NONRECURRING COST CHIVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAM 1967A THE NONRECURRING COST CHIVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAM 1967A THE NONRECURRING COST CHIVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAM 1967A THE NONRECURRING COST CHIVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAM 1967A THE NONRECURRING COST CHIVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAM 1967A THE NONRECURRING COST CHIVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAM 1967A THE NONRECURRING COST CHIVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAM 1967A THE NONRECURRING COST CHIVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAM 1967A THE NONRECURRING FOR THAN SCIENTIFIC EQUIP. BEGINS EARLIER THAM 1967A THE NONRECURRING	SYSTEM 117101_HAS NO MISSION-ORIENTED USES BUT IS USED FOR A TEST LAUNCH OF ANOTHER SYSTEM.
SYSTEM 1117204 HAS NO. WISSION-ORTENTED USES BUT IS USED FOR A TEST LAUNCH OF ANDTHER SYSTEM. SYSTEM 121101 HAS NO WISSION-ORTENTED USES BUT IS USED FOR A TEST LAUNCH OF ANOTHER SYSTEM. SYSTEM 121401 HAS NO WISSION-ORTENTED USES BUT IS USED FOR A TEST LAUNCH OF ANOTHER SYSTEM. THE NONREGURATING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONREGURATING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONREGURATING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONREGURATING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONREGURATING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONREGURATING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONREGURATING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONREGURATING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONREGURATING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONREGURATING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A	FIGEN 117204 HAS NO MISSION-ORTENTED USES BUT IS USED FOR A TEST LAUNCH OF ANDHER SYSTEM. SYSTEM 121101 HAS NO MISSION-ORTENTED USES BUT IS USED FOR A TEST LAUNCH OF ANDHER SYSTEM. SYSTEM 121401 HAS NO MISSION-ORTENTED USES BUT IS USED FOR A TEST LAUNCH OF ANDHER SYSTEM. THE NONAECURATING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONAECURATING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONAECURATING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONAECURATING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONAECURATING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONAECURATING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONAECURATING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONAECURATING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONAECURATING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONAECURATING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONAECURATING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONAECURATING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONAECURATING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONAECURATING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONAECURATING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONAECURATING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONAECURATING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONAECURATING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONAECURATING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONAECURATING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONAECURATING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NON	SYSTEM 117202 HAS NO MISSION-ORIENTED USES BUT IS USED FOR A TEST LAUNCH OF ANOTHER SYSTEM.
SYSTEM 121101 HAS NO MISSION-ORIENTED JASE BUT IS USED FOR A TEST LAUNCH OF ANOTHER SYSTEM. SYSTEM 121401 HAS NO MISSION-ORIENTED USES BUT IS USED FOR A TEST LAUNCH OF ANOTHER SYSTEM. THE NONREGURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. A THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. A THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. A THAN 1967A THE NONRECURRING FOR MINOR SCIENTIFIC EQUIP. A THAN 1967A THE NONRECURRING FOR MINOR SCIENTIFIC EQUIP. A THAN 1967A THE NONRECURRING FOR MINOR SCIENTIFIC	SYSTEM 121101 MAS NO MISSION-ORIENTED JEES BUT IS USED FOR A TEST LANKH OF ANOTHER SYSTEM. SYSTEM 121401 MAS NO MISSION-ORIENTED USES BUT IS USED FOR A TEST LANKH OF ANOTHER SYSTEM. THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINO	SYSTEM 117204 HAS NO MISSION-ORIENTED USES BUT IS USED FOR A TEST LAUNCH OF ANOTHER SYSTEM.
SYSTEM IZINOI HAS NO MISSION-ORIENTED USES BUT IS USED FOR A TEST LAUNCH OF ANOTHER SYSTEM. THE NONRECURRING COST CUNVE FOR HINOR SCIENTFIC EQUIP. BEGINS EARLER THAN 1967A THE NONRECURRING COST CUNVE FOR MINOR SCIENTFIC EQUIP. BEGINS EARLER THAN 1967A THE NONRECURRING COST CUNVE FOR MINOR SCIENTFIC EQUIP. BEGINS EARLER THAN 1967A THE NONRECURRING COST CUNVE FOR MINOR SCIENTFIC EQUIP. BEGINS EARLER THAN 1967A THE NONRECURRING COST CUNVE FOR MINOR SCIENTFIC EQUIP. BEGINS EARLER THAN 1967A THE NONRECURRING COST CUNVE FOR MINOR SCIENTFIC EQUIP. BEGINS EARLER THAN 1967A THE NONRECURRING COST CUNVE FOR MINOR SCIENTFIC EQUIP. BEGINS EARLER THAN 1967A THE NONRECURRING COST CUNVE FOR MINOR SCIENTFIC EQUIP. BEGINS EARLER THAN 1967A THE NONRECURRING COST CUNVE FOR MINOR SCIENTFIC EQUIP. BEGINS EARLER THAN 1967A THE NONRECURRING COST CUNVE FOR MINOR SCIENTFIC EQUIP. BEGINS EARLER THAN 1967A THE NONRECURRING COST CUNVE FOR MINOR SCIENTFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CUNVE FOR MINOR SCIENTFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CUNVE FOR MINOR SCIENTFIC EQUIP. BEGINS EARLIER THAN 1967A	SYSTEM 121401 HAS NO MISSION-ORIENTED USED FOR A TEST LAUNCH OF ANOTHER SYSTEM. HE NONRECURRING COST CUNVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE MONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE MONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE MONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE MONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE MONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE MONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE MONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE MONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE MONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE MONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE MONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE MONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE MONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE MONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE MONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. B	SYSTEM 121101 HAS NO MISSION-ORIENTED USES BUT IS USED FOR A TEST LAUNCH OF AMOTHER SYSTEM.
THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A	THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC AUTO- BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC AUTO- BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC AUTO- BEGINS EARLIER THAN 1967A	SYSTEM 121401 HAS NO MISSION-ORIENTED USES BUT IS USED FOR A TEST LAUNCH OF ANOTHER SYSTEM.
THE MONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP, BEGINS EARLIER THAN 1967A THE MONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP, BEGINS EARLIER THAN 1967A THE MONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP, BEGINS EARLIER THAN 1967A THE MONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP, BEGINS EARLIER THAN 1967A THE MONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP, BEGINS EARLIER THAN 1967A THE MONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP, BEGINS EARLIER THAN 1967A	 THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE MONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE MONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE MONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A 	THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A
THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A	THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC FOULP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC FOULP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC FOULP. BEGINS EARLIER THAN 1957A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC FOULP. BEGINS EARLIER THAN 1957A	THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A
THE NOMRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP, BEGINS EARLIER THAN 1967A THE MONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP, BEGINS EARLIER THAN 1967A THE NOMRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP, BEGINS EARLIER THAN 1967A THE NOWRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP, BEGINS FARLIER THAN 1967A	THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THF NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THF NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THF NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THF NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THF NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THF NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THR NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THR NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THR NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THR NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THR NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THR NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THR NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A	THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A
THE NOWRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THF NOWRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THF NOWRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A	THE MONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. REGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A	THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A
THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A	THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A FIG. 8-1 Example Computer Printout – Integration and Cost	THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. REGINS EARLIER THAN 1967A
THE NOWRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLER THAN 1967A	THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A	THF NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A
	Fig. 8-1 Example Computer Printout – Integration and Cost	THE NONRECURRING COST CURVE FOR MINOR SCIENTIFIC EQUIP. BEGINS EARLIER THAN 1967A
	Fig. 8-1 Example Computer Printout – Integration and Cost	
	Fig. 8-1 Example Computer Printout - Integration and Cost	
	Fig. 8-1 Example Computer Printout - Integration and Cost	
	Fig. 8-1 Example Computer Printout - Integration and Cost	
	Fig. 8-1 Example Computer Printout - Integration and Cost	
	Fig. 8-1 Example Computer Printout - Integration and Cost	
	Fig. 8-1 Example Computer Printout – Integration and Cost	

LOCKHEED MISSILES & SPACE COMPANY

•											
THE FOLLOWING COSTS, IN M	ILLIONS, WERE	CALCULATED	FOR TH	IS LUN	AR EXPL	ORATIO	PROGRI	W			
TOTAL NONRECURRING COST Novrecurring development Cost of facilities	6565.4 287.5	6852 . 9		TOTAL Re Re	RECURRI CURRINC CURRINC	ING COSI PROCUE OPER 1	r Rement FIONS		18705.8 15125.2 3580.6		- :
	101	AL PROGRAM	COST	. 255	58.7						
THE FOLLOWING ITEMS OF EXI	PLORATION EQUI	PMENT ARE R	EQUIRE	0		•			Y (MTI LTONS		
		R / D FI	RST	NUMBER	PROCUE	RD	NONRECL	JRRING	REC	JRRING	
EOUIPMENT NAME	IDENTIFIER	START US	ED	NISS T	1 SPA	SES F	0 / 2	C 0F F	HDWE	SNGO	
PERS TRANSP-LOR(2-MAN)	1111-01	N / A 19	714	ŝ	0		0.	0.	6 T	0.	1 • :
PEAS TRANSP - AAP FUNDED	1111-04	N / A 19	708	1	0		•	•	0.	-	
PERS TRANSP-LOR(3-MAN)	1112-05	19738 19	76A	6			33.6	•	7.5	c •	
PERS IRANSPHUIR(BEMAN) 106- TRANEDH 02-8000VG	10-4211	1979A 19	HIL	11	0 0		10.7	• •	60 u 11 11	•	
LOG. TRANSP - AAP FUNDED	1132-04		709	•				•••			
LUG. TRANSP-DIR, 16000KG	1144-01	1974A 19	764	14	0		5.7	•	3.0	.0.	
LOG. TRANSP-DIK,25000KG	1147-01	19798 19 N / A 19	619 719		00	.	2°4	•	ທ ີ ຄ	ç	
ORBITER TRANS-AAP FUNDED	1152-03		70A	,							
DEVELOPMENT TEST TRANSP	1171-01	N / A 19	75R	0	3		•	•		0	
DEVELOPMENT TEST TRANSP	1172-02		754	-	~		•	.	••••••••••••••••••••••••••••••••••••••	c.	
ATLAS-AGENA LAUNCH VEH.	1211-01		004 V04	0	v			••	9" "		
SATURN 18 LAUNCH VEHICLE	1214-01	N / A 19	758	0	4		•	•	c	0	
SALURN 5 LAUNCH VEH. Addit 0 sat v Lainch veh	1221-01		704	n r	0 0		•	••	2099.5	270.0	
UPRATED SATURN 5	1231-01	1972A 19	764	23.			570.0	42.0	2760.3	483.6	
UPRATED SATURN 5	1251-01	19778 19	818	28	0	~	318.0	135.0	3673.1	576.6	
CM-LOK-J-MAN CM-LOR-J-MAN	1311-01	N / A 19 1973B 19	760	r1 6	0 0	N	- U	••	903.0 608.5	226.5	
SM-LOR(STOR)-3-MAN	1321-01	N / A 19	714	13	0		0.		328.1	61.5	
SM-LOR(STOR)-J-MAN DEPC STG-LOP CTOP	1321-04	19738 19 19678 19	76A	6 4	00		47.6 20 E	•	212.4	0.05	
PERS DESC STG-LOR STOR	1332-02	1972B 19	764	6	0		29.0	•••	206.5	35.0	
PERS ASC STG-LOR 2-MAN	1341-01	19678 19	1708	9	0	_	11.1	•	246.8	62.3	
PERS ASC STG-LOR 3-MAN	1342-01	19728 19	764	a 1			17.0	0	354.1	90°0	
CM-DIRFCT-6-MAN	1412-02	1977R 19	818 818	`:			1.001		172.5	67.2 205 b	
PERS BRAKING STG-DIR	1421-02	19768 19	818	11			59.4	24.0	78.8	15.6	
LOG. GRAK STG-DIR CRYO	1423-02	N / A 19	76A		0		•	•	113.1	20.8	
PERS LANDING STG-DIR	1431-03	N / A 19	818 818	11			• •		131.3	24.7	
								,		1	

LOCKHEED MISSILES & SPACE COMPANY

LMSC-A847943 Vol. I

:

,

GR/
DRATI
IDENTI
1433-0
1454-0
2132-0
2222-1
2325-
2421-
2433-0
2512-0
2521-0
3213-0
3223-(
3231-0
3242-(
3243-0
-

i

••

.

8-5

LOCKHEED MISSILES & SPACE COMPANY

Scientific Paoeuk A7 Mission M1 Mission M1 Mission M1 Scientific Paoeuk THE Following Exhomation concert words are Redured Cost Symmetric Million Cost Symmetric Million Cost Symmetric Million WOIL Total A1 site Total A1 site Total A1 site Following Million MOIL Total A1 site Total A1 site Total A1 site Following Million MOIL Total A1 site Total A1 site Total A1 site Following Million MOIL Total A1 site Total A1 site Total A1 site Total A1 site Following MOIL Total A1 site MOIL Total A1 site MOIL Total A1 site MOIL Total A1 site MOIL Total A1 site Total A1 site Total A1 site	SCIENT									PAGE 3	
THE FOLLONING EXPLORATION CONCEPT MODES ARE REQUIRED EOST CONCENTRATION CONCERT MODES 10001 CREW LEVEL STAY TYRE FIRST MODE RECONSTRUCT 10001 CREW LEVEL STAY TYRE FIRST MODE RECONSTRUCT 10001 CREW LEVEL TONLS USED WODE PEC PEC 1001 CREW LEVEL TONLE TONL TONL PEC PEC 1001 CREW LEVEL TONN USED NO PEC PEC 1001 DON 2 1971A 2 1971A 2 1971A 1001 2 1971A 2 1971A 2 1971A 2 1971A 1001 2 2 1971A 1 2 1971A 2 1971A 1001 2 2 1971A 1 2 2 1971A 1001 2 2 2 1971A 1 2 1971A 1001 2		IFIC PRO	SRAM A7	MISSIM	ON MIX 03		NUMBER OF MISSI	= SNO	26		
UNMERTER CACHE LEVEL CACHE LEVEL CACHE LEVEL CALMAN SITE CANNEL AT SITE CANNEL CANNEL CANNEL CANNEL CANNEL CANNEL CANNEL CANNEL CANNEL <th< th=""><th>THE FOLLOWING EXP</th><th>LORATION</th><th>CONCEPT</th><th>MODES ARE RE</th><th>OUIRED</th><th></th><th></th><th></th><th></th><th></th><th>;</th></th<>	THE FOLLOWING EXP	LORATION	CONCEPT	MODES ARE RE	OUIRED						;
	MODE IUFNTIFIER	CREW LE TOTAL	LVEL	STAY TIME (Days)	FIRST USED	NUMBER USED	R / D C OF		RECURR	ING OPNS	
	303-10021-02 303-10022-02	юn	คต	28	1970A 1971B	~	0.		•	25•2 50-0	•
	402-10011-01 402-10011-01	m#	~ ~	16	19714	r r		0.0		16.7	
	402-10012-01	ייי ר <u>י</u> מיי ר	v v		19708	ר נ ו	0			50.1 16.7	
	402-10081-01 406-30161-01	in v	N V	14	19748 19858		Ģ	•	• •	54.5	
	406-30161-02	 - -	 و ر	720	19864		0	0		93.4 68.2	
	503-20162-01 503-20191-01	r) r	P) P	06	1976A	r.	Ģ	•	•	176.4	
	503-20201-01	רא נ	0 10	60	19808			•••	20	34°5	
	503-30083-01 503-30084-01	5	ب در	06	1982A	- 0	0.	•	0	25.2	
203-Jolit-oli J J J 90 19838 I J J 90 10 90 90 90 90 90 90 90 90 90 90 90 90 90	503-30103-01	מיוס	ი ო	06 06	1981H	NN	0.0		•	109.0	
	503-30114-01	5	3	60	19838	1	•0	0	0	68.2	
	:				:	:					
							ŧ			- - -	
	• • • • •	•	•		:		•			•	
							•	ı			
							named and the state of the stat				
							:		:		

LOCKHEED MISSILES & SPACE COMPANY

LMSC-A847943

Vol. I

	141			Ĩ		NOTCOT	C0 X1W	-	NUMBER UP	- SNOTSSTW	02	
Ŧ	FOLLOWING SI	CHEDULE	S AND C	OSTS W	ERE CA	LCULA TED	FOR THIS	LUNAR EXPL	ORATION P	ROGRAM		
ENDAR	NOISSIM	MISS	SATURN LON USE	LAUNC	H RATE Velopmi	ENT USE	NONRE	ST SUMMARY	C (MILLION	IS) RRING	FY ANNUAL COST	CUM.
AR	START	18	5 500	æ	18	5 SUPR	R / C	C OF F	HDWE	OPNS	(WIL.)	(816.)
57A							4.8				4.8	
VHY							27.3		25°5		0 071	c
	•	•	;	:	:		105.3	6	179.9	4.0	C	
9A							127.7	-	305.6	6.5 19.9	730.0	6.
VOV	10		1				59.4	~	461.9	73.8	1121.3	2.0
714	61		2			:	59.9	6.6	436.7	92•3 111-1	1183.3	3.2
	10						74.5	1.0	388.8	0.79		
72A	11		~ ~				124.8	4.0	397.4 398.1	89.9 86.8	1177.4	5.4
73A							228.5	19.0	393.4	72.5	1388.0	5.A
	14 13		'n				299.0	19.0	358.5	128.2		
. 4 4			. .				0.025. 	11.0	5/1°1	89.1 86.8	1600.9	
75A	•		>			1	250.4	.1	458.9	50.2	1571.9	8.9
!					-	-	116.4		508.8	79.4		
76A	16		~ ~	- /			5°.7°	6 . ,	467.5	119.7	1340.1	10.3
774	14		~ ~				10.4	2 U - F	400.5	125.0	1271.8	11.6
:	19	:					117.7	17.0	411.4	123.8		
784	20		ŝ				240.1	0.00 10.00	389.1	125.3	1456.4	13.0
79A	22 21		t				537.4	58.9	403.2	130.2	2071.7	15.1
	23		N				589.8	3***D	1.044	89.0	C OFFC	
AUC	24		4			-	1.740	+ + + + + + + + + + + + + + + + + + +	540.0	1.01	9.2662	
818	. u		· •		n	-	235.6	2.4	529.8	1001	1941.0	19.4
J2A	26		א ר 				143.0		517.5	66.0	1458.7	20.8
	27		ι.				154.7		507.9	99.1	9 0 9 9 9	-
	28			•		•	128.1	-	482.9	142.8	C*0+CT	
94A	29		in i				79.9		445.7	110.7	1390.0	23.8
15A	20		۳. ۱				27.8		360.2	128.3	0 1 10	
	31		9	_					198.0	129.2	20746	
36A	32	:			:				160.1	75.7	563.0	25,3
87A									104.5	11.2	266.7	25.5
									+	29.0	29.4	
TAL		0	16 51		J	4	6565.4	287.5	16136 3	TERD. 6	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	3E 4

LMSC-A847943 Vol. I

8-7

LOCKHEED MISSILES & SPACE COMPANY

	JC4EN11715		TE MOTODYE	- - - -		- CHATCOTH LO H	0		
тне	: FOLLOWING FISCAL YE	AR COSTS W	ERE CALCULATED FOR	t THIS LUNAR (EXPLORATIO	N PROGRAM	-		
I SCAL YEAR	х о х К С С С К С С К К С С	ING COFFC	COST SUMMARY (MI 0 S T S TOTAL	R E C U R R	I N G C	O S T S Total	FY ANNUAL COST (MIL.)	CUM. Cost (Bil.)	
1967	8 • t	;	4.8	Ale and a second second second second second second second second second second second second second second se		anter a la ser de la ser a compañía	4°8		
1968	92.1	1.1	93.2	76.8	:	76+8	169.9	•2	
1969	233.0	6 •	233.9	485.6	10.5	496.1	730.0	6.	
1970	165.3	.	165.6	862.0	93.7	955.7	1121.3	2.0	
1971		1.7	114.9	A65.0	203.4	1068.4		3.2	
1972	199.3	5.0	204.3	786.2	186.9	973.1	1177.4	t * 7	
1973	406.1	31.0	437.1	791.5	159.3	950.8	1388.0	5.8	
1974	624.0	30.0	654.0	729.6	217.3	946.9	1600.9	7.4	
1975	575.0	· • •	575.1	859.9	137.0	6*966	1571.9	A.9	
1976	163.6	1.1	164.7	976.3	199.1	1175.4	1340.1	10.3	
1977		5•3	96.3	951.8	223.7	1175.5	1271.8	11.6	
1978	357.8	50.0	407 . B	799.5	249.1	1048.6	1456.4	13.0	
1979	6.1.9	115.4	1057.3	781.4	233.0	1014.4	2071.7	15.1	
1940	1137.0	38.8	1175.8	1.166	165.1	1156.8	2332.6	17.4.	
1981	. 658.0	6.8	664 . R	1064.1	212.1	1276.2	1941.0	19.4	
1982	264.8		264.8	696	224.7	1194.0	1458.7	20.A	
1983			302.8	1030.7	215.0	1245.7	1548.5	22.4	
1984	208.0		208.0	928.5	25 3 .5	1182.0	1390.0	23 . A	
1985	27.8		27.8	665+5	247.9	913.4	941.2	24.7	
1986				358.1	204.9	563•0	. 563.0	25,3	
1987				151.3 .4	115.4 29.0	266.7 29.4	266.7 29 <u>.4</u>	25•5	
TOTAL	6565.4	287.5	6852.9	15125.2	3580.6	18705.8	25558.7	25.6	

ſ

LMSC-A847943 Vol. I

8-8

LOCKHEED MISSILES & SPACE COMPANY

		SCIENTIFI	C PROGRAM	A7	MISSI	SU MIX 03		NUMBER	OF MISSIONS =	: 26			
THE	FOLLOWI	NG SUMMAR	IZES EXPLOR	ATION EC	QUIPMENT	COSTS CALCU	LATED FOR	נ דאוג רט	μάα εχριοκάτι	ION PROGRA	•		i
		ROV	SYSTEM 242 ING VEH 3 M	3-03 AN 90 D	AYS	-		TRA	SYSTEM 2433- ILER-MULTIPUF	-XX RPOSE-LRG		1	
 	R / D	CURRING C OF F	NO. PROCU MISS T/L	RED SPR	RECURI	TING OPNS	NONREC R / D	C OF F	NO. PROCUR	R HOW	ECURPING E OPN	S	
67A													
68A								1		ı			:
69A	18.0												
70A	28.0	- ~ ~											
71Å	22.0	8	-		,		-					-	
72A	55.0												
73A	#0°0						·						
744	30 0 25 0				24.5		•	:		:			: : :
1740	17.0			1	27.6	1.4 4	5	-					
76.0					19.8	1.5	12.0	~~					
,			• • • •	-	15.5	1.5	13.0	60		• œ	ט ט ו		
17A			٩		21.9	8° 1'	8.3	•2·	1	1 4.		~ ~	
78A					10.5	2.3				-	•	,	
79A			1		8.4 10.3	8.				N 3	~ ~		
And	1		-	•	17.6				1		*		
			• -• •	1	19.7	5.2					= 1	(
			-		12.8	8.					1.	5	
182A	·				18.2 13.6				-			-4	
183A			-		10.0	1.5					 		! !
184A			4		5.5	c.			-	÷'n	N 6		
185A						e.			1	2.	~		
186A		1		•						: :			
NA A													
T A1	421.0	0			(1					

LMSC-A847943 Vol. I

The cost printout has five major parts: a comments page, a breakdown of costs by equipment item and by mode, a schedule and funding summary given by 6-mo increments, a funding summary by fiscal year, a launch rate distribution table, and a usage and funding summary for each item of equipment. The computer provides almost 50 pages of printout; only 7 sample pages are shown in the figure. The remainder consists of cost and procurement histories of each mode and system used in the program in a format similar to Page 8-1, Fig. 8-1.

The annual funding summary shows a peak post-Apollo funding requirement of over \$2.3 billion in FY 1980. This peak, which could be smoothed by adjustment of mission start dates, is caused by combined costs for nonrecurring R&D and recurring equipment procurement at the time of the second Saturn uprating.

LOCKHEED MISSILES & SPACE COMPANY

Chapter 9 PROGRAM DATA SUMMARY

In the MIMOSA analysis procedure, a lunar exploration planning question is answered by studying one or more associated candidate lunar programs. To do this efficiently requires means of collecting and displaying the pertinent calculated data. This chapter suggests a standard format for data display in the form of an Exploration Program Summary.

The experience gained from the MIMOSA contract study resulted in a special summary format for presenting program data considered of greatest value to planners. A sample of this format is given in Appendix A. The format consists of the following seven parts:

- <u>General Description</u> gives a brief description of the exploration program and defines the assumptions and guidelines used for program generation
- <u>Scientific Program Description</u> gives a general verbal explanation of the science program objectives
- <u>Lunar Surface Exploration Pattern</u> displays locales and paths for a program on a lunar map
- <u>Cost and Schedule Summary</u> summarizes the major program cost and schedule data. The Saturn launch rate appears above a bar chart showing the timing of the major steps in the equipment evolution. Numbers inside each bar indicate the number of missions performed each year. In the middle of the summary is a graphical display of nonrecurring and total costs by fiscal year. A table at the bottom shows the breakdown of the required annual funding.
- <u>Post-Apollo Mission Summary</u> identifies and tabulates data on each mission in the program. The mission summary lists mission dates,

mission locations, mode used, number of launches, staytime, total and scientific manhours, total cargo and scientific mass, mass reserve, and total traverse range. Total cargo mass refers to the total useful landed mass (mission equipment + scientific equipment). The mass reserve represents the unused mass delivery potential of the mode (i. e., landed payload capability less total cargo) and indicates the "overkill" of the mission by mode. Large values occur in the example in Appendix A because a fixed scientific program size was established before any missions and modes were matched. The resulting missions did not fully tax the mode capabilities. In practice, the scientific requirements would be tailored to produce more efficient usage of the mode capability. The traverse range given in the mission summary is the actual distance traveled in the performance of the mission, and this often exceeds scientific requirements

- <u>Equipment Usage and Cost Summary</u> records the major items of exploration equipment used in a program, details the quantities procured, and displays the resulting cost breakdown for each item. The identification numbers given for individual exploration equipment allow easy reference to the MIMOSA Data Book for detailed descriptions and performance data
- <u>Scientific Program Operation Summary</u> displays cumulative scientific manhours and mass for a program and relates these to cumulative cost as a function of calendar time

9-2

LOCKHEED MISSILES & SPACE COMPANY

Chapter 10

MIMOSA COMPUTER PROGRAM INPUT INSTRUCTIONS

The MIMOSA Program will require the routine handling of large amounts of technical and resource data. A computer program has been devised to assist the data organizing task and to perform certain calculations. Both of these tasks would require many manhours of tedious labor if done by hand.

A computer performs only the instructions it is given and acts only on the data it is presented. The purpose of this chapter is to present the steps that must be followed to provide the computer with the proper data and instructions.

10.1 GENERAL COMPUTER INFORMATION

The MIMOSA computer program has been written in FORTRAN IV for the Univac 1107 and is compatible with the faster Univac 1108. The program consists of three major routines as shown in Table 10-1.

The logic instructions to run the program are stored on reels of magnetic tape. The instructions for the first and third routines are written on one reel. The Mission/Mode Comparison Routine is written on another reel. A list of all the instructions on these tapes is held by NASA/MSFC. Flow diagrams of the program logic for routines 2 and 3, are presented as separate Appendixes (K and L) to this volume.

All input data used by the three routines are stored on two additional reels of magnetic tape. The input data for a given lunar exploration program are grouped into six categories on the two tapes in the order shown in Table 10-2.

÷

Table 10-1

MIMOSA COMPUTER PROGRAM ROUTINES

No	Routine	Functions	Approxi- mate Number of	Estimated (min	Run Time
no.	name		Instruction Cards	. 1107	1108
1	Input and Update	 Generates memory tapes from card inputs Updates tapes Copies all or part of any data record 	7,000	15 min to perform any function	3 min to perform any function
2	Mission/ Mode Comparison	 Compares mode capabilities with mission require- ments (~45 tests) Indicates possible problems in match- ing mode to mission 	9,000	12 min for ~ 30 com- parisons	2-1/2 min for ~ 30 com- parisons
3	Integration and Cost	 Establishes lunar program schedule Determines systems used vs. projected time Calculates non- recurring and recurring costs vs. projected time 	8,000	12 min for ~ 30 mission programs	2-1/2 min for ~ 30 mission programs

s.

Table 10-2

DATA FILES USED BY MIMOSA COMPUTER PROGRAM

Reel	File Order	Max. No. of Entries Per File
Mission/Mode Tape	Missions Exploration Concept Modes Emergency Modes	50 100 25
System/Experiment Tape	Systems Subsystem State-of-the-Art Advances Experiments	200 50 500

A new Mission/Mode Tape is normally generated for each different lunar exploration program investigated because of the limited number of missions permitted on a tape. However, a common System/Experiment Tape can usually be maintained for all programs.

10.1.1 Man/Computer Interface

Figure 10-1 presents the man/computer interface associated with the MIMOSA computer program. The program planner must perform the following three principal tasks:

- Select the magnetic tapes to be used
- Prepare control cards (for program control)
- Prepare data cards

The computer assigns tape and drum units based on instructions in the control cards. It then reads in the appropriate program tape and initiates the Complex Utility Routine (CUR) built into the computer by Univac.



.*

10-4

As shown in Fig. 10-1, the computer program is written so that the Integration and Cost routine may be used back-to-back with the Input and Update Routine. Any other combination of routines requires submittal of separate sets of control cards.

10.1.2 Tape Handling Philosophy

All computer instructions and input data required by the MIMOSA computer program are stored on reels of magnetic tape. Figure 10-1 lists the total assembly of tapes that may be utilized by the computer. Only Tapes 3, 4, and 5 (Latest Mission/Mode, Latest System/[¬]xperiment, and Program CUR) are required for operation of the Mission/Mode Comparison and the Integration and Cost Routines.

Tapes 1 or 2 (Old Mission/Mode or Old System/Experiment tapes) are needed only if any part of the data on one of these tapes is to be transferred to another tape. This occurs when updating or copying all or part of a data file. For example, if it is desired to update a mission contained on an existing Mission/Mode Tape, the existing tape becomes the Old Mission/Mode Tape. All mission and mode data contained on the Old Mission/Mode Tape are read into the computer, the revision is made, and the updated files are written onto the Latest Mission/Mode Tape. The planner ends up with two tape reels of mission/mode data. The old tape can be a backup to the latest tape, or the old tape can be reused if a mistake occurs in the updating operation.

The List Tape (tape 6) is used to store the computer output for subsequent printout. The Univac 1108 uses on-line printers so that a list tape is only needed if more than one copy of the output is desired.

10.1.3 Sequence Numbers

Another key concept in the operation of the MIMOSA program is the use of sequence numbers. Sequence numbers are punched into columns 77-80 of nearly every input

data card.* The computer program requires sequence numbers for the following reasons:

- (1) The Univac 1107/1108 cannot read any card properly unless it knows in advance the format in which every item is written on that card. It must know whether it is going to read letters or numbers and, if numbers, whether they are in decimal or exponential form. Sequence numbers provide a reference for the computer and allow the computer program to determine the proper format for each control card or data card.
- (2) The data files for modes, emergency modes, systems, subsystem stateof-the-art (SOTA) advances, and experiments are fixed in length. This means that the precise location on tape or drum storage is known for every possible bit of data allowed in each of these files. Sequence numbers are used to assure that each data bit is placed in its proper location.

The sequence number on an input card takes the form:

ABCD

- where A = data category number (i.e., 2 = control, 3 = mission, 4 = mode, 5 = system, 6 = subsystem SOTA advance, 7 = emergency mode)
 - BC = subfile number
 - D = number to indicate location of the data on the input card in relation to the rest of the data for the subfile

A subfile is a set of related information in a data file. For example, in the system file, separate subfiles are maintained such as general data, nonrecurring R&D cost, nonrecurring cost of facilities, recurring procurement cost, recurring operations costs, expendables usage, and manhours usage.

^{*}Some cards for the mission data file are excepted; SKIP and LAST cards are used instead. (See paragraph 10.1.4.) Cards for the experiment data file are excepted because only two cards are input for each experiment.
D numbers run consecutively up to the last card for the subfile. For the last card, an "L" is punched instead of a number. This special signal alerts the computer of subfile completion. It should be emphasized that sequence numbers require special attention. Mistakes in sequencing cause the computer to lose its way. If a mistake is encountered, the computer has been programmed to print a diagnostic message and to skip to the next meaningful set of cards; if this is not possible, abort of the computer run follows immediately.

10.1.4 SKIP, LAST, and STOP Cards

Other special sequence cards are required for the proper assimilation of data by the computer. These cards have only the letters SKIP, LAST, or STOP, punched in columns 77-80.

SKIP and LAST cards have a major influence on mission data input. A mission data file is variable in length. This means that each element of data is packed on the computer memory tape or drum with no gaps. Because some missions are small and others large, the precise address of any particular bit of data is unknown. The computer, therefore, needs a guide to help it find the limits of any mission subfile; the SKIP card is used for this purpose.

The data for any mission is arranged in the following order (see also paragraph 10.5.1):

- (1) General mission data (one card fixed amount of data)
- (2) Experiments to be performed at base (one card per experiment)
- (3) General data for first traverse (one card fixed amount of data)
- (4) Experiments to be performed on first traverse (one card per experiment)
- (5) General data for second traverse (one card fixed amount of data)
- (6) Experiments to be performed on second traverse (one card per experiment).

The sequence can continue for up to 24 traverses.

Obviously, the computer must be told when the subfile on a new traverse is being started. The SKIP card alerts the computer that "data on a new traverse follows." It must precede each general data card for a new traverse.

The LAST card signals the computer that data input on each mission has been completed and to prepare for data on a new mission. LAST cards also signal the end of data input for each mode, emergency mode, system, and subsystem SOTA advance (Note: experiments are excepted since each has a fixed number of input cards).

A STOP card signals the computer that data input has ended for each input category. Thus, a STOP card is used (in place of the LAST card) when data on the <u>final</u> mission, mode, system, etc., in the file is input. The STOP card has the same message for every file (including experiments). In addition, STOP cards are placed at the end of the set of control cards for the Mission/Mode Comparison Routine and the Integration and Cost Routine.

Any combination of two or more SKIP, LAST, and STOP cards back-to-back will abort a computer run. These cards, along with the sequence numbers on the data cards, guide the computer through the input data, and, incidently, serve as important aids in rapidly stacking cards for computer input.

10.2 STACKING CARDS FOR COMPUTER RUNS

For any single run, only one of the following computer routine combinations may be employed:

- Input and Update Routine only
- Mission/Mode Comparison Routine only
- Integration and Cost Routine only
- Input and Update Routine followed by Integration and Cost Routine

Separate runs must be made to obtain other combinations.

The Input and Update Routine operates on the Mission/Mode data tape and the System/Experiment data tape. The routine can only do <u>one</u> of two things to <u>each</u> tape on any single run: generate a new tape from cards, or revise an existing tape. Hence, if any one of the three data files on either tape is to be generated from cards, all three data files must be generated from cards.

Figures 10-2 through 10-5 display the card stacking required for each of the above computer routine combinations. Note that card stacking can be done on a building-block concept. The procedure for stacking each block of cards can be found in the subsections indicated.

10.3 GENERAL INPUT INSTRUCTIONS

At the beginning of a computer run with the Univac 1107/1108, cards are required that perform three functions:

- Ensure proper cost accounting for use of the computer
- Establish that tapes are properly assigned
- Begin the program

Table 10-3 presents the cards required by Lockheed Computer Operations. Similar cards will probably be required by other organizations operating this program. Cards 5 through 12 should be standard on all machines.

10.3.1 Tape Assignment

Because all data and instructions for the MIMOSA Program are on tape, it is essential that there be proper identification of tape usage. For Lockheed submittals, three cards (No. 1, 5, and 11 in Table 10-3) are devoted to tape assignment.

Possible confusion on tape usage occurs because the three principal tape handlers (the tape library, the computer itself, and the computer operator) identify tapes by different methods:

• The libraries at most computer centers store tapes according to a specific inhouse numbering system. Each reel of tape has its own number.



Fig. 10-2 Computer Card Stack - Input and Update Routine Only

10-10















LMSC-A847943 Vol. I

Number	Name	Card Punch (Column)					n)			Function				
		1	2	3	4	5	6	7	8	9	10	11	12	Function
1 2	Operations Request Run Card					Se	e No	te A						Standard form request for computer run (Lockheed only) Establishes max computer run time allowed and max number of output pages (Lockheed only)
3	Output Tape Request	7 8	D	s		T.	P	R ,						Sets up machine to write Fortran output on magnetic tape. (This card used only when output tape is desired.)
4	LID Card			Loc	khe	ed I	denti	ificat	tion	Car I	d I			Standard LMSC cost accounting card
5	Tape Assignment	7 8	R	φ	н			A	s	G				Tape setup assignment, and check to ensure proper tape use. See Section 10.1.2 for additional instructions.
6	Execute CUR	7	L		x	Q	т		с	U	R			Initiates Complex Utility Routine in machine.
7	CUR Tape							I	N		D			Reads in MIMOSA CUR tape
8	Tape Rewind							т	R	I		D		Rewinds and interlocks CUR tape
9	Table of Contents							т	ø	с				Sets up printout of subroutines on the CUR tape which gives the date that sub- routine became effective.
10a	Subroutine Execution	7 8		x	ବ	т		М	I	м	ø	s	A	Initiates start of Input and Update Routine or Integration and Cost Routine (used only if one or both of these routines are desired).
10b	Subroutine Execution	7 8		x		т		м	N	Р	R	ø	G	Initiates start of Mission-Mode. Com- parison Routine (used only if this Routine is desired).
11	Control Card 1-01				s	ee 7	able	9 10-	 5 					Establishes for the program which tapes are to be used and purpose of each tape.
12	Control Card 1–02				s	ee 1	able	e 10-	 6 					Establishes for program logic which MIMOSA ROUTINE is desired.

GENERAL INPUT INSTRUCTION CARDS

NOTE A: The first fourteen columns of this card are punched standard by Lockheed Computer Operators. The following is punched beginning in Column 15:

, XXX YYY

where

XXX = run time in minutes (e.g., 005, 010) YYY = max page count (e.g., 050, 100, 400)

For most computer submittals, XXX = 005 and YYY = 400 will suffice.

- The Univac 1107 and 1108 identify tapes being used on any given run with a two-digit numbering system. For example, a tape 16 must be read, a tape 17 is needed for additional data, a tape 25 must be written on, etc.
- The computer operator handles tapes only by letter (tapes A, B, C, etc.).

Fortunately, Univac has established an equivalence between the letters used by the computer operator and the two-digit numbers internal to the machine. Table 10-4 shows the letter and number designations that apply to the MIMOSA tapes.

Tape Name	Letter Code Assigned	Equivalent Tape Unit No.
Old Mission/Mode Tape	В	16
Old System/Experiment Tape	C	_17
Latest Mission/Mode Tape	E	25
Latest System/Experiment Tape	F	26
Program CUR Tape	D	18

Table 10-4 MIMOSA TAPE ASSIGNMENTS

It is necessary to ensure that the proper reel of tape coming from the library is assigned the right letter code by the operator. For example, a reel of tape containing mission and mode data must be assigned as either B or E. At Lockheed the letter code to be assigned to each tape requested from the library for a given run is designated on the Operation Request Card (card 1, Table 10-3). The same designation is punched into the Tape Assignment Card (card 5, Table 10-3). If the assignments on the two cards do not agree, the computer operator aborts the run.

Beginning in Column 15 of the Tape Assignment Card, the tapes to be used are assigned letter codes in the following format:

Column 15

B = XXXX, C = YYYY, E = ZZZ, etc.

with no spaces permitted. The letters XXXX, YYYY, etc., represent the numbers given the tape reels by the library. If it is not desired to use an Old Mission/Mode Tape or an Old System/Experiment Tape, they are not mentioned on this card.

It is also necessary on another card to tell the computer which Mission/Mode and System/Experiment tapes it will be operating on. This is done with Control Card 1-01 (card 11, Table 10-3). Table 10-5 gives the required punch inputs for this card depending on the tapes desired.

Table 10-5INSTRUCTIONS FOR FILLING OUT CONTROL CARD 1-01

If it is desired to use any of these tapes \rightarrow	Old Mission/Mode	Latest Mission/Mode	Old System/Exper.	Latest System/Exper.	
Punch these corresponding numbers →	16	25	17	26	
Into these columns \rightarrow	1, 2	3, 4	5,6	7, 8	

For example, the Mission/Mode Comparison Routine uses only the Latest Mission/ Mode and Latest System/Experiment tapes. Control Card 1-01 for this routine would be punched as follows:

Column	1	2	3	4	5	6	7	8
Punch			2	5			2	6

10.3.2 Selection of Routine

Com

Control Card 1-02 instructs the machine to call in the MIMOSA program routine that is desired. As discussed in subsection 10.2, only four possibilities are permitted for any one run. These are shown in Table 10-6.

	Table 10-6					
INSTRUCTIONS FOR FILLING OUT CONTROL CARD 1-02						
uter Routine Desired	Required Input on Control Card 1-02 (Column)					
	1 2 3 4 5 6 7 8 9 10 11 12 13 14 15 16 17 18					

Computer Routine Desired	1 2 3 4 5 6 7 8 9 10 11 12 13 14 15 16 17 18 19
Input and Update	UPDATE
Mission/Mode Comparison	CØMPA R
Integration and Cost	INTEG
Input and Update followed by Integration and Cost	UPDATE INTEG

10.4 INPUT AND UPDATE CONTROL CARD

The Input and Update Routine of the MIMOSA Program is used to input and revise the great amount of technical data used by the other two computer routines. Seven options are available. These are summarized in Table 10-7.

A Control Card 2-01 is used to specify the options to be performed by the computer. Table 10-8 gives the instructions for filling out this card. Three key rules must be followed:

1. If one of the options is required for one data file (see Table 10-2) on a data tape, that option or some other option must be specified for the other two files on that tape. For example, if it is desired to UPDATE Systems only, then the Copy option must be used for Subsystem SOTA Advances and for Experiments.

OPTIONS FOR INPUT AND UPDATE ROUTINE

Name	Code Name	Purpose
Generate	GENER	Generate a data file on tape from card input
Update	UPDATE	Update or modify a data file from card input
Delete	DELETE	Remove a set of data records from an existing data file
Insert	INSERT	Insert (i.e., add) a new set of data records to an existing file
Сору	СОРҮ	Copy a complete data file on a category from an Old tape onto a Lates tape without modification or change of order.
Copsel	COPSEL	Copy a selected subset of data records from an Old tape onto a Latest tape without modification, but in any desired order, to form a new data file.
List	LIST	Obtain a printout of the contents of the Old Data file for any category

Table 10-8

INSTRUCTIONS FOR FILLING OUT CONTROL CARD 2-01

Таре	File	Begin Update Option Code Name (Table 10-7) in Column	If Print of Tape File Before Update is Desired, Write ' LIST in Columns
Mission/Mode	Missions Modes Emergency Modes	1 7 13	43-46 47-50 51-54
System/ Experiment	Systems Subsystem Experiments	25 31 37	59-62 63-66 67-70

IMPORTANT: Be sure to input sequence number "201L" in Columns 77-80.

- 2. If the generate option (code name GENER) is used for one file on a data tape, the generate option must be used for <u>all three</u> files on that tape. This means that the Old tapes (Tapes 1 and 2 in Fig. 10-1) are not used, and that the Latest tapes (Tapes 3 and/or 4) are generated directly from cards.
- 3. If it is desired to LIST the file that exists for one of the files on the Old tape, then one of the other options (except GENER) must be exercised for all the files on that tape. The most common option is to COPY the files from the Old tape on to a Latest tape.

10.5 MISSION-MODE TAPE GENERATION

The Mission/Mode Tape contains three data files: missions, modes, and emergency modes. To generate one of these data files from cards requires generation of all three data files. Figure 10-6 shows how the cards are stacked.

10.5.1 Mission Input

Table 10-9 summarizes the input cards that must be filled out for each mission used in the computer program. Figure 10-7 shows how these cards would be stacked for three example missions.

The following five important rules must be observed in setting up the cards:

- If there are no base experiments on the mission, 3-02 control cards should be omitted.
- A SKIP card must precede every 3-03 card.
- At least one 3-04 card must follow each 3-03 card.
- A LAST card must follow the final data card for each mission except the last mission. A STOP card must follow the data on the last mission. If the mission is to have no experiments, the LAST or STOP card may follow the 3-01 card.
- The experiments intended for a particular traverse must follow the 3-03 card for that traverse.



10-20





10-21

MISSION INPUT CARDS

Card ID No.	Card Name	Maximum Cards	Se ((1	Sequence No. (Columns) 77 78 79 80			Comments
3-01	Mission Input — General	1	3	0	1	L	Use one card per mission.
3-02	Mission Input — Base Experiments	100		*			Use one card per base experiment; 100 experi- ments permitted. Omit cards if no base experi- ments are desired.
3-03	Mission Input – Traverse Data	24	3	0	3	L	Use one card per traverse. Input flying traverses first. This card must always be preceded by a card with SKIP written in columns 77-80
3-04	Mission Input – Traverse Experiment	50		*			At least one card, but no more than 50 cards, must follow each 3-03 card.

*Sequence number is not needed.

Examination of the instructions for the 3-2 and 3-04 cards shows much commonality between the two required inputs. If care is taken, the same card can be prepared for an experiment whether that experiment is used at base or on a traverse.

Appendix B presents the detailed instructions for filling out each mission input card. This appendix was written by a computer reading punched cards. This computergenerated documentation will be relatively easy to maintain and to keep up-to-date.

10.5.2 Mode Input

Table 10-10 summarizes the input cards that must be filled out for each exploration concept mode used in the MIMOSA program. All cards must be input in the order shown. With the exception of Card No. 4-20, at least one of each card type must be filled out for every mode. Card 4-20 may be omitted if an emergency mode identification is not desired.

Instructions for filling out each card type are given in Appendix C. A LAST card (see paragraph 10.1.4) must be placed at the end of the set of cards for each mode except the last mode. A STOP card is placed at the end of the last mode to be input.

10.5.3 Emergency Mode Input

The use of the emergency mode concept is an optional feature of the MIMOSA Program. This concept allows a crew rescue emergency mode and a logistic emergency mode to be identified (see Card No. 4-20) for each exploration concept mode. When each exploration concept mode is used the first time in a lunar program, the equipment association with each of its emergency modes is procured once. The purpose of the emergency-mode input is to give the required costs and to identify those systems that must be procured.

MODE INPUT CARDS

Card		Maximum	Sequence No. (Columns)				
ID No.	Card Name	Cards	77	78	79	80	
4-01	Mode Input – General	1	4	0	1	L	
4-02	Mode Input – Operations	1	4	0	2	L	
4-03	Mode Nonrecurring R&D Cost Curve	3	4	0	3	*	
4-04	Mode Cost of Facilities Curve	3	4	0	4	*	
4-05	Mode Recurring Hardware Cost Curve	3	4	0	5	*	
4-06	Mode Operations Support Cost Curve	3	4	0	6	*	
4 -07	Mode Input – Launch Identification	8	4	0	7	*	
4-10	Mode Input – Systems on Each Launch	120	4	1	**	*	
4-20	Mode Input – Emergency Modes	1	4	2	0	L	

*Column 80 = 1, 2, ... L, depending on number of cards needed. (The letter "L" is always used on the last card of a series.)

**Column 79 on Card No. 4-10 is the launch identification number.

Table 10-11 summarizes the input cards that must be filled out for each emergency mode. All cards must be input in the order shown.

Appendix D gives instructions for filling out each card type. A LAST card must be placed at the end of the set of cards for each emergency mode except the last mode. A STOP card is placed at the end of all emergency modes to be input.

Emergency modes may be ignored in the cost of any lunar exploration program by proper input of one of the control cards (see paragraph 10.10, Card No. 2-21) of the

Integrating and Cost Routine. If it is desired to ignore emergency modes, the following should be done:

- (1) Input a card with:
 - DUMMY in columns 1-5
 - 701L in columns 77 80
- (2) Place a STOP card behind the dummy card to signal the end of emergency mode input.

Table 10-11

EMERGENCY MODE INPUT CARDS

Card	Card Name	Maximum	Sequ	ence l	No. (C	olumns)
ID No.		Cards	77	78	79	80
7-01	Emergency Mode Input – General	1	7	0	1	$\mathbf L$
7-03	Emergency Mode Nonrecurring R&D	3	7	0	3	*
	Cost Curve					
7-04	Emergency Mode Cost of Facilities	3	7	0	4	*
	Curve				i	
7-05	Emergency Mode Recurring Hardware	3	7	0	5	*
	Cost Curve					
7-06	Emergency Mode Operations Support	3	7	0	6	*
	Cost Curve		_			
7–11	Emergency Mode Input – Systems	15	7	1	1	*
	Procured					

*1, 2, ... L, depending on number of cards needed. (The letter "L" is always used on the last card of a series.)

10.6 SYSTEM/EXPERIMENT TAPE GENERATION

The System/Experiment Tape contains three data files: system, subsystem state-ofthe-art advances*, and experiments. To generate one of these data files from cards requires generation of all three data files. Figure 10-8 shows how the cards are stacked.

10.6.1 System Input

The data on exploration equipment (i.e., systems) used for lunar programs are placed first on the System/Experiment Tape. Table 10-12 summarizes the input cards required for each system. Ten of the 16 card types listed are required input for all systems. For certain groups of systems, one or two additional cards must be input. For example, the input for a lunar flying vehicle (LFV) requires cards 5-01 through 5-10 plus 5-13 and 5-14.

Instructions for filling out each card type are given in Appendix E. A LAST card must be placed at the end of the set of cards for each system except the last system. A STOP card is placed at the end of all systems to be input.

Three points require special attention:

 Certain system identification numbers trigger subroutines in the computer program. The following identifiers must be used:

11XX-XX	=	Transportation systems
121X-XX	=	Non-Saturn launch vehicles
1213-XX	=	Saturn IB launch vehicle
122X-XX	=	Saturn V launch vehicle

^{*}SOTA advances refer to large development programs required to advance technology to ensure the feasibility of a particular subsystem. For the Data Book, they may be treated in the same manner as systems.





10-27

123X-XX t	hru	127X-XX = Saturn V upratings
222X-XX	=	Manned orbiters
232X-XX	=	Personnel shelters
24XX-XX	=	Roving vehicle of any type
243X-XX	=	Roving vehicle trailer
25XX-XX	=	Flying vehicle
2711-XX	=	Fuel resupply for flying vehicles
2712-XX	=	Expendables resupply
2713-XX	=	Fuel regeneration systems
3XXX-XX	=	Major scientific equipment

.

Table 10-12

SYSTEM INPUT CARDS

Card ID	Card Name	Maximum	Where	Sequence No. (Columns)			No. ns)	Comments	
No.		Cards	Applicable	77	77 78 79		80		
5-01	System Input – General	1	All systems	5	0	1	L		
5-02	System Input – Sub System SOTA ID	3.		5	0	2	*		
5-03	System Cost Curve	3		5	0	3	*	Nonrecurring R&D	
5-04	System Cost Curve	3		5	0	4	*	Nonrecurring Cost of Facilities	
5-05	System Cost Curve	3		5	0	5	*	Recurring hardware procurement	
5-06	System Cost Curve	3		5	0	6	*	Recurring operations support	

LMSC-A847943 Vol. I

.

Table 10-12 (Cont.)

Card ID	Card Name	Maximum Cards	Where Applicable	Sequence No. (Columns)				Comments
No.	7		77	78	79	80		
5-07	Percent Cost Reduction	2	All systems	5	0	7	*	Learning curve
5-08	R&D Flight Table	2		5	0	8	*	Development flights
5-09	System Expendables	5		5	0	9	*	One card for each crew level
5-10	System Manhour Requirement, Nonscientific Operations	5	All systems	5	1	0	*	One card for each crew level
5-11	System Input – Personnel Shelter Only	1	Shelter only	5	1	1	L	
5-12	System Input – LR Only	75	LRV only	5	1	2	*	One card for each crew level
5-13	System Input – LFV Only	5	LFV only	5	1	3	*	One card for each crew level
5-14	System Input – LFV Stops vs. Range/Table	2	LFV	5	1	4	*	
5-15	System Input – Major Science Equipment Only	5	Major scien. eqpmt. onl	5 y	1	5	*	

Table 10-12 (Cont.)

Card ID	Card Name	Maximum Cards	mum Where cds Applicable		que Col	nce um	NO. ns)	Comments
NO.				77	78	79	80	
5-16	System Input – Transportation System Only	1	Trans. system only	5	1	6	L	Identifies flight systems

*1, 2, ... L, depending on the number of cards needed. (The letter "L" is always used on the last card of the series.)

- (2) When a trailer is combined with a rover in a tandem rover-trailer configuration, that combination exhibits different performance characteristics than either the rover or trailer acting by itself. To ensure proper performance data input, each trailer concept can be input into the computer by use of one of several identification numbers. The first four digits of the identification number establish the trailer concept and the last two digits establish how the trailer is to be used in the mode. Thus, 2436-01 stands for Trailer 2436 used by itself, 2436-02 stands for Trailer 2436 used in combination with LRV "A", 2436-03 stands for use in combination with LRV "B", etc. The performance data input with each identifier is the performance of the trailer or the particular trailer-rover combination. This is different for each identifier. However, the same cost data can be input for all variations of a given trailer concept if it is assumed that the cost of a particular trailer concept is independent of the manner in which it is used.
- (3) Unmanned systems are treated as manned systems with a crew level of zero.
 If a system can operate both manned and unmanned, inputs for expendables (Card 5-09), nonscientific manhour requirements (Card 5-10), LRV-only data (Card 5-12), and LFV-only data (Card 5-13) may be made for zero as well as other crew levels.

10-30

10.6.2 Subsystem State-of-the-Art Advance Input

The ability to display and separately cost advances in the state-of-the-art (SOTA) of subsystem concepts is another feature of the MIMOSA computer program. Use of this feature results in the tabulation of nonrecurring R&D and cost of facilities expenditures vs. projected time for each advance required in a given lunar program.

The subsystem SOTA advances required to achieve a given system are listed for that system using the 5-02 card (see Table 10-12). Table 10-13 summarizes the cards that must be filled out for each SOTA advance that is identified. Instructions for filling out these cards are given in Appendix F. A LAST card must be placed at the end of the set of cards for each subsystem advance except the last subsystem. A STOP card is placed at the end of all subsystems to be input.

If no subsystem SOTA advance is called out in the entire system file, the following should be done:

- (1) Input a card with:
 - DUMMY in columns 1-5
 - sequence number 601L in columns 77 80
- (2) Follow the dummy card with a STOP card to signal the end of the subsystem file.

Card ID	Card Name	Maxi- mum	Sequence No. (Columns)				
No.	ouru numo	Cards	77	78	79	80	
6-01	Subsystem Input – General	1	6	0	1	L	
6-02	Subsystem Cost Curve – Nonrecurring	3	6	0	2	*	
6-03	Subsystem Cost Curve – Nonrecurring Cost of Facil- ities	3	6	0	3	*	

Table 10-13

SUBSYSTEM SOTA ADVANCE INPUT CARDS

*1, 2,..., L, depending on number of cards needed. (The letter "L" is always used on the last card of a series.)

10 - 31

10.6.3 Experiment Input

The format for input of data on scientific program experiments was derived by North American Aviation under NASA contract NAS-8-20258 (Scientific Mission Support for Extended Lunar Exploration). North American chose six types of data cards with which to describe an experiment for computer handling. Lockheed selected two of these six for use in MIMOSA.

> Type 1 Card – Experiment Summary Type 3 Card – Experiment Operations

A single Type 1 card presents summary information on an experiment. This information includes the experiment name, equipment mass, development time, and cost.

A single Type 3 card presents the number of manhours estimated to perform an experiment. The performance of an experiment is broken up into four phases:

> Phase 1 - Setup Phase 2 - Operate Phase 3 - Analysis Phase 4 - Takedown

Shirtsleeve and spacesuit manhours at base or on traverse (as needed) are given for each phase. The manhours, however, are for performance of an experiment in a shirtsleeve environment on Earth. They have to be multiplied by a special factor to compensate for lunar gravity and (where applicable) for spacesuit use. The factors are input with one of the control cards for the Mission/Mode Comparison Routine (see paragraph 10.9).

North American also adopted special formats for presentation of numbers on a data card. Two of these formats applicable to MIMOSA are shown in Table 10-14.

10 - 32

FORMATS USED FOR NUMBERS ON EXPERIMENT INPUT CARDS

Format Type	Format	Meaning
. 1	A B 0 C A B 5 C	$A.B \times 10^{C}$ $A.B \times 10^{-C}$
2	A B	$A \times 10^{B}$

Appendix G presents the instructions for preparing the two cards for each experiment. It is essential that both cards be input for every experiment. No LAST card is required after each experiment; however, a STOP must follow the final Type 3 card.

10.7 MISSION INPUT REVISION

The discussion in paragraphs 10.5 and 10.6 has centered on making original data inputs to the computer. The next two sections discuss how to revise the data on an existing tape.

The space reserved on the magnetic tape or computer drum for data concerning any specific mission varies with the quantity of that data. The data is packed in much the same way as taking several shopping lists and attaching them end-to-end. This "variable length" format is required because up to 15,779 data cells would have to be set aside for a single mission, and to fix this length for 50 missions would saturate the computer core storage.

"Variable length" format places a severe restriction on updating. <u>No experiment or</u> <u>traverse can be added or removed from any given mission</u>. To do so would require moving all following data to new locations on the tape. This is a most complicated maneuver.

Another restriction must be born in mind when using the updating procedure. <u>It is not</u> <u>possible to correct or update the identifier of a mission or of an experiment called out</u> <u>for that mission</u>. This is because of several checks built into the logic to assure proper card sequencing. In order to correct any one identifier, a whole new mission, with a different mission identifier, must be added to the tape file.

There are four updating options possible for missions:

- Change existing data (except identifiers)
 - Delete whole missions
- Add whole missions
- Copy a selected subset of missions onto a new tape in any order

10.7.1 Procedure for Changing Existing Data (Except Identifiers)

- (a) Write UPDATE into columns 1-6 of the 2-01 card (see paragraph 10.4).
 Do this only once for a single computer run.
- (b) For each mission to be revised, supply a card that contains the following:
 - The mission identifier in columns 1-8, exactly as it appears in columns 1-8 of the 3-01 card
 - The sequence number 202L in columns 77 80
- (c) Supply the 3-01 card. This may be the original or a card with revised data in columns 9-50.
- (d) Follow the 3-01 card with any 3-02 card correcting the data for any base experiment. Update any experiments in the order in which they appear on the tape file. Only revisions in columns 28 75 are permitted.
- (e) After the last revised 3-02 card, or after the 3-01 card if no base experiments are to be revised, place a SKIP, LAST, or STOP card for whichever message applies:
 - SKIP = revisions for traverse data follow
 - LAST = revisions for additional missions follow
 - STOP = no more mission update follows

- (f) Follow a SKIP card with the 3-03 card for any traverse in which data revisions are desired. This may be the original 3-03 card for that traverse or a card with revised data in columns 31 60. Correct the traverses in the order in which they appear on the mission tape file.
- (g) Follow the 3-03 card by any 3-04 card correcting the data for any experiment on the indicated traverse. Update any experiments in the order in which they appear on the traverse file. Only revisions in columns 28 74 are permitted.
- (h) After the last 3-04 card for the traverse, place a SKIP, LAST, or STOP card for whichever message applies.
 - SKIP = revisions for another traverse of the same mission follows
 - LAST = revisions for another mission follows
 - STOP = no more mission update follows
- 10.7.2 Procedure for Deleting Whole Missions
 - (a) Write DELETE into columns 1-6 of the 2-01 card for the computer run (see paragraph 10.4).
 - (b) Supply cards for each mission to be deleted. Fill out this card as instructed in paragraph 10.7.1 above.
 - (c) Place a STOP card after the last card from 2(b).
- 10.7.3 Procedure for Adding Whole Missions
 - (a) Write INSERT into columns 1 6 of the 2-01 card for the computer run (see paragraph 10.4).
 - (b) Assemble the cards for the additional missions in the manner used to generate a mission file (see paragraph 10.5.1).
 - (c) Make sure a LAST card is at the end of the data for each mission except the final mission. Place a STOP card at the end of the final mission.

10.7.4 Procedure for Copying a Selected Subset of Missions onto Another Tape

- (a) Write COPSEL into columns 1-6 of the 2-01 card for the computer run (see paragraph 10.4).
- (b) Prepare cards for each mission desired on the new tape. Prepare the card as instructed in paragraph 10.7.3 (b) above.)
- (c) Place these cards in the order desired for the missions on the new tape. Place a STOP card at the end of the stack.

10.8 MODE, SYSTEM, SUBSYSTEM, AND EXPERIMENT INPUT REVISION

The space permitted on magnetic tape and in the computer memory for the data file for each mode, emergency mode, system, subsystem, SOTA advance, and experiment is fixed. Each file can be thought of as a box into which data of any amount can be placed up to the limit of the box. Fixed record length permits the insertion, deletion, or correction of any item of data.

Each item of data is presented to the machine in the form of "words." These words can take on two formats: alphabetic (called "A") or numeric with a variable decimal point position (called "F"). A-format is used for identifiers and text that is read by the machine exactly as presented. F-format presents numbers in a manner that permits flexible use of the decimal point. Tables H-1 through H-5 in Appendix H identify the format relating to each input item. Different techniques are required by the Univac 1107 and 1108 to handle A-format and F-format data. This constraint of the computer forces the use of a special updating procedure.

Three types of cards are used to update the fixed length data records. These cards are summarized in Table 10-15. The first card (Card 2-02) locates the identifier of the specific data record to be changed. The second card (Card 2-03) updates A-format information. The third card (Card 2-04) allows F-format revisions. See Appendix H for instructions on preparing these cards.

Table 10-15 FIXED RECORD UPDATE INPUT CARDS

ID No.	Card Name	Maxi- mum	Sec (C	quen Colu	ce 1 mns	No.)	Card Function	
		Cards	77	78	79	80		
2-02	Computer Control	1	2	0	2	L	To locate tape data record	
2-03	A-Format Changes	Any	2	0	3	L	To make A-format update	
2-04*	F-Format Changes	Any	2	0	4	L	To make F-format update	

*The first 2-04 card of any sequence must be preceded by a SKIP card.

After a tape record file has been generated, there are five options that can be used to modify that file:

- Change an existing record (except identifiers)
- Delete an entire data record
- Add an entire data record
- Copy an entire record file onto another tape
- Copy a selected subset of records onto another tape in any order

Any changes to file records must be made for each file <u>in the order in which that file</u> <u>appears on the tapes</u> (i.e., Tape 1: modes, emergency modes; Tape 2: systems, subsystem, experiments). If one option is selected for one file on a tape, an option must be selected for all other file on that tape. If no changes to the other files are desired, then the COPY option should be used.

10.8.1 Procedure for Changing an Existing Record

To change an existing record (including the adding or deleting of data bits but not including changes in identifier number), perform the following steps.

- (a) Write UPDATE in the appropriate columns of the 2-01 card (see paragraph 10.4) for the data file to be changed. Do this only once for a single computer run.
- (b) Prepare a 2-02 card giving the identifier of each specific data record within the file that is to be revised. Directions for filling out the 2-02 card are presented in Appendix H.
- (c) Prepare a 2-03 card for each A-format change to be made in the data record. Directions for filling out the 2-03 card are presented in Appendix H.
- (d) After the last 2-03 card, or after the 2-02 card if no A-format changes are desired, place a SKIP, LAST, or STOP card for whichever message applies:
 SKIP = F-format changes for this data record follow
 - LAST = No more changes for this data record are desired; changes to another data record within the data file follow.

STOP = No more changes in this file are desired.

- (e) Follow a SKIP card with 2-04 cards for all of the F-format changes desired.
- (f) After the last 2-04 card, place a LAST or STOP card for whichever of the messages in (d) applies.
- (g) See Fig. 10-9 as an aid in stacking cards.
- 10.8.2 Procedure for Deleting Entire Data Records

ې

- (a) Write DELETE in the appropriate columns of the 2-01 card (see paragraph 10.4) for the data file in which deletions are desired.
- (b) Prepare 2-02 cards for each data record to be deleted. Directions for filling out a 2-02 card are presented in Appendix H.
- (c) Place a STOP card after the last 2-02 card.

10.8.3 Procedure for Adding Entire Data Records

- (a) Write INSERT in the appropriate columns of the 2-01 card (see paragraph 10.4) for the data file to which records are to be added.
- (b) Assemble cards for each new data record in the same manner as when generating a data tape (see 10.5.2, 10.5.3, and 10.6).







10.8.4 Procedure for Copying an Entire Record File

To copy an entire record file onto another tape, write COPY in the appropriate columns of the 2-01 card (see paragraph 10.4) for the data category. No other cards are needed.

10.8.5 Procedure for Copying a Selected Subset of Records onto Another Tape in any Order

- (a) Write COPSEL in the appropriate columns of the 2-01 card (see paragraph 10.4) for the data file for which the data records are to be transferred.
- (b) Prepare 2-02 cards for each data record desired on the new tape. Directions for filling out a 2-02 card are presented in Appendix H.
- (c) Arrange the 2-02 cards into the order desired for the data records on the new tape.
- (d) Place a STOP card after the last 2-02 card.

10.9 MISSION-MODE COMPARISON CONTROL CARDS

Operation of the Mission/Mode Comparison Routine requires a set of control cards. Table 10-16 summarizes the types of cards required. Only one 2-10 card is required for each computer run. At least one each of the 2-11 and 2-12 cards are needed for every comparison of mission and mode. Figure 10-10 shows how to stack the cards for a run. Instructions for preparing the 2-11 and 2-12 cards are given in Appendix I.

The number punched in column 1 of the 2-10 card determines the output format for the run. If a zero is punched, the computer prints the standard output of 6-8 pages for each comparison (see Fig. 7-1). If a number 1 is punched in column 1, a detailed diagnostic printout of the calculation process will be generated. The detailed printout is only valuable in troubleshooting the program logic.

10-40





LMSC-A847943 Vol. I

MISSION-MODE COMPARISON CONTROL CARDS

Card ID	cd Card Name Maxi- Sequence . Card Name mum (Colum Cards 77 78 7		Maxi- Sequence No. mum (Columns)				Comments		
No.			79	80					
2-11	Printout Option	1	-	-	-	-	0 in column 1 yields standard printout; 1 in column 1 yields diagnostic printout.		
2-11	Mission/Mode Identification	1	2	1	1	L	Any number or comparisons can be run at one time.		
2-12	Traverse-Vehicle Identification	3	2	1	2	*	At least one 2-12 card must follow each 2-11 card.		

*1, 2, or L, depending on maximum number of cards needed. (The letter L is always used on the last card of a series.)

The 2-11 card specifies which mission and mode are to be matched. Provision also is made for multiplying the shirtsleeve and spacesuit manhours stored with the experiment data (see paragraph 10.6.3) by special factors. These factors alter the manhour numbers to account for effects of reduced gravity and spacesuit mobility problems.

The 2-12 card identifies the roving and flying system configurations used during the mission and establishes the traverses performed by each. For traverses performed by a rover alone, the rover identifier is input on the control card. For traverses performed by a rover-trailer combination, the trailer identifier (i.e., the special 6-digit Data Book number that identifies a trailer concept used with a particular rover) is input on the control card. In addition, a space is provided on the control card to input the identifier of the particular performance data from the trailer input (which is the performance of the rover-trailer combination) and to ignore the rover performance data (see paragraph 10.6.1).

10.10 INTEGRATION AND COST CONTROL CARDS

Operation of the Integration and Cost Routine requires three types of control cards. Table 10-17 summarizes these cards. The 2-20 card inputs the title for the lunar
program printout, establishes the maximum number of Saturn launches, N, permitted by existing facilities during each calendar year, and allows exercising of various cost options. Two 2-21 cards input a table of maximum permitted launches during each year. The 2-22 card establishes exactly which mission/mode combinations will constitute the lunar program. The 2-23 card allows input of the cost curves for a change of facilities that will allow an increase in annual launch rate from N to some value M, from M to P, and so on. Up to 10 possible launch-rate increments may be input.

Table 10-17

Card ID No	Card Name	Maxi- mum	Se (egme Colu	ent l imn	No. s)	Comments
NO.		Carus	77	78	79	80	
2-20	Integration and Cost Control – General	1	2	2	0	L	One card per run
2-21	Launch Rate Limit vs. Calendar Year	2	$2 \\ 2$	$2 \\ 2$	1 1	1 L	Two cards re- quired each run
2-22	Mission/Mode Combinations	17	2	2	2	*	50 mission/mode combinations may be us e d per lunar program
2-23	Launch Rate Improvement Cost Curve	20	2	**	**	*	Two sets of curves required per step in launch rate improvement

INTEGRATION AND COST CONTROL CARDS

*1, 2,..., L depending on maximum number of cards desired. (The letter "L" is always used on the last card of a series.)

**Columns 78 and 79 of Card 2-23 take the form "XX"

where XX = 23 for nonrecurring R&D of first step in launch rate

= 24 for nonrecurring Cost of Facilities of first step in launch rate

- = 25 for nonrecurring R&D of second step in launch rate
- = 26 for nonrecurring Cost of Facilities of second step in launch rate

= 41 for nonrecurring R&D of tenth step in launch rate

= 42 for nonrecurring Cost of Facilities of tenth step in launch rate

10-43

The cards are input in the order shown in Fig. 10-11. Care must be taken in assembling the 2-22 cards so that missions that are dependent on each other are called out in proper order. A STOP card at the end of the stack signals completion of the card input. Instructions for filling out the cards are given in Appendix J.

10.11 COMPUTER INPUT CHECKLISTS

Despite the care taken in preparing input cards, errors still occur. The use of checklists may help to avoid these errors.

10.11.1 Mission Input Checklist

The following checklist should be used before submitting cards for generation of each mission tape:

- (1) List each mission ID number exactly as it appears in columns 1-8 of the 301L card.
- (2) Check that the mission start date on each 301L card begins in column 31.
- (3) Check that the "related mission" number in columns 39 40 of the 301L is <u>exactly</u> as it appears in columns 7 8 of the related mission's 301L card.
- (4) Check that a SKIP card precedes each 303L card used to call out a traverse for the mission.
- (5) Check that the traverse ID for each 303L card begins in column 13.
- (6) Check that a LAST card appears after the end of the data cards for each mission, except that after the final mission a STOP card appears.

10.11.2 Mode Input Checklist

The following checklist should be used before submitting cards for generation of each new mode tape:

- (1) Check that the mode ID appears in columns 1 12 of each 401L card.
- (2) Check that the number given in column 75 of the 401L card equals the number of 407X cards.

10-44



10-45

- (3) Check that <u>all</u> systems required for the mode as given in the 41XX cards are included in the Latest System/Experiment Tape. Any missing systems must be added to that tape.
- (4) Check that the number in column 15 of each 41XX cards also appears in column 79 of that same card.
- (5) Check that the sequence numbers in columns 77 80 of each card are in proper order.
- (6) Check that a LAST card appears after the end of the data cards for each mode, except that a STOP card appears after the final mode.

10.11.3 Mission-Mode Match Checklist

The following checklist should be used before submitting control cards for each missionmode match computer run:

- (1) Check that the mission ID number appears in columns 13 20 of each 211L card.
- (2) Check that the data for the mission appear on the Latest Mission/Mode tape.
- (3) Check that the mode ID number appears in columns 25 36 of the appropriate 211L card.
- (4) Check that the mode appears on the Latest Mission/Mode Tape.
- (5) Check that the shirtsleeve and spacesuit manhour multipliers appear in columns 50 55 and 60 65, respectively, on the 211L card.
- (6) Check that the mode ID number on each 212X card <u>exactly</u> matches the mode ID number on the 211L card.
- (7) Check that the traverse ID numbers given on the 212X card start in columns 31, 37, 43, 49, 55, and 61.
- (8) Check that each traverse called out on the 212X cards appears on the mission tape (and vice versa).

10-46

10.11.4 Exploration Program Cost Checklist

The following checklist should be used before submitting control cards for each exploration program cost run on the computer.

- (1) Check that only mission ID numbers appear in columns 1-8, 26-33, and 51-58 of the 222X cards.
- (2) Check that only the accompanying mode ID numbers appear in columns 9-20, 34-45, and 59-70 of the 222X cards.
- (3) Check that the data for each mission appears on the Latest Mission/Mode Tape.
- (4) Check that each mode appears on the Latest Mission/Mode Tape.
- (5) Check that the sequence numbers on the 222X cards are in the proper order.

LOCKHEED MISSILES & SPACE COMPANY

,

Appendix A

EXAMPLE LUNAR PROGRAM SUMMARY

(Ref. Chapter 9)

00 manhour scientific program using	-Apollo missions use Saturn Apollo A
55,0	post-
nes a	nine
accomplish	The first
t program	ogram A).
ratior	fic Pr
explo	cienti
lunar	sh" (S
example	th approac
Γhε	'pa

GENERAL DESCRIPTION

the so-called

pplications (AAP) hardware and extend through 1974. The second evolutionary hardware step utilizes an uprated Saturn (125% Saturn V) and a direct logistics landing vehicle (LLV). The earliest date allowed for introduction of this new hardware is A further objective is to introduce the 175% Saturn V at the earliest practicable date "path approach" (S 1976.

Eight locales and twenty different paths are to be explored. No rigid cost constraints were specified but the funding target was set at 1 to 2 billion dollars per year. Launch rates are three or four per year through 1975, four Transportation system costs for the first two postper year through 1977, and six to eight per year thereafter. Apollo missions are included in the Apollo run-out costs.

25.6 billion dollars spread over about 20 years. The program indicates a cost of about 2 billion dollars for develmission to Grimaldi, beginning in late 1985 and requiring 10 launches. The total number of Saturn launches is 77, 125% Saturn and the LLV in 1976 and the 175% Saturn in 1981. The program is completed with a large astronomy including 6 Apollo missions and 4 test launches (2 each for the 2 Saturn upratings). The total post-Apollo cost is As developed with these objectives and within these constraints, the example exploration program introduces the opment of the 125% Saturn and the LLV. Funds must be committed in 1969. A further commitment of the order of 3 billion dollars for development of the 175% Saturn must be made in about 1976.

A-2

total mass of scientific equipment is 153,000 kg; the masses of the geoscientific and astronomical equipment are The During its early and intermediate phases, Program A is essentially a plan for geoscientific exploration of the moon, with emphasis toward the end on establishment of major facilities for radio and optical astronomy. about equal

length of 8,500 km. In addition, there are missions to eight widely separated locales, each covering an explora-Program A uses the path approach for surface exploration of the moon, which involves 20 paths with a total path tion area of about 100 sq km.

refraction techniques. Experiments in active seismology require detonations or surface impacts with energies The subsurface structure of the moon is probed to depths of 200 km by means of active seismic reflection and ranging from the equivalent of 0.5 to 6,000 kg of TNT. The total amount of explosives used in Program A is equivalent to 156,000 kg of TNT. Core-drill samples from a depth of 300 m are acquired at five locales

optical telescopes are 1 and 2.5 m, and they are designed primarily for operation at wavelengths in the far ultrathat are troublesome to earth-based radio telescopes because of atmospheric interference. The apertures of the The astronomical facility comprises two types of radio telescopes, three optical telescopes, and one x-ray telescope. The radio telescopes operate at relatively low frequencies (0.3 to 1.0 MHz) that are not observable with violet region, which is excluded from observation by earth-based telescopes because of atmospheric absorption. earth-based radio telescopes and at relatively short wavelengths (in the millimeter and submillimeter ranges) The x-ray telescope uses grazing-incidence double-reflection optics with an aperture diameter of 1 m and an angular resolution of 1 arcsec.

The total number of distinct scientific experiments, considering all scientific disciplines, is 198, and a total of 54,000 manhours is allocated for the performance of those experiments.



Launch Total^(a) Non-Recurring Post Apollo Costs Apollo Costs TOTALS 15,237 3, 498 25, 588 6, 566 lStep 3 | -Missione 1985 1986 1987 -æ ~ æ Step 2 Missions -1 ~ £ F ~ н Step 1 Missions ŝ ø e -Evolution ~ ŝ Equipment ч Þ ~ -ł er. 791. Apollo Missions .01 ŝ ~ n Run-Out Costs S7, 478, 600, 000 en ł -Π -Apollo n i Þ ŝ ŝ ø ÷, TOTAL (NILLIONS) PROCUREMENT OPERATIONS N - Number of Missions Started Calendar Years ▼- Earliest Reqⁱd R&D Start LAUNCH RATE T – Required Test Lunches KEY MILESTONES VARANT COST (BILLIONS OF DOLLARS) C OF F Fiscal Years RED Non-Recur ระเนาของม POST A POLLO COSTS

COST AND SCHEDULE SUMMARY

Includes six Apollo launches.

e

POST-APOLLO MISSION SUMMARY

Ĺ		Mindae	Mada	Ministra -			1 - 1 - 2	0-11161-				1
		WATGOTIN	Identification	Jo	Map	Time	Man-	Man-	Total Cargo	Equipment	Reserve	Traverse
ģ	Start	Location (paths - locales)	Number	Launches		(days)	hours	hours	(94)	(kg)	(kg)	(km)
2	70A	Orbiter 1	303-10021-02	-	ı	27	1908	636	4183	1737	(11,230)	٢
80	70B	Ranger VIII Landing Site	402-10012-01	8	17	15	729	180	4716	575	381	150
6	71A	Moltke B	402-10011-01	~	26	16	733	180	4719	575	378	150
10	71B	Orbiter 2	303-10022-02	-	ı	27	1908	636	5031	1777	(10,382)	1
1	72A	Tycho	402-10011-03	2	11	16	780	197	4173	482	920	150
12	72B	First Visit to Grimaldi, Astronomical Location	402-10011-03	5	21	16	747	190	4221	581	855	06
13	73B	Orbiter 3	303-10022-02	-	ł	27	1908	636	5031	1777	(10,382)	ſ
14	73B	South Pole	402-10011-03	01	16	18	837	213	4165	458	973	150
15	74B	Second Visit to Grimaldi, Astronomical Location	402-10081-01		21	18	864	115	63.75	2407	3674	r
16	76A	Capella M	503-20162-01	2	8	10	678	204	10,502	623	6798	300
17	76B	Copernicus - Dark Halo Craters Southeast of Copernicus	503-20162-01	2	n - 5	27	1937	ζυ\$	11,135	1235	5165	300
18	77A	Southcast Screnitatis	503-20162-01	5	น	10	667	199	20,964	602	5818	300
19	77B	Mare Crisium	503-20162-01	2	æ	6	647	189	10,323	443	5977	300
20	78A	M Imbr, M Imbr to M Seren, P Putr to M Vapor	503-20191-01	5	a,b,c	105	7582	2372	62, 608	42,024	2592	3900
21	79.A	Mare Humorum	503-20162-01	8	0	15	1035	310	10,517	637	5783	600
22	V61	North Pole	503-20162-01	2	۲	6	647	189	10,323	443	5977	300
23	79B	Mare Smythii	503-20162-01	7	۵.	27	1948	544	10,493	593	5807	1200
24	80B	Central Farside	503-20201-01	4	Ţ	48	3470	1216	48,900	29,548	0	1200
25	81B	Marius Hills to Aristarchus	503-30094-01	ñ		41	2949	906	37,499	19,331	13, 301	1200
26	82.A	Ptol to Alphon, C High, M Nub to St Wall - Alphon, St Wall	503-30083-01	2	g,h-2,18	65	4636	2192	25,400	. 8539	0	2300
27	82B	Mare Orientale	503-30103-01	3	æ	36	2600	1213	48,762	29,536	2278	1200
28	83B	P Putr to M Vapor, M Vapor to M Tranq, Si Medii	503-30114-01	4	c,d,e	73	5226	1695	61,949	42,417	14,251	2400
29	84A	Southern Highlands	10-00100-005			36	2597	1211	48,748	29,522	2292	1200
30	84B	Northern Farside	503 -30094-01	e	*	34	2429	748	37,209	19,094	14.734	1200
31	85B	Third Visit to Grimaldi. Astronomical Location	406-30161-01	9	21	822	118.43:	37,466	118,051	75,720	61-68	60
32	86B	Fourth Visit to Grimaldi, Astronomical Location	406-30161-02	4	21	0	0	o	0	0	0	•
		Total				1547	167,94-	54,538	605,515	310,676	106,905	18, 650

EQUIPMENT USAGE AND COST SUMMARY

.

	Equipment Identification	4	First/	Number	Procured		Cost Sum	mary (Millions	of Dollars)	
		Start	Last			Nonrec	urring	Recuri	gui	Total
Ð ‰.	Name		U 66 •	Operations	Bpures	R&D	C of F	Procurement	Operation	Cust
	Launch Vehicles									
10-1771	I WW Salurn V	1	IA/74B	13	61	1	1	2100	270	2370
1221-03	100% Saturn V (Apollo funded)	1	8A/70B	6,	1	ł	ł	1	I	1
1231-01	125% Saturn V	72A 7	6A /79B	23	s	570	42	2760	484	3856
1251-01	1757 Saturn V	477A	80A/86A	28	e	818	135	3799	548	5300
	Subtotals					1388	177	8659	1302	11,526
	Flight Systems									
1311-01/1321-01	CSM - LOR - Three-Man	1	1A/74B	13	5	4	1	1231	288	1519
1311-02/1321-04	CSM - LOR - Three-Man	73B	6A/79B	თ	1	76	1	821	190	1087
1331-01/1341-01	LM Taxi - Two-Man	67B	0B/74B	9	1	32	1	348	85	465
1332-02/1342-01	LM Taxi - Three-Man	72B	16A/79B	Ø	1	246	ł	561	125	932
1412-02/1454-01	CSM - Direct - Six-Man	77B 8	31A/86A	11	61	006	12	1070	172	2154
1421-02/1431-03	Personnel Braking and Landing Stages	76B	31A/86A	11	2	459	24	221	31	735
1423-02/1433-02	LLV – 125% Saturn V	71B	16A/79B	14	2	599	24	328	55	1006
1423-04/1433-04	LLV – 175% Saturn V	76B 6	80A/85A	17	2	687	24	405	69	1185
	Subtotal	ſ				2999	84	4985	1015	9083
	Mission Equipment									
2121-01	Orbit-Launched Probe	68B	0A/73B	7	1	58	1	26	~	86
2132-02	Surface - Launched Probe	74B	8A/84B	20	2	38	ſ	46	4	88
2222-02	Orbiter Rack	67B	0A/73B	e	I	1	I	1	I	-
2321-01/1351-03	LM Shelter - Two-Man	67B	0B/74B	6/7(a)	1	217	I	318	92	627
2325-05	Shelter – Six-Man	80A 8	15A/-	1	1	484	ı	88	н	583
2421-01	ILSSM	67B 7	'0B/73B	7	1	58	21	39	4	103
2423-03	LRV - Three-Man - 90 Days	69B 7	6A/85A	16	8	421	2	347	27	797
2433-04	Trailer – Large Multipurpose	75A 7	8A/85A	ð	I	52	. 6	22	2	141
2512-03	LFV – Two-Man Exploration	67B 7	0B/71A	8	1	48	1	æ	1	57
2521-02	LFV – Return to Orbit	74A	8A/84B	S	1	62	ı	20	5	84
2721-02	Nuclear Power Supply - 20 kw	78B 8	SA/-	1	1	136	20	28	'	184
	Subtotal					1575	27	1004	145	2751
	Major Scientific Equipment									
3213-02	300-m Drill With Fuel Cell	74B 7	6A/84B	S	1	11	ı	n	ı	22
3223-01	15-m Parabolic Radio Telescope	81 A	15A/-	1	1	4	1	4	I	80
3224-02	Radio Telescope – Mills Cross	79B	5A/-	-	1	30	ı	4	1	35
3231-01	X-Ray Telescope	78A B	5A/-	I	1	39	ł	9	-	46
3242-01	1-m Optical Telescope	7 A07.	4B/-	1		29	1	n	-	41
3242-04	2-m Optical Telescope	79A 8	5A/-	I	1	152	ł	25	1	178
3243-01	2.5-m Optical Telescope	78A B	5A/-	-	1	227	ŀ	35	1	263
	Subtotal					492	1	96	5	593
	Total					6454	288	14.744	2467	23,953
(a) One 1351-03 (ised to deliver 1-m optical telescope.									



÷ .:

Appendix B

DETAILED INSTRUCTIONS FOR PREPARING MISSION INPUT CARDS

(Ref. Paragraph 10.5.1)

LOCKHEED MISSILES & SPACE COMPANY

CARD I.D. 3-01

PAGE 1 OF 2

MISSION INPUT - GENERAL

THIS CARD IS USED TO IDENTIFY EACH MISSION AND TO RELATE THE MISSION TO OTHERS IN THE LUNAR PROGRAM.

1-28 MISSION IDENTIFIER. INPUT AN IDENTIFIER THAT UNIQUELY LABELS THIS MISSION WITHIN THE FIRST EIGHT COLUMNS. THE FOLLOWING INPUT FORMAT WILL BEST MATCH THE COMPUTER INTERNAL LOGIC AND OUTPUT PRINT ROUTINE.

> 1-3 SCIENTIFIC PROGRAM IDENTIFIER. USE ANY COMBINATION OF LETTERS AND NUMBERS TO LABEL THE SCIENTIFIC PRO-GRAM UNDER CONSIDERATION. START THE INPUT IN COL-UMN 1.

5-6 MISSION MIX IDENTIFIER. IDENTIFY THE GROUP OF MIS-SIONS BEING DERIVED TO ACCOMPLISH THE SCIENTIFIC PROGRAM GIVEN ABOVE.

7-8 NUMBER OF THIS MISSION IN THE SEQUENCE OF MISSIONS.

9-26 LOCATION OF THIS MISSION ON OR ABOUT THE MOON. SPEC-IFY THE ACTUAL LOCATION.

27-23 NUMBER OF TIMES THIS LOCATION WILL HAVE BEEN VISITED COUNTING THIS MISSION.

31-35 NOMINAL MISSION START DATE. INPUT THE CALENDAR YEAR TO BE USED AS THE NOMINAL STAPT DATE FOR THE MISSION. ALSO INDICATE WITH AN A OR B WHETHER THE START DATE IS IN THE FIRST OR SECOND HALF OF THE YEAR. FOR EXAMPLE, 1973A = FIRST HALF OF 1973 1973B = SECOND HALF OF 1973 THIS DATE IS USED BY THE COMPUTER AS THE STAPTING POINT

IN ITS ITERATIONS TO FIND THE MOST DESIRABLE START DATE.

39-40 RELATED MISSION NUMBER. INPUT HERE THE MISSION SEQUENCE NUMBER (SEE 7-9 ABOVE) OF THE MISSION UPON WHICH THE START OF THIS MISSION DEPENDS. FOR EXAMPLE, IF THIS IS MISSION 7 AND IT CANNOT START UNTIL MISSION 5 IS COMPLETED, THE NUMBER 5 IS INPUT HERE. IF THERE IS NO RELATED MISSION, LEAVE THESE SPACES BLANK.

B-2

NOTE- BE SURE TO PUT THE NUMBER OF THE RELATED MISSION IN COLUMNS 39-40 JUST AS IT APPEARS IN COLUMNS 7-9 OF THE IDENTIFIER FOR THAT MISSION.

CARD I.D. 3-01 PAGE 2 OF 2 MISSION INPUT - GENERAL (CONTINUED) MINIMUM NUMBER OF 6-MONTH PERIODS THAT MUST ELAPSE BETWEEN 41-50 THE LAST LAUNCH OF THE RELATED MISSION GIVEN IN 39-40 AND THE FIRST LAUNCH OF THE MISSION BEING IDENTIFIED WITH THIS CARD, INPUT O (ZERO) FOR 0-6 MONTHS, 1 FOR 7-12 MONTHS, THE INPUT NUMBER MAY BE PLACED ANYWHERE IN THE SPACE ETC. PROVIDED. BE SURE TO SHOW THE LOCATION OF THE DECIMAL POINT. IF COLUMNS 39-40 ARE BLANK, LEAVE THIS INPUT BLANK. SEQUENCE NUMBER. INPUT THE FOLLOWING NUMBER-77-80 301L .

CARD I.D. 3-02

PAGE 1 OF 4

MISSION INPUT - BASE EXPERIMENT

THIS CARD IS USED TO IDENTIFY THOSE EXPERIMENTS FOR A MISSION WHICH ARE NOT DONE ON A TRAVERSE. THIS CARD SHOULD ALSO BE FILLED OUT FOR PROBE AND ORBITER MISSIONS. USE AS MANY CARDS AS THERE ARE NON-TRAV-ERSE EXPERIMENTS. UP TO 100 CARDS MAXIMUM MAY BE USED ON ANY ONE MISSION. IF ALL EXPERIMENTS ARE DONE ON TRAVERSE, DO NOT INPUT THIS CARD.

16-27	EXPERIMENT IDENTIFIER. INPUT HERE THE NUMBER WHICH UNIQUELY
	IDENTIFIES AN EXPERIMENT. IF POSSIBLE, USE THE NUMBER
	PROVIDED BY THE NORTH AMERICAN AVIATION STUDY OF LUNAR
	MISSIONS. EXPAND THEIR 8-DIGIT NUMBER INTO FIVE SECTIONS
	WITH DECIMAL PDINTS AS FOLLOWS.
	Y Y YY YY

BEGIN THIS 12-PLACE NUMBER IN COLUMN 16.

28 PHASE 1 CODE. INPUT HERE A CODE LETTER TO DESIGNATE THE UNITS OF THE NUMBER THAT FOLLOWS FOR THE FREQUENCY OF PHASE 1 (SETUP) OF THE GIVEN EXPERIMENT. USE ONE OF THESE LETTERS-

> A = REPLICATIONS PER MISSION D = REPLICATIONS PER MONTH

- 29-32 FREQUENCY OF PHASE 1. INPUT HERE THE NUMBER OF TIMES THE GIVEN EXPERIMENT MUST BE SET UP. BE SURE TO USE A NUMBER CONSISTENT WITH THE CODE IN COLUMN 28. SEE NOTE # ON PAGE 4 OF 4.
- 33 PHASE 2 CODE. INPUT HERE A CODE LETTER TO DESIGNATE THE UNITS OF THE NUMBER THAT FOLLOWS FOR THE FREQUENCY OF PHASE 2 (OPERATION) OF THE GIVEN EXPERIMENT. USE ONE OF THE LETTERS IDENTIFIED IN 28 ABOVE.
- 34-37 FREQUENCY OF PHASE 2. INPUT HERE THE NUMBER OF TIMES THE GIVEN FXPERIMENT IS PERFORMED. BE SURE TO USE A NUMBER CONSISTENT WITH THE CODE IN COLUMN 33. SEE NOTE * ON PAGE 4 DE 4.
- 39 PHASE 3 CODE. INPUT HERE A CODE LETTER TO DESIGNATE THE UNITS OF THE NUMBER THAT FOLLOWS FOR THE FREQUENCY OF PHASE 3 (ANALYSIS) OF THE GIVEN EXPERIMENT. USE ONE OF THE LETTERS IDENTIFIED IN 29 ABOVE.
- 39-42 FREQUENCY OF PHASE 3. INPUT HERE THE NUMBER OF TIMES THAT DATA OR SAMPLES TAKEN DURING THIS EXPERIMENT ARE ANALYZED. BE SURE TO USE A NUMBER CONSISTENT WITH THE CODE IN COL-UMN 38. SEE NOTE # ON PAGE 4 OF 4.

			+
		CA	RD I.D. 3-02 PAGE 2 OF 4
•	<u>M</u>	ISSION INPUT - B	ASE EXPERIMENTS (CONTINUED)
43	PH, 1	ASE 4 CODE. INP INITS OF THE NUM PHASE 4 LTAKE DO	UT HERE A CODE LETTER TO DESIGNATE THE BER THAT FOLLOWS FOR THE FREQUENCY OF WN) OF THE GIVEN EXPERIMENT. USE ONE OF
	-	THE LETTERS IDEN	TIFIED IN 28 ABOVE.
44-47	FR	EQUENCY OF PHASE THE EXPERIMENT E ISE A NUMBER CON	4. INPUT HERE THE NUMBER OF TIMES THAT QUIPMENT MUST BE TAKEN DOWN. BE SURE TO SISTENT WITH THE CODE IN COLUMN 43. SEE
	!	NOTE * ON PAGE 4	
48	<u>ן אין אין אין אין אין אין אין אין אין אי</u>	<u>INCH PAYLOAD MAS</u> DESIGNATE THE IN MASS LANDED ON T	S CODE USE ONE OF THE FOLLOWING CODES TO PUT DESIRED FOR CALCULATION OF EQUIPMENT HE MOON FOR THIS EXPERIMENT-
		INPUT	MEANING
		LEAVE BLANK	THIS EXPERIMENT USES MINDR SCIENTIFIC
			EQUIPMENT. USE THE NUMBER GIVEN WITH THE EXPERIMENT
· · · · · · · · · · · · · · · · · · ·		Δ	THIS EXPERIMENT USES MAJOR SCIENTIFIC EQUIPMENT. USE THE NUMBER IN THE FOLLOW- ING 6 COLUMNS INSTEAD OF THAT GIVEN WITH THE EXPERIMENT.
		B	THIS EXPERIMENT USES MINOR SCIENTIFIC
			EQUIPMENT. HOWEVEP, USE THE NUMBER IN THE FOLLOWING 6 COLUMNS INSTEAD OF THAT GIVEN WITH THE EXPERIMENT
49-54	1	AUNCH PAYLOAD M INPUT IN COL	ASS OVERRIDE. INPUT AS FOLLOWS DEPENDING ON UMN 48 (SEE NOTE* ON PAGE 4 OF 4) -
		INPUT IN 48	INPUT HERE
	I	NONE	NONE
		Д	MASS OVERRIDE, IN KG, IN PLACE OF MASS GIVEN WITH THE EXPERIMENT
		, R	MASS OVERRIDE, IN KG, IN PLACE OF MASS GIVEN WITH THE EXPERIMENT
		and the second second second second second second second second second second second second second second second	

B-5

CARD 1.D. 3-02

PAGE 3 OF 4

MISSION INPUT - BASE EXPERIMENTS (CONTINUED)

55 NONRECUPRING COST CODE. USE ONE OF THE CODES IN 48 ABOVE TO INDICATE THE INPUT THAT IS DESIRED FOR THE NONRECURRING COST OF THE SCIENTIFIC EQUIPMENT ASSOCIATED WITH THIS EX-PERIMENT.

56-61 NONRECURRING COST OVERRIDE. INPUT AS FOLLOWS DEPENDING ON THE INPUT IN COLUMN 55 (SEE NOTE * PAGE 4 DF 4) -

	INPUT IN 55	INPUT HERE
	NONE	ΝΟΝΕ
· · · · · · · · · · · · · · · · · · ·	٨	NONRECURRING COST, IN MILLIONS, FOR
		MINOR SCIENTIFIC EQUIPMENT ONLY. THIS COST IS TO BE USED IN PLACE OF THE COST GIVEN WITH THE EXPERIMENT.
	3	NONPECURRING COST, IN MILLIONS, TO BE USED IN PLACE OF THE COST GIVEN
		WITH THE EXPERIMENT.

- 62 RECURRING COST CODE. USE ONE OF THE LETTER CODES GIVEN IN 48 ABOVE TO INDICATE THE INPUT THAT IS DESIRED FOR RECUR-PING HARDWARE PROCUREMENT OF THE SCIENTIFIC FOULPMENT AS-SOCIATED WITH THIS EXPERIMENT.
- 63-69 RECHRRING COST OVERRIDE. INPUT AS FOLLOWS DEPENDING ON THE INPUT IN COLUMN 62 (SEE NOTE * PAGE 4 OF 4) -

	INPUT IN 62	INPUT HERE	1
· · ·	NONE	NONE	
	Δ	PECHRPING COST, IN MILLIONS, FOR MINOR SCIENTIFIC EQUIPMENT ONLY. THIS COST IS TO BE USED IN PLACE OF THE COST GIVEN WITH THE EXPERIMENT.	4
	<u> </u>	RECURBING COST, IN MILLIONS, TO BE USED IN PLACE OF THE COST GIVEN WITH THE EXPERIMENT.	4

|--|

PAGE 4 OF 4

69-75 MINIMUM TIME FOR ALL MANNED REPLICATIONS. INPUT THE TIME, IN DAYS, THAT IS REQUIRED FOR THE CREW TO COMPLETE THE GIVEN EXPERIMENT. COUNT THE ENTIRE SPAN FROM START TO FINISH EVEN THOUGH LITTLE OR NO WORK MAY BE DONE ON SOME OF THE DAYS IN THE SPAN. IF THE EXPERIMENT IS PERFORMED WITHOUT CREW HELP, LEAVE THIS SPACE BLANK. SEE NOTE * BELOW.

NOTE * - PLACE THE INPUT NUMBER ANYWHERE IN THE SPACE PROVIDED. BE SURE TO SHOW THE LOCATION OF THE DECIMAL POINT.

IMPORTANT - RECHECK THE INPUT OF THIS CARD. IF THE CODE LETTER A IS USED IN COLUMNS 48, 55, OR 62, IT MUST APPEAR IN ALL THREE COLUMNS.

CARD I.D. 3-03
MISSION INDUT - TRAVERSE DATA
MISSIUN INPUL - TRAVERSE UATA
THE CARD IS HERE TO INDUT CERTAIN DATA ACCOCLATED HITH FACH TO MEDIC
TO BE ACCOMPLISHED DURING THE MISSION. FILL OUT ONE CARD FOR EACH
TRAVERSE.
13-16 TRAVERSE IDENTIFIER, INPUT & SEQUENCE IDENTIFIER FOR THE
TRAVERSE. THE FORM
R-XX OR F-XX
R = ROVING TRAVERSE
F = FLYING TRAVERSE
XX = NUMBERS IN SEQUENCE FOR EACH TYPE OF TRAVERSE
BE SURE TO START THE INPUT IN COLUMN 13.
31-40 TRAVERSE RANGE. INPUT THE TOTAL DISTANCE TRAVELED, IN KILD-
METERS, FROM START TO FINISH OF THE TRAVERSE. SEE NOTE *
BELOW
41-50 NUMBER OF STOPS. INPUT THE NUMBER OF MAJOR STOPS (1 HOUR OP
LONGER FOR AN LRV OR ANY LEV LANDING) THAT OCCUR IN THE
THE TRAVERSE. SEE NOTE * BELOW.
51-60 LEV VELOCITY INCREMENT. IF THIS IS A FLYING TRAVERSE, INPUT
LEV IN ACCOMPLISHING THE TRAVERSE. COUNT ALL VELOCITY
INCREMENTS REQUIRED INCLUDING TAKEOFF, LANDING, FLIGHT,
GRAVITY LOSSES, AND RESERVE (IF ANY). SEE NOTE * BELOW.
IF THIS CARD IS FOR A POVING TRAVEPSE, LEAVE THIS SPACE
BLANK.
77-80 SEQUENCE NUMBER. INPUT THE FOLLOWING NUMBER- 303L
NOTE # - PLACE THE NUMBER ANYWHERE IN THE SPACE PROVIDED. BE SURE
TO SHOW THE LOCATION OF THE DECIMAL POINT.
B−8

							•
		CAR	D I.D.	3-04		PAGE 1 DF 4	
	MIS	SION INPUT	- TRAV	ERSE EX	PERIMENTS	·	
		·		· · ·	•	•	
THIS CA	RD IS USE	D TO IDENTI	Y THO	SE EXPE	RIMENTS PERFO	RMED ON EACH	
TRAVERS	SE IDENTIE	IED BY A CAN	$\frac{1}{1}$	• NU• 3	-03. USE AS	MANY CARDS AS	
THERE A	ARE EXPERIS	MENIS FUR E	AUH IR	AVERSE	THAT IS IVENT	IFIED. UP TO	
DU CARL	15 MAT 35			IKAVEK)C •		
			- •				·
16-27		NI IDENTIFI ETES AN EYD	ERTMENI	NPULHE T. TE	RE THE NUMBER	THE NUMBER	Y
		ED BY THE N	IRTH A	MERICAN	HTZ MOTTATVA		
	MISSIO	NS. EXPAND	THETR	8-DIGI	T NUMBER INTO	FIVE SECTIONS	
	WITH DI	ECIMAL POIN	TS AS	FOLLOWS	•	SAVE SCOTIONS	
			Χ.Χ.	XX.XX.X	X		·····
	BEGIN	THIS 12-PLA	CE NUM	BER IN	COLUMN 16.		
				A CODE			
28	PHASE I (LUUE. INPU	1 71EKE	4 CUUE T EOLLO	LEITEK IU DE Me Endo The End	SIGNATE THE	
		JE THE NUMBI	: TUC /	1 FULLU Čiven e	WS FUR THE FR	EQUENCT UP . Se ome of tues	c
	I ETTER		1112	GIVEN E	AFEALMENT. U	SE UNE UF IMES	C
	LETTERS			IONS DE	RTRAVEDCE		
		A = RC		TONS PE	R KTLOMETER		
	<u> </u>	$C = RE^{1}$	PLICAT	IONS PE	8 STOP		
		v -					
29-32	FREQUENCY	Y OF PHASE	I. IN	PUT HER	E THE NUMBER	OF TIMES THE	
	GIVEN	EXPERIMENT	UST B	E SET U	P. BE SURE T	O USE A NUMBER	
	CONSIS	TENT WITH TH	IE COD	E-IN CO	LUMN 29. SEE	NOTE * ON	
	PAGE 4	ŊF 4.					
		,					
33	PHASE 2 (CODE. INPU	T HERE	A CODE	LETTER TO DE	SIGNATE THE	
	UNITS (DE THE NUMB	R THA	T FOLLO	WS FOR THE FR	EQUENCY OF	
	PHASE 2	Z (UPERATIUS		THE GIV	EN EXPERIMENT	USE UNE OF	
	THE LE	TIERS LDENT	LEIED	IN 28 A	BUVE•		
34-37	FREQUENCY	Y DE PHASE	2. IN	PUT HER	F THE NUMBER	OF TIMES THE	
5.5	GIVEN	FXPERIMENT	IS PERI	FORMED.	BE SURE TO	USE A NUMBER	
	CONSIST	TENT WITH TH		E IN CO	LUMN 33. SFE	NOTE * ON	
*****	PAGE 4	JF 4.					
38	PHASE 3 (CODE INPUT	HFRE	4 CODE	LETTER TO DE	SIGNATE THE	
	UNITS	DE THE NUMBS	ER THA	T FOLLO	WS FOR THE FR	EQUENCY OF	
	PHASE 3	3 (ANALYSIS)) <u>OF</u> TI	HE GIVE	N EXPERIMENT.	USE ONE OF	
	THE LET	TTEPS IDENT		IN 28 A	BOVE.		
			<u></u>				
					•		
•	•	•				-	
· .	· · · · · · · · · · · · · · · · · · ·				· · ·	· · · · · · · · · · · · · · · · · · ·	

B-9

CARD I.D. 3-04

PAGE 2 OF 4

MISSION INPUT - TRAVERSE EXPERIMENTS (CONTINUED)

- <u>39-42</u> FREQUENCY OF PHASE 3. INPUT HERE THE NUMBER OF TIMES THAT DATA OR SAMPLES TAKEN DURING THIS EXPERIMENT ARE ANALYZED. BE SURE TO USE A NUMBER CONSISTENT WITH THE CODE IN COL-UMN 38. SEE NOTE * ON PAGE 4 OF 4.
- 43 PHASE 4 CODE. INPUT HERE A CODE LETTER TO DESIGNATE THE UNITS OF THE NUMBER THAT FOLLOWS FOR THE FREQUENCY OF PHASE 4 (TAKE DOWN) OF THE GIVEN EXPERIMENT. USE ONE OF THE LETTERS IDENTIFIED IN 28 ABOVE.
- 44-47 FREQUENCY OF PHASE 4. INPUT HERE THE NUMBER OF TIMES THAT THE EXPERIMENT EQUIPMENT MUST BE TAKEN DOWN. BE SURE TO USE A NUMBER CONSISTENT WITH THE CODE IN COLUMN 43. SEE NOTE * ON PAGE 4 OF 4:
- 48. LAUNCH PAYLOAD MASS CODE USE ONE OF THE FOLLOWING CODES TO DESIGNATE THE INPUT DESIRED FOR CALCULATION OF EQUIPMENT MASS LANDED ON THE MOON FOR THIS EXPERIMENT-

••••••••••••••••••••••••••••••••••••••	INPUT	MEANING	
	LEAVE BLANK	THIS EXPERIMENT USES MINOR SCIENTIFIC	
		EQUIPMENT. USE THE NUMBER GIVEN WITH THE EXPERIMENT	
••••	A	THIS EXPERIMENT USES MAJOR SCIENTIFIC. EQUIPMENT. USE THE NUMBER IN THE FOLLOW- ING 6 COLUMNS INSTEAD OF THAT GIVEN WITH	
<u></u>		THE EXPERIMENT.	
	<u> </u>	THIS EXPERIMENT USES MINOR SCIENTIFIC EQUIPMENT. HOWEVER, USE THE NUMBER IN THE FOLLOWING 6 COLUMNS INSTEAD OF THAT GIVEN WITH THE EXPERIMENT	
	·		

NOTE- IF THE MASS OF ANY SCIENTIFIC EQUIPMENT USED ON THIS TRAVERSE HAS ALREADY BEEN ACCOUNTED FOR IN THE BASE EXPERIMENT INPUT (CARD I.D. 3-02), BE SURE TO INPUT A OR B (AS APPROPRIATE) . IF NOT, THE COMPUTER WILL COUNT THE EQUIPMENT MASS TWICE.

	<u> </u>	RD	L.C		8-04
--	----------	----	-----	--	------

PAGE 3 OF 4

t

MISSION INPUT - TRAMERSE EXPERIMENTS (CONTINUED)

49-54 LAUNCH PAYLOAD MASS OVERRIDE. INPUT AS FOLLOWS DEPENDING ON INPUT IN COLUMN 48 (SEE NOTE* ON PAGE 4 OF 4) -

۰.	INPUT IN 48	INPUT HERE
	NONE	NONE
. ·	Δ	MASS OVERRIDE, IN KG, IN PLACE OF MASS GIVEN WITH THE EXPERIMENT
	5	MASS OVERRIDE, IN KG, IN PLACE OF MASS GIVEN WITH THE EXPERIMENT

55 NONRECURRING COST CODE. USE ONE OF THE CODES IN 48 ABOVE TO INDICATE THE INPUT THAT IS DESIRED FOR THE NONRECURRING COST OF THE SCIENTIFIC EQUIPMENT ASSOCIATED WITH THIS EX-PERIMENT.

⁵⁶⁻⁶¹ NONRECHRRING COST OVERRIDE. INPUT AS FOLLOWS DEPENDING ON THE INPUT IN COLUMN 55 (SEE NOTE * ON PAGE 4 OF 4) -

<u></u>	INPUT IN 55	INPUT HERE
	NONE	NONE
	Δ	NONRECURRING COST, IN MILLIONS, FOR MINOR SCIENTIFIC EQUIPMENT ONLY. THIS COST IS TO BE USED IN PLACE OF THE COST GIVEN WITH THE EXPERIMENT.
	9	NONRECURPING COST, IN MILLIONS, TO BE USED IN PLACE OF THE COST GIVEN WITH THE EXPERIMENT.

62 RECURRING COST CODE. USE ONE OF THE LETTER CODES GIVEN IN 48 ABOVE TO INDICATE THE INPUT THAT IS DESIRED FOR RECUR-RING HARDWARE PROCUREMENT OF THE SCIENTIFIC EQUIPMENT AS-SOCIATED WITH THIS EXPERIMENT. CARD I.D. 3-04

PAGE 4 OF 4

MISSION INPUT - TRAVERSE EXPERIMENT (CONTINUED)

63-69 RECURRING COST OVERRIDE. INPUT AS FOLLOWS DEPENDING ON THE INPUT IN COLUMN 62 (SEE NOTE * ON PAGE 4 OF 4) -

• •	INPUT IN 62	INPUT HERE
·	NONE	NONE
	Α	RECURRING COST, IN MILLIONS, FOR MINOR SCIENTIFIC EQUIPMENT ONLY. THIS COST IS TO BE USED IN PLACE OF
	B	RECURRING COST, IN MILLIONS, TO BE USED IN PLACE OF THE COST GIVEN
		WITH THE EXPERIMENT.

69 LRV OR LEV CARGO MASS CODE. USE ONE OF THE CODES IN 49 ABOVE TO DESIGNATE THE INPUT DESIRED FOR CALCULATION OF EQUIPMENT MASS TO BE CARRIED ON THE LRV OR LEV FOR THIS EXPERIMENT.

> NOTE- IF TWO EXPERIMENTS ON THE TRAVERSE USE COMMON EQUIP-MENT, BE SURE TO INPUT A OR B (AS APPROPRIATE) ON AT LEAST ONE OF THE EXPERIMENTS. OTHERWISE, THE COMPUTER MAY CALCULATE MORE CARGO MASS THAN IS DESIRED.

70-75 LRV OR LFV CARGO MASS OVERRIDE. INPUT AS FOLLOWS DEPENDING ON THE INPUT IN COLUMN 69 (SEE NOTE * BELOW) -

 INPUT IN 69	INPUT HERE
 NONE	NONE
 Α	MASS OVERRIDE, IN KG, IN PLACE OF MASS GIVEN WITH THE EXPERIMENT
 3	MASS OVERRIDE, IN KG, IN PLACE OF MASS GIVEN WITH THE EXPERIMENT

THE COMPUTER WILL CALCULATE THE CARGO CARRIED BY THE LRV OR LEV ON THIS PARTICULAR TRAVERSE BASED ON THE INPUT HERE. HENCE, CARE SHOULD BE TAKEN WITH THIS INPUT.

NOTE * - PLACE THE INPUT NUMBER ANYWHERE IN THE SPACE PROVIDED. BE SURE TO SHOW THE LOCATION OF THE DECIMAL POINT.

<u>IMPORTANT - RECHECK THE INPUT OF THIS CARD. 'IF THE CODE LETTER A</u> IS USED IN COLUMNS 43, 55, 62, OR 69, IT MUST APPEAR IN ALL FOUR COLUMNS.

Appendix C

DETAILED INSTRUCTIONS FOR PREPARING MODE INPUT CARDS

(Ref. Paragraph 10.5.2)

LOCKHEED MISSILES & SPACE COMPANY

CARD I.D. 4-01

MODE INPUT - GENERAL

THIS CARD IS USED TO SUPPLY GENERAL INPUT DATA ON EACH MODE.

- 1-12 MODE IDENTIFIER. USE THE NUMBER SHOWN IN THE MIMOSA DATA BOOK. INCLUDE ALL DASHES. START THE IDENTIFIER IN COLUMN 1.
- 16-20 ACTIVE MODE CREW LEVEL. INPUT HERE THE NUMBER OF MEN WHO WILL ACTIVELY PERFORM THE MISSION EXPERIMENTS. FOR LUNAR SURFACE MISSIONS, COUNT ONLY THOSE MEN LANDED. SEE NOTE * BELOW.
- 21-25 TOTAL MODE CREW LEVEL. INPUT THE TOTAL CREW LEVEL SENT TO THE MODN. SEE NOTE * BELOW.
- 26-35 MODE STAY TIME. INPUT, IN DAYS, THE MAXIMUM TIME FOR ACCOM-PLISHING SCIENTIFIC EXPERIMENTS PERMITTED BY THE MODE. FOR LUNAR SURFACE MISSIONS, THIS WILL BE THE TIME PERMITTED ON THE SURFACE. SEE NOTE # BELOW.
- 36-45 SAMPLE RETURN CAPABILITY. INPUT THE MAXIMUM MASS, IN KILOGRAMS, THAT CAN BE RETURNED TO EARTH BY THE MODE TRANSPORTATION SYS-TEM. SEE NOTE # BELOW.

46-55 MINIMUM DECIMAL FRACTION TO CHECK FOR EXCESSIVE CAPABILITY. IN-PUT THE DECIMAL FRACTION OF THE MODE STAYTIME THAT MUST BE USED. AS A MINIMUM, DURING THE MISSION. THE COMPUTER WILL THEN CALCULATE THE STAYTIME REQUIRED FOR THE MISSION AND THEN CHECK TO SEE IF IT IS LESS THAN. THIS FRACTION OF THE MODE STAYTIME CAPABILITY. IF THE REQUIREMENT CALCULATED IS TOO LOW, A SPECIAL MESSAGE WILL BE PRINTED OUT. SUGGESTED INPUT = 0.75

SEE NOTE * BELOW.

56-65 LAUNCH LEAD TIME FOR MINOR SCIENTIFIC EQUIPMENT. INPUT NUMBER OF 6-MONTH PERIODS PRIOR TO THE FIRST LAUNCH OF THE MODE THAT ALL MINOR SCIENTIFIC EQUIPMENT MUST BE DELIVERED FOR PAYLOAD INTEGRATION. INPUT 1 FOR 0-6 MONTHS, 2 FOR 7-12 MONTHS, ETC. SEE NOTE # BELOW. SUGGESTED INPUT = 2.0

72	FLAG FOR	CONCUPR	ENT SHELTERS.	INP	UT AS	S FOLL	OWS.	
		0 (OR	LEAVE BLANK)	= NO	CONC	CURREN	IT SHELTERS	
<u>-</u>		1		= TW	<u>0 0Ř</u>	MORE	CONCURRENT	SHELTERS

75 NUMBER OF PAYLOADS LAUNCHES IN THE MODE. IF MORE THAN 8, YOU ARE TN TROUBLE. THE COMPUTER WILL ACCEPT UP TO 8.

77-80 SEQUENCE NUMBER. INPUT THE FOLLOWING NUMBER- 401L

NOTE = - PLACE THE INPUT NUMBER ANYWHERE IN THE SPACE PROVIDED . BE SURE TO SHOW THE LOCATION OF THE DECIMAL POINT.

CARD I.D. 4-02	
MODE INPUT - OPERATIONS	
THIS CARD IS USED TO INDUT CERTAIN OPERATIONAL DATA ON EACH MODE	
THIS CARD TO USED TO INFOR CENTAIN DISCATINAL DATA ON LACH HODE.	
BOOK. INCLUDE ALL DASHES. START THE IDENTIFIER IN COLUMN	1.
19-24 TOTAL TIME FOR BASE DEPLOYMENT. INPUT THE TIME, IN DAYS,	
TIME REQUIRED FOR DEPLOYMENT BY THE CREW. DO NOT COUNT TIME REQUIRED FOR DEPLOYMENT BEFORE THE CREW ARRIVES.	
USE INTEGER DAYS, NOT FRACTIONS. IF MORE THAN 1 HOUR	
OF DEPLOYMENT ACTIVITY IS NEEDED IN A DAY, COUNT THE	
ENTIRE DAY. SEE NOTE * BELOW.	
25-30 SHIRTSLEEVE MAN-HOURS FOR SPECIAL OPERATIONS. COUNT ALL MA	N-
HOURS IN SHIRTSLEEVE ENVIRONMENT FOR DEPLOYMENT OF ALL	
SYSTEMS AND FOR ANY OTHER SPECIAL OPERATIONS. SEE NOTE * BELOW	
31-36 SPACESUIT MAN-HOURS FOR SPECIAL OPERATIONS. SEE NOTE * BELC	W
37-42 MAN-HOURS OF SPACESULT LIFETIME. INPUT THE ESTIMATED NUM-	
BER OF MAN-HOURS OF SPACESUIT USE BEFORE REPLACEMENT OF	
THE SUIT IS CONSIDERED TO BE REQUIRED. SEE NOTE * BELOW.	
SUGGESTED INPUT (MAN-HOURS) = 240.0 (HARD SUIT)	
43-49 MAN-HOURS OF BACKPACK LIFETIME. INPUT ESTIMATED LIFETIME	
OF NON-EXPENDABLE ELEMENTS OF THE SPACESUIT BACKPACK.	
$\frac{\text{SFF NDTF * BEL(IW.}}{\text{SUGGESTED INPUT (MAN-HOURS)}} = 100.0$	
49-54 MASS OF SULT REPLACEMENT. INPUT MASS. IN KILOGRAMS. OF	
REPLACEMENT SPACESUIT. MASS INCLUDES ALL GARMENTS, BUUTS, AND HELMET - SEE NOTE * BELOW	
SUGGESTED INPUT (KG) = 33.0 (HARD SUIT)	
55-60 MASS OF BACKPACK REPLACEMENT. INPUT MASS, IN KILUGRAMS, UF BACKDACK PEDIACEMENT COUNT ALL MASS ITEMS EXCEPT EXPEND-	
ABLES. SEE NOTE * BELOW.	••••••
SUGGESTED INPUT (KG) = 10.0	
77-80 SEQUENCE NUMBER. INSERT THE FULLOWING NUMBER- 402L	
NOTE +	
SUPE TO SHOW THE LOCATION OF THE DECIMAL POINT.	

	· (ARD I.D. N	105 4-05		-05, 4-00		PAGE I	0F 2	
· ·			COST CUR	VE - GENI	FRAL				
		· ·	· · ·						
THIS ST REQUIRE SEQUENC BEING C EACH SE COSTS F	ANDARI D FOR E NUME IVEN. T MAY FOR FIN	CARD IS U THE COMPUT SER (SPACES EACH COST CONTAIN UP /E 6-MONTH TO 7.5 YEA	JSED TO IN FER PROGRA S 77-30) E T CURVE WI P TO THREE TIME PERI ARS. IF I	PUT ALL S M. THE STABLISH LL BE GIV CARDS. ODS. USE T IS NOT	NONRECURI IDENTIFII EXACTLY VEN BY A AS FACH OF THREI DESIRED	RING AN ER (SPA WHICH (SET DF CARD PI E CARDS TO INPI	C RECUR CES 1-1 COST CU THESE ERMITS ALLONS JT A CO	RING (2) ANI RVE I CARDS ENTRY INPU' ST	
CURVE,	FILL	DUT COLUMNS	5 1-12 AND	77-80 01	۹ L Y .				
AT LEAS Quired	FOR FA	OF THESE C	CARDS MUST	BE INPU	FOR EVE	ERY COS	I CATEG	ORY RI	E
1-12		IDENTIFIEF	R. USE TH	E NUMBER	SHOWN I	N THE M	IMOSĂ D		
	3500								• •
21-25	<u></u> OCC TO DE!	UR FROM TH AND INCLUE	HE FIRST N DING THE 6 E. THE FO	-MONTH T	ENTRY IN IME PERI(DEADLINE)	THE COS DD OF A S SHOULD	ST CURV SPECIF D BE US	E UP IC ED -	<u> </u>
			·						
· · ·		COST I	INPUT		DEADL	INE			
· ·		MODE NONE	RECURRING	DAT	E OF FIR	ST LAUN	сн		
	· · ·	MODE RECU	JRRING	DAT	E OF FIRS	ST LAUN	сн	·	
								TCAR	ים כ דו
	N01	E- THIS ZE	ERO POINT F THAT WILL	ENTRY NE	ED ONLY E F FACH CO	BE PUT (DST CUR)	DN FIRS /E. TH	E INPL	-
	NO1	THIS ZE THE SET NUMBER SHOW A	ERO POINT <u>T THAT WIL</u> MAY BE PU DECIMAL P	ENTRY NEI L <u>Decribi</u> T Anywhei Dint with	ED ONLY I E EACH OF E IN THE E EACH EM	BE PUT (DST CUR) E FIVE S NTRY.	DN FIRS /E. TH SPACES		DED
26-35	NO1 COST THE VIT	E- THIS ZE THE SEI NUMBER SHOW A ENTRY, IN ENTRY, IN ENTRY, IN ED, SHOW L	ERO POINT T THAT WILL MAY BE PU DECIMAL P MILLIONS O MSER MAY B LOCATION O	ENTRY NEI L <u>DECRIBI</u> T ANYWHEI DINT WITH DF DOLLAF E PUT ANY E DECIMAN	ED ONLY I <u>E EACH C</u> E IN THE E EACH E ¹ S, FOR C WHERE WI POINT.	BE PUT (DST CUR) FIVE S NTRY. S-MONTH ITHIN TH	DN FIRS /E. TH SPACES FUNDIN HE 10 S	E INPI PROVII G PER PACES	
26-35 36-45	NO1 COST THE VIT	E- THIS ZE THE SET NUMBER SHOW A ENTRY, IN E INPUT NUM ED, SHOW L AS ABOVE.	ERO POINT T THAT WILL MAY BE PU DECIMAL P MILLIONS MSER MAY B LOCATION O	ENTRY NEI L DECRIBI T ANYWHEI DINT WITH DF DOLLAF E PUT ANY F DECIMAN	ED ONLY I E EACH CO E IN THE E EACH E' S, FOR C WHERE WI POINT.	BE PUT (DST CUR) FIVE S NTRY. 6-MONTH ITHIN TH	DN FIRS <u>/E. TH</u> SPACES FUNDIN HE 10 S	E INPI PROVI G PER PACES	
26-35 36-45 46-55	NOT COST THE VIT SAME SAME	E- THIS ZE THE SET NUMBER SHOW A ENTRY, IN EINPUT NUM ED, SHOW L AS ABOVE. AS ABOVE.	ERO POINT T THAT WIL MAY BE PU DECIMAL P MILLIONS MSER MAY B LOCATION O	ENTRY NEI L <u>DECRIBI</u> T ANYWHEI DINT WITH OF DOLLAN E PUT ANY E DECIMAN	ED ONLY I E EACH CI E IN THE E EACH E S, FOR (WHERE WI POINT.	BE PUT (DST CURY FIVE S NTRY. 5-MONTH ITHIN TH	DN FIRS <u>/E. TH</u> SPACES FUNDIN HE 10 S	E INPI PROVII G PER PACES	
26-35 36-45 46-55 56-65	NOT COST THE VIT SAME SAME SAME	E- THIS ZE THE SEI NUMBER SHOW A ENTRY, IN ENTRY,	ERO POINT T THAT WILL MAY BE PU DECIMAL P MILLIONS MBER MAY B LOCATION O	ENTRY NEI L DECRIBI T ANYWHEI DINT WITH DF DOLLAR E PUT AN E DECIMAN	ED ONLY I E EACH CI RE IN THE H EACH EN RS, FOR (WHERE N POINT.	BE PUT (DST CURY E FIVE S NTRY. 6-MONTH ITHIN TH	DN FIRS <u>/E. TH</u> SPACES FUNDIN HE 10 S	E INPI PROVII G PER PACES	
26-35 36-45 46-55 56-65 66-75	NOT COST THE VIT SAME SAME SAME	E- THIS ZE THE SET NUMBER SHOW A ENTRY, IN ENTRY,	ERO POINT T THAT WIL MAY BE PU DECIMAL P MILLIONS MSER MAY B LOCATION O	ENTRY NEI L <u>DECRIBI</u> T ANYWHEI DINT WITH DF DOLLAN E PUT AN F DECIMAN	ED ONLY I E EACH CI RE IN THI E EACH E S, FOR (WHERE W POINT.	BE PUT (DST CURY FIVE S NTRY. 5-MONTH ITHIN TH	DN FIRS <u>/E. TH</u> SPACES FUNDIN HE 10 S	E INPI PROVII G PER PACES	

	COST CHOVE - CENERAL (CONTINUED)	
· · · · · · · ·	LUST CORVE - GENERAL TOUNT INDEDT	
77-80	SEQUENCE NUMBER. INPUT NUMBER AS SHOWN BELOW. NOTE THAT SYMBOL * HAS SPECIAL SIGNIFICANCE.	тне
	COST CURVE	INPU.
. <u></u>	NONRECURBING R AND D COST - MODE LEVEL	403
	NONRECURRING COST OF FACILITIES - MODE LEVEL	404
	RECURRING HARDWARE PROCUREMENT COST - MODE LEVEL	405 ³
	RECURRING OPERATIONS SUPPORT COST - MODE LEVEL	406
	<pre># = LAST DIGIT OF SEQUENCE NUMBER. USE AS FOLLOWS- L - IF LAST CARD IN SEQUENCE OF ANY SIZE</pre>	
	1 - IF FIRST CARD IN SEQUENCE OF 2 OR 3 CARDS 2 - IF SECOND CARD IN SEQUENCE OF 3 CARDS	
	·	
•		
		. <u>,</u>

	CARD I.D. 4-07
•	MODE INDUT - PAYLOAD LAUNCH IDENTIFICATION
· · · · · · · · · · · · · · · · · · ·	ABBE INPUT - PATEDAD LAUNCH IDENTIFICATION
THIS CA	ARD IS USED TO INPUT CERTAIN GENERAL DATA ON EACH PAYLOAD
EICHT I	OF A MUDE. THE CUMPUTER MEMORY ALLUCATION IS LIMITED TO DAVIDADS DER MODE. THUS, HD TO EICHT DE THESE CARDS MAY BE
INPUT.	ATEUADS PER MODEL FINDS, OF TO EIGHT OF THESE CANDS MAT BE
1-12	MODE IDENTIFIER. USE THE NUMBER SHOWN IN THE MIMOSA DATA
	BODK. INCLUDE ALL DASHES. START THE IDENTIFIER IN COLUMN 1.
19	PAYLOAD LAUNCH IDENTIFIER. INPUT THE NUMBER OF THIS LAUNCH
• •	IN THE SEQUENCE OF LAUNCHES IN THE PLOSE.
25	PAYLOAD LAUNCH IDENTIFIER OF RELATED LAUNCH. INPUT THE
	LAUNCH IDENTIFIER OF THE NEAREST PREVIOUS LAUNCH UPON
	WHICH THE CURRENT LAUNCH DEPENDS. FOR EXAMPLE, LAUNCH
	A CREW BOTATION LAUNCH MAY DEPEND ON COMPLETION OF LAUNCH 1, OK
	PREVIOUS CREW DELIVERY. IF NO LAUNCH DEPENDENCY EXISTS,
	LEAVE SPACE BLANK.
31-40	MINIMUM NUMBER OF 6-MONTH TIME PERIODS THAT MUST FLAPSE
21 10	BETWEEN THE TWO LAUNCHES GIVEN ABOVE. IF LESS THAN SIX
	MONTHS IS ACCEPTABLE, INPUT O. SEE NOTE * BELOW. IF
	COLUMN 25 IS BLANK, LEAVE THIS SPACE BLANK.
41-50	MAXIMUM NUMBER OF 6-MONTH TIME PERIODS THAT MAY ELAPSE BE-
	TWEEN THE TWO LAUNCHES GIVEN ABOVE. IF LESS THAN 6-MONTHS
·	IS REQUIRED, INPUT O. SEE NOTE * BELOW. IF COLUMN 25 IS
	BLANK, LEAVE THIS SPACE BLANK.
51-60	MASS MARGIN AVAILABLE. INPUT THE MASS MARGIN, IN KILDGRAMS,
	THAT IS AVAILABLE WITH THIS LAUNCH FOR CARRYING MINDR
	SCIENTIFIC EQUIPMENT, EXTRA EXPENDABLES, AND SPACESUIT
	AND BACKPACK REPLACEMENTS. THE MASS MARGIN WILL GENERAL ALLY BE FOUND BY SUBTRACTING THE MASS OF ALL SYSTEMS AND CREW
	ARRIVING AT THE FINAL DESTINATION OF THE LAUNCH (EXCEPT
·	TRANSPORTATION EQUIPMENT USED IN THE OUTBOUND TRIP) FROM
	THE PAYLOAD DELIVERY CAPABILITY OF THE TRANSPORTATION
	SYSTEM. SEE NUTETA BELOW.
77-20	SEQUENCE NUMBER. INPUT 407X WHERE
	X = 1, 2, ETC UP TO THE LAST LAUNCH
	= L FOR THE LAST LAUNCH /
-	
NOTE #	- PLACE INPUT NUMBER ANYWHERE IN THE SPACE PROVIDED. BE
	SURE TO SHOW THE LOCATION OF THE DECIMAL POINT.

CARD I.D. NO. 4-10

PAGE 1 OF 2

MODE INPUT - SYSTEMS ON EACH PAYLOAD

THIS CARD IS USED TO IDENTIFY EACH SYSTEM MAKING UP EACH PAYLOAD LAUNCH. THE COMPUTER MEMORY ALLOCATION ALLOWS FOR UP TO 15 SYSTEMS. ON EACH LAUNCH. THUS, UP TO 15 OF THESE CARDS CAN BE USED TO FORM A SET FOR A GIVEN LAUNCH. UP TO 9 LAUNCHES (HENCE, 8 SETS OF CARDS) ARE ALLOWED PER MODE. THUS, FOR THE MAXIMUM CASE, 120 OF THESE CARDS WOULD BE NEEDED.

FOR THE GIVEN MODE, FILL OUT AS MANY CARDS TO FORM A SET AS THERE ARE SYSTEMS ON THE LAUNCH. FORM AS MANY SETS AS THERE ARE LAUNCHES IN THE MODE. BE SURE ALL SYSTEMS ARE INCLUDED EXCEPT THOSE SYSTEMS MAKING UP THE TRANSPORTATION SYSTEM. WHEN THE TRANSPORTATION SYSTEM IS SPECIFIED, THE COMPUTER WILL AUTOMATICALLY FIND THE LIST OF REQUIRED FLIGHT SYSTEMS AND LAUNCH VEHICLE. THESE SYSTEMS ARE COUNTED TOWARD THE 15 MAXIMUM.

1-12	MODE IDENTIFIER. USE THE NUMBER SHOWN IN THE MIMOSA DATA BOOK. INCLUDE ALL DASHES. START THE IDENTIFIER IN COLUMN 1.
15	PAYLOAD LAUNCH IDENTIFIER. INPUT THE NUMBER OF THIS LAUNCH IN THE SEQUENCE OF LAUNCHES IN THE MODE.
22-27	SYSTEM IDENTIFIER. USE THE NUMBER SHOWN IN THE MIMOSA DATA BOOK. USE ONLY THE SIX DIGITS. DO NOT INCLUDE THE DASH SEPARATING THE DIGITS. START THE NUMBER IN COLUMN 22.
31-40	NUMBER OF MEN SUPPORTED BY THIS SYSTEM. FOR TRANSPORTATION SYSTEMS, LEAVE THIS SPACE BLANK. SEE NOTE * BELOW.
41-50	LAUNCH LEAD TIME FOR SYSTEM. INPUT NUMBER OF 6-MONTH PERIODS PRECEDING THE LAUNCH DATE WHEN DELIVERY OF THE FLIGHT-
	QUALIFIED SYSTEM IS REQUIRED. INPUT 1 FOR 0-6 MONTHS, 2 FOR 7-12 MONTHS, ETC. SEE NOTE * BELOW. FOR TRANSPOR- TATION SYSTEMS, LEAVE THIS SPACE BLANK.
51-60	EXPENDABLES REQUIRED FOR SPECIAL OPERATIONS. INPUT THE MASS FOR EXPENDABLES, IN KILOGRAMS, REQUIRED FOR ANY
	SPECIAL OPERATIONS. THESE SPECIAL OPERATIONS INCLUDE DEPLOYMENT, CREW APPIVAL AND DEPARTURE, AND SUCH OTHER OPERATIONS AS BEQUIRED. COUNT ALL EXPENDABLES NEEDED
	WHETHER THEY ARE DUMPED OVERBOARD DR NDT. SEE NOTE * BELOW. FOR TRANSPORTATION SYSTEMS, LEAVE THIS SPACE BLANK.

C-7

		· · · · · · · · · · · · · · · · · · ·		·
		CARD 1.D. NO. 4-1	0	PAGE 2 OF 2
	MODE INPUT	- SYSTEMS ON FACH	PAYLOAD (CON	NT)
				•
77-90	SEQUENCE NUMBER A B	• INPUT 41AB = PAYLOAD LAUNCH = NUMBER OF THE C	WHERE IDENTIFIER ARD IN THE S	SET FOR THIS
		LAUNCH. FOR MU DIGIT. BE SURE	MBERS 10-15 TO INPUT TH	USE ONLY LAST TE LETTER L
· · · · ·		<u> </u>		<u>1400 // 191 0L10</u>
10TE ±		ER ANVWHERE IN THE	SPACE PROV	
	TO SHOW THE LO	CATION OF THE DECT	MAL POINT.	101.9• <u>90</u> 30 4
		· · · · · · · · · · · · · · · · · · ·		
	•	:		
	······································			
				·
		· · · · · · · · · · · · · · · · · · ·		
		•	<u> </u>	·
			·	
	• <u> </u>	·····		
				·····
		<u> </u>		,
	·			
	··			
			······	
· · ·				
	and the second second second second second second second second second second second second second second second			



	÷ •
CARD I.D. NO. 4-20	
MODE INPUT - EMERGENCY MODES	······
THIS CARD IS USED TO IDENTIFY THE EMERGENCY MODES ASSOC AN EXPLORATION CONCEPT MODE. ONLY ONE CARD IS REQUIRED	IATED WITH
1-12 MODE IDENTIFIER. USE THE NUMBER SHOWN IN THE M BOOK. INCLUDE ALL DASHES. START THE IDENTIE	IIMOSA DATA IER IN COLUMN L.
25-36 IDENTIFIER FOR CREW RESCUE EMERGENCY MODE. STA TN COLUMN 25. IF NO RESCUE MODE IS NEEDED, L SPACE BLANK.	RT THE NUMBER EAVE THE
49-60 IDENTIFIER FOR LOGISTICS EMERGENCY MODE. START IN COLUMN 49. IF NO RESCUE MODE IS NEEDED, L SPACE BLANK.	THE NUMBER EAVE THE
77-80 SEQUENCE NUMBER. INPUT 420L	
	· · · ·

Appendix D

DETAILED INSTRUCTIONS FOR PREPARING EMERGENCY MODE INPUT CARDS

(Ref. Paragraph 10.5.3)

	<u> </u>						
		CARD I	• • • • • • • • • • • • • • • • • • • •				
EMERGENCY MODE INPUT - GENERAL							
	<u>y - pictus and the city of th</u>						
THIS CA	RD IS USED TO SUPP	LY GENERAL	INPUT DATA ON EAC	H EMERGENCY MODE.			
1-12	EMERICENT, Y MIDE ID	ENITETER.	START THE IDENTIE	IER IN LULUMN I.			
7-80	SEQUENCE NUMBER.	INPUT THE	FOLLOWING NUMBER	- 701L			
			· · · · · · · · · · · · · · · · · · ·				
	99 - 2000 - 2014 - 699 - 200 - 200 - 200 - 200 - 200 - 200 - 200 - 200 - 200 - 200 - 200 - 200 - 200 - 200 - 20	1999 <u>- 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999</u>					
<u> </u>	aga a ga an ann an 19 ann an 19 an 19 an 19 an 19 an 19 an 19 an 19 an 19 an 19 an 19 an 19 an 19 an 19 an 19 a		· · · · · · · · · · · · · · · · · · ·				
		,					

•
CARD I.D. NOS. 7-03, 7-04, 7-05, 7-06 PAGE 1 OF 2

COST CURVE - GENERAL

THIS STANDARD CARD IS USED TO INPUT ALL NONRECURRING AND RECURRING COSTS REQUIRED FOR THE COMPUTER PROGRAM. THE IDENTIFIER (SPACES 1-12) AND SEQUENCE NUMBER (SPACES 77-80) ESTABLISH EXACTLY WHICH COST CURVE IS BEING GIVEN. EACH COST CURVE WILL BE GIVEN BY A SET OF THESE CARDS. EACH SET MAY CONTAIN UP TO THREE CARDS. AS EACH CARD PERMITS ENTRY OF COSTS FOR FIVE 6-MONTH TIME PERIODS, USE OF THREE CARDS ALLOWS INPUT OF COSTS FOR UP TO 7.5 YEARS.

AT LEAST ONE OF THESE CARDS MUST BE INPUT FOR EVERY COST CATEGORY RE-QUIRED FOR EACH MODE. IF IT IS NOT DESIRED TO INPUT A COST CURVE, FILL OUT COLUMNS 1-12 AND 77-90 ONLY.

1-12 MODE IDENTIFIER. USE THE MUMBER INPUT WITH THE 7-01 CARD.

21-25 ZERO POINT. INSERT HERE THE NUMBER OF 6-MONTH TIME PERIODS THAT OCCUR FROM THE FIRST NON-ZERO ENTRY IN THE COST CURVE UP TO AND INCLUDING THE 6-MONTH TIME PERIOD OF A SPECIFIC DEADLINE DATE. THE FOLLOWING DEADLINES SHOULD RE USED -

COST INPUT	DEADLINE	
 MODE NONRECURAING	DATE OF FIRST LAUNCH	
MODE RECUPRING	DATE OF FIRST LAUNCH	

NOTE- THIS ZERO POINT ENTRY NEED ONLY BE PUT ON FIPST CARD OF THE SET THAT WILL DESCRIPE FACH COST CURVE. THE INPUT NUMBER MAY BE PUT ANYWHERE IN THE FIVE SPACES PROVIDED. SHOW A DECIMAL POINT WITH EACH ENTRY.

26-35 COST ENTRY, IN MILLIONS OF DOLLARS, FOR 6-MONTH FUNDING PERIOD. THE INPUT NUMBER MAY BE PUT ANYWHERE WITHIN THE 10 SPACES PRO-VIDED. SHOW LOCATION OF DECIMAL POINT.

36-45 SAME AS ABOVE.

46-55 SAME AS ABOVE.

56-65 SAME AS ABOVE.

66-75 SAME AS ABOVE.

	CARD I.D. 7-03, 7-04, 7-05, AND 7-06 PAG	E 2 0F 2
,	COST CURVE - GENERAL (CONTINUED)	
7-80	SEQUENCE NUMBER. INPUT NUMBER AS SHOWN BELOW. NOTE	THAT THE
	SYMBOL * HAS SPECIAL SIGNIFICANCE.	
	COST CURVE	INPUT
	NONRECHRRING R AND D COST - MODE LEVEL	703*
	NONRECUBRING FACILITIES COST - MODE LEVEL	704*
	RECURRING HARDWARE PROCUREMENT COST - MODE LEVEL	705*
	RECURRING OPERATIONS SUPPORT COST - MODE LEVEL	706*
á	2 - IF SECOND CARD IN SEQUENCE OF 3 CARDS	····
<u></u>		

	. +
	CARD I.D. 7-07
EMERGENCY MODE INPU	JT - PAYLOAD LAUNCH IDENTLEICATION
THIS CARD IS USED TO INPUT CE OF THE EMERGENCY MODE.	ERTAIN GENERAL DATA ON THE PAYLOAD LAUNCH
1-12 MODE IDENTIFIER. USE	THE NUMBER INPUT WITH THE 7-01 CARD.
19 PAYLOAD LAUNCH IDENTI	FIER. INPUT THE NUMBER 1.
77-80 SEQUENCE NUMBER. INP	PUT THE FOLLOWING NUMBER - 707L.
	· ·
· · · · · · · · · · · · · · · · · · ·	
	· · · · · · · · · · · · · · · · · · ·
	· · · · · · · · · · · · · · · · · · ·
·	

CARD I.D. 7-11

EMERGENCY MODE INPUT - SYSTEMS ON THE PAYLOAD

THIS CARD IS USED TO IDENTIFY EACH SYSTEM MAKING UP THE PAYLOAD LAUNCH. THE COMPUTER MEMORY ALLOCATION ALLOWS FOR UP TO 15 SYSTEMS ON THE LAUNCH. THUS, UP TO 15 OF THESE CARDS CAN BE USED.

FOR THE GIVEN MODE, FILL OUT AS MANY CARDS TO FORM A SET AS THERE ARE SYSTEMS ON THE LAUNCH. BE SURE ALL SYSTEMS ARE INCLUDED EXCEPT THOSE SYSTEMS MAKING UP THE TRANSPORTATION SYSTEM. WHEN THE TRANSPORTATION SYSTEM IS SPECIFIED, THE COMPUTER WILL AUTOMATICALLY FIND THE LIST OF REQUIRED FLIGHT SYSTEMS AND LAUNCH VEHICLE. THESE SYSTEMS MUST BE COUNTED TOWARD THE 15 MAXIMUM.

1-12 MODE IDENTIFIER. USE THE NUMBER INPUT WITH THE 7-01 CARD.

15 PAYLOAD LAUNCH IDENTIFIER. INPUT THE NUMBER 1.

22-27 SYSTEM IDENTIFIER. USE THE NUMBER SHOWN IN THE MIMOSA DATA BOOK. USE ONLY THE SIX DIGITS. DO NOT INCLUDE THE DASH SEPARATING THE DIGITS. START THE NUMBER IN COLUMN 22.

31-40 NUMBER OF MEN SUPPORTED BY THIS SYSTEM. FOR TRANSPORTATION SYSTEMS, LEAVE THIS SPACE BLANK. SEE NOTE * BELOW.

41-50 LAUNCH LEAD TIME FOR SYSTEM. INPUT NUMBER OF 6-MONTH PERIODS PRECEDING THE LAUNCH DATE WHEN DELIVERY OF THE FLIGHT-OUALIFIED SYSTEM IS REQUIRED. INPUT 1 FOR 0-6 MONTHS, 2 FOR 7-12 MONTHS, ETC. SEE NOTE * BELOW. FOR TRANSPOR-TATION SYSTEMS, LEAVE THIS SPACE BLANK.

77-80 SEQUENCE NUMBER. INPUT THE FOLLOWING NUMBER - 711X WHERE X = 1, 2, ETC UP TO THE LAST SYSTEM = L FOR THE LAST SYSTEM

NOTE * - PLACE THE NUMBER ANYWHERE IN THE SPACE PROVIDED. BE SURE TO SHOW THE LOCATION OF THE DECIMAL POINT.

D-6

Appendix E

DETAILED INSTRUCTIONS FOR PREPARING SYSTEM INPUT CARDS

(Ref. Paragraph 10.6.1)

SYSTEM INPUT - GENERAL

ONE OF THESE CAPDS MUST BE*SUPPLIED FOR EVERY SYSTEM.

- 1-6 SYSTEM IDENTIFIER. USE THE NUMBER SHOWN IN THE MIMOSA DATA BOOK. USE ONLY THE SIX DIGITS. DO NOT INCLUDE THE DASH SEPARATING THE DIGITS.
- 7-30 SYSTEM NAME. NAME MUST NOT EXCEED 24 UNITS COUNTING ALL LETTERS, NUMBERS, SPACES, AND PUNCTUATION. PLACE THE NAME IN THE CENTER OF THE 24 SPACES PROVIDED.
- 31-40 PROCUREMENT QUANTITY UNIT. PLACE HERE THE NUMBER OF THE SYSTEMS THAT CAN BE PROCURED BEFORE A BACKUP SYSTEM MUST BE PROCUBED. THE COMPUTER WILL THEN SHOW PROCUREMENT OF BACKUP SYSTEMS WHEN MULTIPLES OF THIS NUMBER ARE COUNTED DURING THE PROGRAM ANALYSIS. SEE NOTE * BELOW. SUGGESTED INPUT = 10.0
- 41-50 MAXIMUM MASS PERMITTED FOR ALL EXPENDABLES. INSERT MAXIMUM MASS, IN KILOGRAMS, PERMITTED FOR ALL EXPENDABLES BY THE CONCEPT DESIGN. THIS INCLUDES LS AND ECS EXPENDABLES, FUEL FOR POWER, WATER, LEV REACTION CONTROL GAS, ETC, DO NOT INCLUDE LEV PROPELLANT. SEE NOTE * BELOW.
- NOTE LEAVE THIS SPACE BLANK FOR ALL TRANSPORTATION EQUIP-MENT. FOR LRV TRAILERS, INPUT THE SUM OF THE EXPENDABLES IN THE TRAILER AND IN THE APPROPRIATE LRV.
- 51-60 MAXIMUM MASS OF USABLE LEV PROPELLANT. INSERT THE MAXIMUM MASS. IN KILOGRAMS, PERMITTED FOR USABLE LEV PROPELLANT BY THE TANK SIZES IN THE CONCEPT DESIGN. SEE NOTE * BELOW.
- NOTE LEAVE THIS SPACE BLANK FOR ALL SYSTEMS EXCEPT LEV CONCEPTS AND THOSE SYSTEMS CARRYING REFUEL PROPELLANT.
- 77-80 SEQUENCE NUMBER. INSERT THE FOLLOWING NUMBER- 501L

NOTE # - PLACE THE INPUT NUMBER ANYWHERE IN THE 10 SPACES PROVIDED. BE SURE TO SHOW THE LOCATION OF THE DECIMAL POINT.

SYSTEM INPUT - IDENTIFICATION OF REQUIRED STATE-DF-THE-ART ADVANCES

THIS CARD IS USED TO IDENTIFY WHICH OF THE SUBSYSTEM CONCEPTS USED ON THE SYSTEM WILL REQUIRE AN ADVANCE IN THE STATE-OF-THE-ART (SOTA). AT LEAST ONE CARD MUST BE INPUT FOR FACH SYSTEM. IF NO SOTA ADVANCE IS REQUIRED, SPACES 13-72 ARE LEFT BLANK. FOR EACH SOTA ADVANCE GIVEN, OTHER INPUT CARDS, DETAILING NON-RECURRING COSTS ASSOCIATED WITH THAT SOTA ADVANCE MUST BE SUPPLIED.

- 1-6 SYSTEM IDENTIFIER. USE THE NUMBER SHOWN IN THE MIMOSA DATA BOOK. USE ONLY THE SIX DIGITS. DD NOT INCLUDE THE DASH SEPARATING THE DIGITS.
- 13-18 SUBSYSTEM IDENTIFIER. SUPPLY THE I.D. NUMBER USED TO IDENTIFY THE SUBSYSTEM CONCEPT. USE THE NUMBER SUPPLIED IN THE DATA BOOK. ALWAYS PUT FIRST DIGIT OF I.D. NUMBER AT FAR LEFT OF THE 6 SPACES PROVIDED. USE ONLY THE SIX DIGITS SUPPLIED. DO NOT INCLUDE THE DASH SEPARATING THE DIGITS.

19-24 FIRST ITEM DELIVERY DATE. INSERT THE NUMBER OF 6-MONTH TIME PERIODS PRECEDING REQUIRED SYSTEM DELIVERY THAT THE NEW SUB-SYSTEM SOTA ADVANCE MUST BE COMPLETED. PLACE THE INPUT NUMBER ANYWHERE IN THE 6 SPACES PROVIDED. SHOW A DECIMAL POINT AFTER THE LAST DIGIT OF THE NUMBER.

REPEAT THIS PROCESS FOR EACH IDENTIFIABLE STATE-OF-THE-ART ADVANCE. USE UP TO 3 CARDS (OR 15 SUBSYSTEM ENTRIES). FILL-OUT EACH SUBSEDIENT PATA SET AS FOLLOWS-

25-30	SAME AS 13-18 ABOVE
31-36	SAME AS 19-24 ABOME
3.7-42	SAME AS 13-18 ABOVE
43-49	SAME AS 19-24 ABOVE
49- 54	SAME AS 13-19 ABOVE
55-60	SAME AS 19-24 ABOVE
61-66	SAME AS 13-18 400V5
67-72	SAME AS 19-24 ABOVE
77-20	SEQUENCE NUMBER. INSERT ONE OF THE FOLLOWING NUMBERS 502L IF THIS IS THE LAST CARD DE A SET OF 1, 2, OR 3 CARDS
	5021 IF THIS IS THE FIRST CARD OF A SET OF 1 OR 2 CARDS

IF THIS IS THE SECOND CARD OF A 3-CAPO SET

5022

CARD I.D. NOS. 5-03, 5-04, 5-05, 5-06

PAGE 1 OF 2

SYSTEM COST CURVE

THIS STANDARD CARD IS USED TO INPUT ALL NONRECURRING AND RECURRING COSTS REQUIRED FOR THE COMPUTER PROGRAM. THE IDENTIFIER (SPACES 1-6) AND SEQUENCE NUMBER (SPACES 77-80) ESTABLISH EXACTLY WHICH COST CURVE IS BEING GIVEN. EACH COST CURVE WILL BE GIVEN BY A SET OF THESE CARDS. EACH SET MAY CONTAIN UP TO THREE CARDS. AS EACH CARD PERMITS ENTRY OF COSTS FOR FIVE 6-MONTH TIME PERIODS. USE OF THREE CARDS ALLOWS INPUT OF COSTS FOR UP TO 7.5 YEARS.

AT LEAST ONE OF THESE CARDS MUST BE INPUT FOR EVERY COST CATEGORY RE-OUIRED FOR EACH SYSTEM. IF IT IS NOT DESIRED TO INPUT A COST CURVE, INPUT COLUMNS 1-6 AND 77-80 ONLY.

- 1-6 SYSTEM IDENTIFIER. USE THE NUMBER SHOWN IN THE MIMOSA DATA BOOK. USE ONLY THE SIX DIGITS. DO NOT INCLUDE THE DASH SEPARATING THE DIGITS.
- 21-25 ZERO POINT. INSERT HERE THE NUMBER OF 6-MONTH TIME PERIODS THAT OCCUR FROM THE FIRST NON-ZERO ENTRY IN THE COST CURVE UP TO AND INCLUDING THE 6-MONTH TIME PERIOD OF A SPECIFIC DEADLINE DATE. THE FOLLOWING DEADLINES SHOULD BE USED -

 COST INPUT	DEADLINE
 SYSTEM NONRECURRING	DELIVERY DATE OF FIRST FLIGHT QUAL- IFIED SYSTEM TO PAYLOAD INTEGRATOR
 SYSTEM RECURRING	DELIVERY DATE OF SYSTEM TO PAYLOAD INTEGRATOR

- NOTE- THIS ZERO POINT ENTRY NEED ONLY BE PUT ON FIRST CARD OF THE SET THAT WILL DECRIBE EACH COST CURVE. THE INPUT NUMBER MAY BE PUT ANYWHERE IN THE FIVE SPACES PROVIDED. SHOW A DECIMAL POINT WITH EACH ENTRY.
- 26-35 COST ENTRY, IN MILLIONS OF DOLLAPS, FOR 6-MONTH FUNDING PERIOD. THE INPUT NUMBER MAY BE PUT ANYWHERE WITHIN THE 10 SPACES PRO-VIDED. SHOW LOCATION OF DECIMAL POINT.

36-45 SAME AS ABOVE.

46-55 SAME AS ABOVE.

56-65 SAME AS ABOVE.

66-75 SAME AS ABOVE.

	+
CARD I.D. NOS. 5-03, 5-04, 5-05, 5-06 PAGE 2	OF 2
SYSTEM COST CURVE (CONTINUED)	
77-80 SEQUENCE NUMBER. INPUT NUMBER AS SHOWN BELOW. NOTE TH	AT THE
Stande - 1943 Succise Stoverionance	
COST CURVE	INPUT
NONRECURRING R AND D COST - SYSTEM LEVEL	503*
NUNRECORMING FACILITIES COST - SYSTEM LEVEL	<u> </u>
RECURRING HARDWARE PROCUREMENT COST - SYSTEM LEVEL	505*
RECURRING OPERATIONS SUPPORT - SYSTEM LEVEL	506*
* = LAST DIGIT OF SEQUENCE NUMBER. USE AS FOLLOWS-	
1 - IF FIRST CARD IN SEQUENCE OF 2 OR 3 CARDS	
2 - IF SECOND CARD IN SEQUENCE DF 3 CARDS	
	tille ann vita fferinge angår verstaaraa op 1 aaneer
	, and a second second second second second second second second second second second second second second second

PAGE 1 DF 2

TABLE OF HARDWARE COST REDUCTION FOR ADDITIONAL COPIES

THIS CARD IS USED TO INPUT⁴ AN APPROXIMATION OF THE COST REDUCTION IN HARDWARE PROCUREMENT THAT OCCURS WITH THE PROCUREMENT OF VARIOUS NUMBER OF COPIES. THE TOTAL NUMBER OF COPIES EXPECTED TO BE PROCURED ARE DIVIDED INTO NO MORE THAN 10 GROUPS. THE COST OF EVERY ARTICLE IN EACH GROUP IS THEN GIVEN AS A DECIMAL ERACTION OF THE FIRST ARTICLE PROCURE-MENT COST.

THE COMPUTER, AFTER IDENTIFYING A SYSTEM DESIRED FOR A MISSION, DETER-MINES HOW MANY COPIES OF THAT SYSTEM HAVE BEEN PROCURED. IT THEN CON-SULTS THIS COST REDUCTION TABLE AND DETERMINES THE APPROPRIATE COST RE-DUCTION FACTOP. IT THEN MULTIPLIES EACH ENTRY OF THE COST CURVE FOR RECURRING SYSTEM HARDWARE PROCUREMENT COSTS BY THE REDUCTION FACTOR.

IT IS POSSIBLE TO ENTER THE DECIMAL FRACTION COST OF 5 GROUPS ON ONE CARD. THUS, NO MORE THAN TWO CARDS ARE NEEDED. IN ANY EVENT, AT LEAST ONE CARD MUST BE INPUT FOR EACH SYSTEM.

- 1-6 SYSTEM IDENTIFIER. USE THE NUMBER SHOWN IN THE MIMOSA DATA <u>ROOK.</u> <u>USE ONLY THE SIX DIGITS.</u> <u>DO NOT INCLUDE THE DASH</u> <u>SEPARATING THE DIGITS.</u>
- 13-19 THE NUMBER OF THE LAST ARTICLE IN THE GROUP. IF THIS IS THE FIRST GROUP, IT MUST CONTAIN THE NUMBER OF ARTICLES THAT COST THE SAME AS THE FIRST ARTICLE. FOR EXAMPLE, IF THE FIRST FOUR ARTICLES COST THE SAME, THE NUMBER 4. WOULD BE ENTERED HERE. PLACE THE NUMBER ANYWHERE IN THE 6 SPACES PROVIDED. BE SURE TO ENTER THE DECIMAL POINT.
- 19-24 DECIMAL FRACTION OF FIRST ARTICLE COST FOR EACH MEMBER OF THE ABOVE GROUP. IF THIS IS THE FIRST GROUP, THE NUMBER 1.0 WOULD BE ENTERED HERE. PLACE THE NUMBER ANYWHERE IN THE 6 SPACES PROVIDED. BE SURE TO ENTER THE DECIMAL POINT.
- 25-30. THE NUMBER OF THE LAST ARTICLE IN THE NEXT GROUP. PLACE THE NUMBER ANYWHERE IN THE 6 SPACES PROVIDED. BE SUBE TO ENTER THE DECIMAL POINT.
- 31-36 DECIMAL FRACTION OF FIRST ARTICLE PROCUREMENT COST FOR EACH MEM-BER OF THE ABOVE GROUP. PLACE THE NUMBER ANYWHERE IN THE 6 SPACES PROVIDED, BE SURE TO ENTER THE DECIMAL POINT.

37-42 SEE 25-30 ABOVE.

43-49 SEF 31-36 ABOVE.

	4
	CARD I.D. NO. 5-07 PAGE 2 OF 2
	TABLE OF HARDWARE COST REDUCTION (CONTINUED)
49-54	SEE 25-30 ABOVE.
55-60	SE5 31-36 ABOVE.
61-66	SEE 25-30 ABOVE.
67-72	SEE 31-36 ABOVE.
77-80	SEQUENCE NUMBER. INPUT NUMBERS AS SHOWN BELOW FOR EACH SYSTEM.
	5071 IF THIS IS FIRST OF TWO CARDS.
	507L IF THIS IS THE ONLY CARD OR IF THIS IS SECOND CAPD
<u></u>	
•	

PAGE 1 OF 2

R AND D FLIGHT TABLE

THIS CARD IS USED TO IDENTIFY THOSE NON-MISSION LAUNCHES REQUIRED TO COMPLETE THE R AND D OF A GIVEN SYSTEM. THE NUMBER OF FLIGHTS PER 6-MONTH TIME PERIOD IS RECORDED. UP TO TWO CARDS ARE PERMITTED PER SYS-TEM. AS THERE IS SPACE ON A CARD FOR DATA ON FIVE 6-MONTH TIME PER-IODS, DATA FOR 5 YEARS CAN BE INPUT INTO THE COMPUTER. FILL OUT DATA ON ONLY THOSE 6-MONTH TIME PERIODS IN WHICH FLIGHTS ARE REQUIRED, EXCEPT WHERE A 6-MONTH PERIOD WITH NO FLIGHTS IS IN BETWEEN TWO PERIODS THAT HAVE FLIGHTS. IN THAT CASE, THE INPUT SPACES ON THE CARD FOR THAT 6-MONTH PERIOD.MUST BE LEFT BLANK.

IF NO R AND D TEST FLIGHTS ARE REQUIRED FOR THE SYSTEM CONCEPT BEING INPUT, FILL IN SPACES 1-6 AND 77-80 ONLY.

1-6 SYSTEM IDENTIFIER. USE THE NUMBER SHOWN IN THE MIMOSA DATA BOOK. USE ONLY THE SIX DIGITS. DO NOT INCLUDE THE DASH SEPARATING THE DIGITS.

7-12 ZERO POINT. INPUT HERE THE NUMBER OF 6-MONTH TIME PERIODS AFTEP THE START OF ELIGHT TESTING THAT OCCUR BEFORE THE FIRST FLIGHT OUALIFIED SYSTEM IS DELIVERED TO THE PAYLOAD INTEGRATOR. SEE NOTE * BELOW.

13-18 NUMBER OF R AND D TEST FLIGHTS IN THE FIRST OR THE SIXTH 6-MONTH TIME PERIOD (DEPENDING ON WHETHER THIS IS THE FIRST OR SECOND CARD OF THE TWO INPUT CARDS PERMITTED). SEE NOTE * BELOW.

19-24 TRANSPORTATION SYSTEM USED FOR THE LAUNCHES IN THE GIVEN 6-MONTH TIME PERIOD. NOTE THAT THIS IMPLIES ONLY ONE TRANSPORTATION SYSTEM CONCEPT IS USED PER 6-MONTH PERIOD. IF TWO OR MORE DIEFERENT TRANSPORTATION SYSTEMS ARE USED, SCHEDULE THE LAUNCHES OF THESE SYSTEMS INTO DIFFERENT 6-MONTH PERIODS. INPUT THE SYSTEM IDENTIFIER REQUIRED. START THE NUMBER TO THE LEFT OF THE SPACE PROVIDED.

25-30 NUMBER OF FLIGHTS IN SUBSEQUENT 6-MONTH TIME PERIOD. SEE NOTE * BELOW.

31-36 SEE 19-24 ABOVE.

37-42 SEE 25-30 AROVE.

43-48 SEE 19-24 ABOVE.

NOTE * - PLACE THE INPUT NUMBER ANYWHERE IN THE SPACE PROVIDED. BE SURE TO SHOW THE LOCATION OF THE DECIMAL POINT.

					••		
			DELICHT	TABLE (CO		PAGE 2 UF 2	<u>.</u>
<u></u>							
49-54	SEE 25-30	AROVE.					
55-60	SEE 19-24	ABOVE.	·				
51-66	SEE 25-30	ABOVE.					
67-72	SEE 19-24	ABOVE		·			
77-80	SEQUENCE N THE FIRS NUMBER	NUMBER. ST OF TWO 508L.	INPUT THE CARDS.	NUMBER IN ALL OT	5081 ONLY HER CASES,	IF THIS IS INPUT THE	
				i			
		1-0				ه د دوره میشوند. بر می خود وی می می در از م	
	·						
			· · · · · · · · · · · · · · · · · · ·				
	·						
							<u></u>
						······	

.

SYSTEM EXPENDABLES

THIS CARD IS USED TO IDENTIFY THE RATE OF USAGE THAT IS ESTIMATED FOR A GIVEN SYSTEM WITH A SPECIFIC CREW LEVEL. BECAUSE A SYSTEM CONCEPT MAY HAVE TO OPERATE AT VARIOUS CREW LEVELS, SEVERAL OF THESE CARDS MAY BE NEEDED. SPACE IS PROVIDED IN THE COMPUTER TO PRESENT DATA ON FIVE DIF-FERENT CREW LEVELS. THE CARDS MUST BE FILLED OUT IN THE DESCENDING ORDER OF CREW SIZE.

FOR TRANSPORTATION SYSTEMS AND FLIGHT SYSTEMS (NOT LEV CONCEPTS), FILL IN SPACES 1-6 AND 77-80 ONEY.

- 1-6 SYSTEM IDENTIFIER. USE THE NUMBER SHOWN IN THE MIMOSA DATA BOOK. USE ONLY THE SIX DIGITS. DO NOT INCLUDE THE DASH SEPARATING THE DIGITS.
- 13-15 CREW LEVEL FOR WHICH THE DATA ON THIS CARD APPLIES. SEE NOTE * BELOW. START FIRST CARD WITH LARGEST CREW SIZE. NOTE - FOR LRV TRAILERS, INPUT THE CREW LEVELS ASSOCIATED WITH THE APPROPRIATE LRV AS INDICATED IN THE MIMOSA DATA BOOK.
- 16-25 TOTAL EXPENDABLES, IN KILDGRAMS, EXPENDED PER DAY AT THE ABOVE CREW LEVEL. THIS RATE INCLUDES ALL POWER SYSTEM FUEL, LS AND ECS EXPENDABLES, WATER, ATMOSPHERE LOSS, ETC. COUNT ALL EX-PENDABLES WHETHER DUMPED OVERBOARD OR NOT EXCEPT THOSE REQUIRED FOR SPACESULT OPERATION. SEE NOTE * BELOW.
- 26-35 TOTAL EXPENDABLES, IN KILOGRAMS PER KW-HR, THAT MUST BE USED OVEF AND ABOVE NORMAL NONSCIENTIFIC SYSTEM OPERATION TO POWER AN EXPERIMENT SUPPORTED BY THE SYSTEM. FOR MAJOR SCIENTIFIC FOULPMENT, INPUT ONLY THE KG/KW-HR SUPPLIED BY THE EQUIPMENT CONCEPT ITSELF. IF POWER IS SUPPLIED TO THE EQUIPMENT FROM A BASE POWEP SUPPLY. SHELTER, LRV, DR LEV, THEN LEAVE THIS INPUT SPACE BLANK. SEE NOTE * BELOW.
- 36-45 TOTAL EXPENDABLES, IN KILOGRAMS PER KILOMETER, REQUIRED TO PRO-VIDE MOBILITY FOR A LUNAP POVING VEHICLE. LEAVE THIS SPACE BLANK FOR ALL OTHER SYSTEMS. FOR TRAILERS (WHEN USED WITH THE LRV), COUNT THE SUM OF BOTH THE LRV AND THE TRAILER USAGE PATE. IF INPUT IS DESIRED, SEE NOTE * BELOW.
- 46-55 TOTAL EXPENDABLES, IN KILOGRAMS PER STOP, REQUIRED FOR ANY ACTI-VATION OF DEACTIVATION OF A ROVING OR FLYING VEHICLE. DO NOT COUNT SPACESUIT EXPENDABLES. LEAVE THIS SPACE BLANK FOR SYS-TEMS OTHER THAN LRV AND LEV CONCEPTS. IF INPUT IS DESIRED, SEE NOTE # BELOW.
- NOTE + THE INPUT NUMBER MAY BE PLACED ANYWHERE IN THE SPACE PROVIDED BE SURE TO SHOW THE LOCATION OF THE DECIMAL POINT.

CARD I.D. NO. 5-09 PAGE 2 DF 2 SYSTEM FXPENDABLES (CONTINUED) TOTAL EXPENDABLES, IN KILOGRAMS PER SPACESUIT MAN-HOUR, THAT 56-65 EXCEED THE EXPENDABLES PER SHIRTSLEEVE MAN-HOUP. IE SYSTEM IS OPERATED SOLELY BY SPACESUIT, IT IS SUGGESTED THAT THE TOTAL SPACESUIT EXPENDABLE RATE BE INPUT. SEE NOTE * BELOW. TOTAL EXPENDABLES, IN KILOGRAMS, THAT ARE NOT USABLE BY THE 66-75 SYSTEM THROUGHOUT ITS USE. THIS INCLUDES UNUSABLE BAILOFF DURING THE MISSION AND PRE-MISSION STORAGE, ULLAGE, REFUELING LOSSES, RESERVES, ETC. DO NOT COUNT USABLE LEV PROPELLANT. FOR LRV TRAILERS (WHEN USED WITH, THE LRV) COUNT BOTH TRAILER AND LRV BOILDES. IF INPUT IS DESIRED, SEE NOTE * BELOW. SEQUENCE NUMBER. INPUT THE NUMBER 50°L IF THIS IS LAST 77-90 CARD IN ANY SEQUENCE. INPUT 5091, 5092, 5093, 5094, AS APPROPRIATE, IF THIS IS FIRST THROUGH FOURTH CARD OF A SEQUENCE. NOTE * - THE INPUT NUMBER MAY BE PLACED ANYWHERE IN THE SPACE PROVIDED. BE SUPE TO SHOW THE LOCATION OF THE DECIMAL POINT.

PAGE 1 OF 2

SYSTEM MAN-HOUR REQUIPEMENTS FOR NONSCIENTIFIC OPERATIONS

THIS CARD IS USED TO IDENTIFY THE NONSCIENTIFIC SPACESUIT AND SHIPTSLEEV MAN-HOURS REQUIRED TO SUPPORT EACH SYSTEM CONCEPT FOR A GIVEN CREW LEVEL BECAUSE A SYSTEM CONCEPT MAY HAVE TO OPERATE AT VAPIOUS CREW LEVELS, SEVERAL OF THESE CARDS MAY BE NEEDED. SPACE IS PROVIDED IN THE COMPUTEP TO PRESENT DATA ON FIVE DIFFERENT CREW LEVELS. THE CREW LEVELS USED MUS BE THE SAME AS THOSE IDENTIFIED ON THE SYSTEM EXPENDABLE CARD (CARD I.D. 5-05). FURTHERMORE, THE CREW LEVELS MUST BE IN THE SAME DESCENDING ORDER.

- FOR TRANSPORTATION SYSTEMS AND FLIGHT SYSTEMS, INPUT COLUMNS 1-6 AND 77-80 ONLY.
- 1-6 SYSTEM IDENTIFIER. USE THE NUMBER SHOWN IN THE MIMOSA DATA BOOK. USE ONLY THE SIX DIGITS. DO NOT INCLUDE THE DASH SEPARATING THE DIGITS.
- 13-15 CREW LEVEL FOR WHICH THE DATA ON THIS CARD APPLIES. SEE NOTE * BELOW.

16-25 SHIRTSLEEVE MAN-HOURS FOR ACTIVATION AND DEACTIVATION OF SYSTEM CONCEPT AT THE ABOVE CREW LEVEL. THIS IS MEANT TO INCLUDE ALL OPERATIONS TO ACTIVATE AND DEACTIVATE CONCEPT, BUT IS NOT MEAN TO INCLUDE DEPLOYMENT. SEE NOTE * BELOW.

26-35 SPACESUIT MAN-HOURS FOR ACTIVATION AND DEACTIVATION AS DESCRIBED ABOVE. SEE NOTE * BELOW.

36-45 SHIRTSLEEVE MAN-HOURS PER DAY FOR NONSCIENTIFIC OPERATIONS. THIS INCLUDES MAN-HOURS PER DAY FOR MAINTENANCE, REPAIRS, STA-TION-KEEPING, AIRLOCK OPERATIONS, AND PERSONAL OPERATIONS. SEE NOTE * BELOW.

- 46-55 SPACESUIT MAN-HOURS PER DAY FOR NONSCIENTIFIC OPERATIONS AS DESCRIBED ABOVE. SEE NOTE # BELOW.
- 56-65 SHIRTSLEEVE MAN-HOURS PER STOP ON A TRAVERSE FOR NONSCIENTIFIC OPERATIONS. THIS INCLUDES OPERATIONS FOR VEHICLE SHUTDOWN. CHECKOUT, AND RESTART AT EACH STOP. IT DOES NOT COUNT ACTIVA-TION AND DEACTIVATION AT THE BEGINNING AND THE END OF THE TRAVERSE. FOR ALL SYSTEMS OTHER THAN LRV AND LEV CONCEPTS, LEAVE THIS SPACE BLANK. IF INPUT IS DESIRED, SEE NOTE * BELOW
- 66-75 SPACESULT MAN-HOURS PER STOP ON A TRAVEPSE AS DESCRIBED ABOVE. IF INPUT IS DESIRED, SEE NOTE # BELOW.
- NOTE # THE INPUT NUMPER MAY BE PLACED ANYWHERE IN THE SPACE PROVIDED. BE SURE TO SHOW LOCATION OF THE DECIMAL POINT.

	CARD I.D. NO. 5-10 PAGE 2 OF 2
	SYSTEM MAN-HOUR REQUIREMENTS (CONTINUED)
77-80	SEQUENCE NUMBER. INPUT THE NUMBER 510L IF THIS CARD IS FOR TRANSPORTATION SYSTEM, FLIGHT SYSTEM, OR UNMANNED SYSTEM, OR IF THIS IS THE LAST CARD IN ANY SEQUENCE. INPUT 5101,
	5102, 5103, OR 5104, AS APPPOPRIATE, IF THIS IS THE FIRST THROUGH THE FOURTH CARD OF A SEQUENCE.
· · ·	
	· ·
	,

.

SYSTEM INPUT - PERSONNEL SHELTER ONLY

THIS CARD IS USED TO RELATE SHELTER STAYTIME TO ALTERNATE SHELTER OFEN LEVELS. THE CREW LEVELS USED MUST BE THE SAME AS THOSE USED IN CAMPS 5-09 AND 5-10. FOR DATA INPUT ON SYSTEMS OTHER THAN SHELTERS, ISHED THIS CARD.

1-6 SYSTEM IDENTIFIER. USE THE NUMBER SHOWN IN THE MIMOSA DATA <u>BOOK.</u> USE ONLY THE SIX DIGITS. DO: NOT INCLUDE THE DASH SEPARATING THE DIGITS.

7-12 MAXIMUM CREW LEVEL. INPUT THE LARGEST CREW SIZE USED FOR INPUT CARDS 5-09 (SYSTEM EXPENDABLES) AND 5-10 (SYSTEM MAN-HOUR RE-OUIREMENTS). SEE NOTE * BELOW.

13-18 MAXIMUM ALLOWABLE STAYTIME, IN DAYS, AT THE GIVEN CREW LEVEL. THIS STAYTIME INPUT IGNORES THE POSSIBILITY THAT PART OF THE CREW MAY SPEND SOME TIME AWAY FROM THE SHELTER. SEE NOTE * BELOW.

<u>19-24 NEXT LARGEST CREW LEVEL CONSIDERED IN CARDS 5-09 AND 5-10. SEE</u> NOTE * BELOW.

25-30 SEE 13-19 ABOVE.

31-36 SEE 19-24 ABOVE.

37-42 SEE 13-18 ABOVE.

43-49 SEE 19-24 ABOVE.

49-54 SEE 13-13 ABOVE.

55-60 SEE 19-24 ABOVE.

61-66 SEE 13-19 ABOVE.

77-80 SEQUENCE NUMBER. INPUT THE FOLLOWING NUMBER - 511L

NOTE * - THE INPUT NUMBER MAY BE PLACED ANYWHERE IN THE SPACE PROVIDED. BE SURE TO SHEW LOCATION OF THE DECIMAL POINT.

PAGE 1 OF 2

SYSTEM INPUT - LRV ONLY

THIS CARD IS USED TO SUPPLY CERTAIN INPUT DATA FOR LRV CONCEPTS WHICH MAY BE SENSITIVE TO CREW LEVEL. IT IS PERMITTED TO INPUT FIVE CARDS FOP FIVE DIFFERENT CREW LEVELS. THE CREW LEVELS USED HERE MUST BE THE SAME AS THOSE USED IN CARDS 5-09 AND 5-10. IN ADDITION, THEY MUST BE IN THE SAME DESCENDING ORDER IN SIZE.

FOR THE PURPOSES OF THIS CARD, A TRAILER, WHEN USED WITH AN LEV, WILL REQUIRE INPUT HERE OF VALUES FOR THE TWO ACTING AS A SINGLE UNIT. FOR DATA INPUT ON SYSTEMS OTHER THAN LEV CONCEPTS, IGNORE THIS CARD.

- 1-6 SYSTEM IDENTIFIER. USE THE NUMBER SHOWN IN THE MIMOSA DATA <u>BOOK.</u> USE ONLY THE SIX DIGITS. DO NOT INCLUDE THE DASH SEPARATING THE DIGITS.
- 11-20 CREW LEVEL. ON THE FIRST CAPD, INPUT THE LAPGEST CREW SIZE USED FOR INPUT CARDS 5-09 (SYSTEM EXPENDABLES) AND 5-10 (SYSTEM MAN-HOUR REQUIREMENTS). THE INPUT ON SUBSEQUENT CARDS IN SEP-IES MUST COVER THE OTHER CREW LEVELS IN 5-09 AND 5-10 IN THE SAME DESCENDING ORDER. SEE NOTE * BELOW.
- 21-30 NOMINAL SINGLE EXCURSION STAYTIME, IN DAYS, PERMITTED FOR THE GIVEN CREW LEVEL. ALL INDIVIDUAL TRAVERSES WITH WHICH THIS CONCEPT IS COMPARED SHOULD HAVE TOTAL TRAVERSE TIMES OF ABOUT THE VALUE INPUT HERE. SEE NOTE * BELOW.
- 31-40 NOMINAL SINGLE EXCURSION RANGE. IN KILOMETERS, PERMITTED FOR THE GIVEN OPEW LEVEL. THIS IS THE NOMINAL RANGE FOR ANY TRAVERSE WITHOUT VEHICLE REFUELING. SEE NOTE # BELOW.
- 41-50 AVEPAGE VEHICLE SPEED, IN KILOMETERS PER HOUR. TRAVERSE PANGE WILL BE DIVIDED BY THIS INPUT TO ESTIMATE THE HOURS REQUIRED FOR TRAVELING. SEE NOTE # BELOW.
- 51-60 MAXIMUM PAYLOAD (OR CARGO) MASS CAPABILITY, IN KILOGRAMS. THIS IS THE MAXIMUM MASS FOR PAYLOAD, SUCH AS SCIENTIFIC EQUIPMENT, PERMITTED BY THE VEHICLE DESIGN FOR THE GIVEN CREW LEVEL. SEE NOTE # BELOW.

CARD' I.D. NO. 5-12 PAGE 2: DE 2 SYSTEM INPUT - LRV ONLY (GONTINUED) 77-80 SEQUENCE NUMBER. INPUT ONE OF THE FOLLOWING NUMBERS. 512L TE THIS IS THE ONLY CARD REQUIRED OR LE THIS IS THE LAST CAPD OF A SERIES. IF THIS IS FIRST CARD OF A SERIES. 5121 5122 IT THIS IS SECOND CARD OF A SERIES. 5123 JE THIS IS THIRD CARD OF A SERIES. IF THIS IS FOURTH CARD OF A SERIES. 5124

÷

CABD 1.0. MD. 5-13.

SYSTEM INPUT - LEV ONLY

THIS CARD IS USED TO SUPPLY CERTAIN INPUT DATA FOR LEV CONCEPTS WHICH MAY BE SENSITIVE TO CREW LEVEL. IT IS PERMITTED TO INPUT FIVE CARDS FOR FIVE DIFFERENT CREW LEVELS. THE CREW LEVELS USED HERE MUST BE THE SAME AS THOSE USED IN CARDS 5-09 AND 5-10. IN ADDITION, THEY MUST BE IN THE SAME DESCENDING ORDER IN SIZE.

FOR DATA INPUT ON SYSTEMS OTHER THAN LEV CONCEPTS, IGNORE THIS CARD.

- 1-6 SYSTEM IDENTIFIER. USE THE NUMBER SHOWN IN THE MIMOSA DATA BOOK. USE ONLY THE SIX DIGITS. DO NOT INCLUDE THE DASH SEPARATING THE DIGITS.
- 11-20 CREW LEVEL. ON THE FIRST CARD, INPUT THE LARGEST CREW SIZE THAT WAS USED FOR INPUT CARDS 5-09 (SYSTEM EXPENDABLES) AND 5-10 (SYSTEM MAN-HOUR REQUIREMENTS). THE INPUT ON SUBSEQUENT CARDS IN A SERIES MUST COVER THE OTHER CREW LEVELS IN 5-09 AND 5-10 IN THE SAME DESCENDING ORDER. SEE NOTE * BELOW.
- 21-30 MAXIMUM SINGLE EXCURSION TIME, IN HOURS, PERMITTED FOR GIVEN CREW LEVEL. ALL INDIVIDUAL TRAVEPSES WITH WHICH THIS CONCEPT IS COMPARED MUST HAVE TOTAL TRAVERSE TIMES LESS THAN THE VALUE INPUT HERE. SEE NOTE * BELOW.
- 31-40 SPECIFIC IMPULSE, IN METERS PER SECOND, FOR THE LEV PROPULSION SUBSYSTEM. IF VALUES ARE NOT AVAILABLE FOR THE CONCEPT DE-SIGN, USE A NOMINAL VACUUM ISP FOR THE DESIRED PROPELLANTS. SEE NOTE * BELOW.
- 41-50 INERT MASS OF LEV IN KILOGRAMS. THE INERT MASS IS GIVEN AS THE SUM OF THE TOTAL NONEXPENDABLES IN THE CONCEPT DESIGN PLUS THE UNUSABLE LEV PROPELLANT. IT DOES NOT INCLUDE THE CREW, THEIR SPACESULTS AND BACKPACKS, USABLE LEV PROPELLANT. AND ALL OTHER EXPENDABLES. SEE NOTE * BELOW.
- 72 SPACESULT USAGE FLAG. INPUT O (ZEPH) OR LEAVE THE SPACE BLANK IF SPACESULT EXPENDABLES ARE USED DURING TRAVELING. INPUT 1 (ONE) IF TRAVELING IS DONE IN SHIRTSLEEVES.

77-80SEQUENCE NUMBER.INPUT ONE OF THE FOLLOWING.513LIF THIS IS THE ONLY CARD REQUIRED OR IF THIS IS THE
LAST CARD OF A SEPIES5131IF THIS IS THE FIRST CARD OF A SERIES.5132IF THIS IS THE SECOND CARD OF A SERIES.5133IF THIS IS THE THIRD CARD OF A SERIES.5134IF THIS IS THE FOURTH CARD OF A SERIES.

MOTE * - THE INPUT NUMBER MAY BE PLACED ANYWHERE IN THE SPACE PROVIDED. BE SURE TO SHOW LOCATION OF THE DECIMAL POINT.

SYSTEM INPUT - LEV STOPS VS RANGE TABLE.

THIS CARD IS USED TO SPECIFY THE RELATIONSHIP BETWEEN THE MAXIMUM SINGLE EXCURSION RANGE OF AN LEV ON A TRAVERSE AND THE NUMBER OF STOPS ON THAT TRAVERSE. THE CARD PROVIDES FOR IMPUT OF FIVE ORDERED PAIRS. UP TO TWO CARDS MAY BE USED.

FOR SYSTEMS OTHER THAN LEV CONCEPTS, IGNORE THIS CARD.

1-6 SYSTEM IDENTIFIER. USE THE NUMBER SHOWN IN THE MIMOSA DATA BOOK. USE ONLY THE SIX DIGITS. DO NOT INCLUDE THE DASH SEPARATING THE DIGITS.

13-18 NUMBER OF STOPS. SPECIFY A VALUE FOR NUMBER OF STOPS BY THE LEV IN ONE TRAVERSE. DO NOT COUNT THE START OR THE FINISH POINT OF THE POSSIBLE TRAVERSE. BEGIN WITH THE NUMBER ZERO AND ON SUCCEEDING ENTRIES, INPUT 1, 2, 3, ETC. IN INCREASING ORDER. SEE NOTE # BELOW.

19-24 MAXIMUM SINGLE EXCURSION RANGE OF THE LEV, IN KILOMETERS, FOR THE NUMBER OF STOPS JUST GIVEN. GENERALLY, THIS WILL BE THE RANGE ACHIEVED WITH ZERO PAYLOAD. SEE NOTE * BELOW.

25-30 SEE 13-18 ABOVE.

:

31-36 SEE 19-24 ABOVE.

37-42 SEE 13-18 ABOVE.

43-48 SEE 19-24 ABOVE.

49-54 SEE 13-18 ABOVE.

55-60 SEE 19-24 ABOVE.

61-66 SEE 13-19 ABOVE.

67-72 SEE 19-24 ABOVE.

77-90 SEQUENCE NUMBER. INPUT THE NUMPER 514L IN ALL CASES EXCEPT IF THIS IS THE FIRST OF TWO CARDS. IN THAT CASE, INPUT THE NUMBER 5141.

NDIE * - THE INDUIT NUMBER MAY BE PLACED ANYWHERE IN THE SPACE PROVIDED. BE SURE TO SHOW LOCATION OF THE DECIMAL POINT.

E-18

+

SYSTEM INPUT - MAJOR SCIENTIFIC FOULPMENT ONLY

THIS CARD IS USED TO IDENTIFY THOSE EXPERIMENTS PERFORMED BY A MAJOR	
SCIENTIFIC FOULPMENT SYSTEM. UP TO FIVE EXPERIMENT IDENTIFIERS CAN BE	-
INPUT ON ONE CARD. UP TO FIVE CARDS MAY BE USED. NOT ALL OF THE 25	
POSSIBLE EXPERIMENTS NEED BE PERFORMED ON ANY ONE MISSION.	
FOR SYSTEMS OTHER THAN MAJOR SCIENTIFIC EQUIPMENT, IGNORE THIS CARD.	
1-6 SYSTEM IDENTIFIER. USE THE NUMBER SHOWN IN THE MIMOSA DATA BOOK. USE ONLY THE SIX DIGITS. DO NOT INCLUDE THE DASH	
SEPARATING THE DIGITS.	
7-18 EXPERIMENT IDENTIFIEP. INPUT THE EXPERIMENT IDENTIFIED OF AN	
EXPERIMENT PERFORMED WITH THE USE OF THE GIVEN MAJOR SCIENTI	C I C
EQUIPMENT CONCEPT. USE THE NUMBER PROVIDED BY THE NOFTH AME	10 -
ICAN AVIATION STUDY OF LUNAR MISSIONS. EXPAND THEIR 8-DIGIT	
NUMBER INTO FIVE SECTIONS WITH DECIMAL POINTS AS FOLLOWS.	
X • X • XX • XX • XX	_
BEGIN THIS 12-PLACE NUMBER AT THE LEFT OF THE SPACE PROVIDED	1
19-30 SEE 6-13 ABOVE.	
31-42 SEE 6-18 ABOVE.	
43-54 SEE 6-18 ABOVE.	
55-66 SEE 6-19 ABOVE.	
77-80 SEQUENCE NUMBER, INPUT ONE DE THE EDITOWING.	
515L IF THIS IS THE ONLY CARD REQUIRED OR IF THIS IS LAST OF A SERIES.	THE
5151 IF THIS IS THE FIRST OF A SERIES.	
5152 IF THIS IS THE SECOND OF A SERIES.	
5153 IF THIS IS THE THIRD OF A SERIES.	
5154 IF THIS IS THE FOURTH OF A SERIES.	
· · · · · · · · · · · · · · · · · · ·	

SYSTEM INPUT - TRANSPORTATION SYSTEMS ONLY

THIS CARD IS USED TO IDENTIFY THE FLIGHT SYSTEMS MAKING UP THE TRANS-PORTATION SYSTEM. IT ALSO INDICATES THE LAUNCH LEAD TIME REQUIRED FOR THE DELIVERY OF EACH FLIGHT SYSTEM.

ONE OF THESE CARDS MUST BE INPUT WITH EACH TRANSPORTATION SYSTEM.

1-6 SYSTEM IDENTIFIER. INPUT THE TRANSPORTATION SYSTEM IDENT-IFICATION NUMBER SUPPLIED IN THE MIMOSA DATA BOOK. USE ONLY THE SIX DIGITS. DO NOT USE THE DASH SEPARATING THE DIGITS.

11-16 FLIGHT SYSTEM IDENTIFIER. INPUT THE SYSTEM IDENTIFIER FOR A FLIGHT SYSTEM USED TO MAKE UP THE TRANSPORTATION SYSTEM. INPUT THE NUMBER SUPPLIED.IN THE MIMOSA DATA BOOK. USE ONLY THE SIX DIGITS. DO NOT USE THE DASH SEPARATING THE DIGITS.

17-20 LAUNCH LEAD TIME FOR THE GIVEN FLIGHT SYSTEM. INPUT THE NUMBER OF 6-MONTH TIME PERIODS PRECEDING THE LAUNCH DATE WHEN DELIVERY OF THE FLIGHT-OUALIFIED SYSTEM IS REQUIRED. INPUT 1 FOR 0-6 MONTHS, 2 FOR 7-12 MONTHS, ETC. PLACE THE NUMBER ANYWHERE IN THE FOUR SPACES PROVIDED. BE SUPE TO SHOW LOCATION OF THE DECIMAL POINT.

21-26 SEE 11-16 ABOVE.

27-30 SEE 17-20 ABOVE.

31-36 SEE 11-16 ABOVE.

37-40 SEE 17-20 ABOVE.

41-46 SEE 11-16 4BOVE.

47-50 SEE 17-20 ABOVE.

51-56 SEE 11-16 ABOVE.

57-60 SEE 17-20 AROVE.

61-66 SEE 11-16 ABOVE.

67-70 SEE 17-20 ABOVE.

77-RO SEQUENCE NUMBER. INPUT THE FOLLOWING NUMBER- 516L

Appendix F

DETAILED INSTRUCTIONS FOR PREPARING SUBSYSTEM SOTA ADVANCE INPUT CARDS

(Ref. Paragraph 10.6.2)

· <u></u>	CARD 1.D. 6-01 .
	CURSYCTEM SOLA ADVANCE = CENEDAL
	SUDSESTEM SULA AUVANUE - GENERAL
ONE OF T	HESE CARDS IS REQUIRED FOR EVERY SUBSYSTEM SOTA ADVANCE.
1-6	SUBSYSTEM IDENTIFIER. START IDENTIFIER IN COLUMN 1.
7-30	SUBSYSTEM NAME FIRST LINE. CENTER THE NAME WITHIN THE SPACE PROVIDED. THIS INPUT WILL APPEAR ON FIRST LINE OF PRINTOUT.
31-54	SURSYSTEM NAME SECOND LINE. CENTER THE REST OF THE NAME WITHIN THE SPACE PROVIDED. THIS INPUT WILL APPEAR ON SECOND
77-90	SECHENCE NUMBER. INDUT THE EDITORING NUMBER - 6011
	SCHOLADE ADDREED THE FOLLOWING ADDREED TO TE
,	
·.	
<u></u>	
<u></u>	F-2

CARD I.D. 6-02 AND 6-03

COST CURVE - GENERAL

THIS STANDARD CARD IS USED TO INPUT ALL NONRECURRING COSTS REQUIRED FOR THE COMPUTER PROGRAM. THE IDENTIFIER (SPACES 1-6) AND SEQUENCE NUMBER (SPACES 77-80) ESTABLISH EXACTLY WHICH COST CURVE IS BEING GIVEN. EACH COST CURVE WILL RE GIVEN BY A SET OF THESE CARDS. EACH SET MAY CONTAIN UP TO THREE CARDS. AS EACH CARD PERMITS ENTRY OF COSTS FOR FIVE 6-MONTH TIME PERIODS, USE OF THREE CARDS ALLOWS INPUT OF COSTS FOR UP TO 7.5 YEARS.

AT LEAST ONE OF THESE CARDS MUST BE INPUT FOR EVERY COST CATEGORY RE-QUIRED FOR EACH SUBSYSTEM SOTA ADVANCE. IF IT IS NOT DESIRED TO INPUT A COST CURVE, FILL OUT COLUMNS 1-12 AND 77-90 ONLY.

1-6 SUBSYSTEM IDENTIFIER. USE THE NUMBER INPUT WITH THE 6-01 CARD.

21-25 ZERO POINT. INSERT HERE THE NUMBER OF 6-MONTH TIME PERIODS THAT OCCUR FROM START OF FUNDING UP TO AND INCLUDING THE 6-MONTH TIME PERIOD OF THE DELIVERY DATE OF FIRST FLIGHT OUAL LELED SUBSYSTEM TO SYSTEM INTEGRATOR.

> NOTE- THIS ZERO POINT ENTRY NEED ONLY BE PUT ON FIRST CARD OF THE SET THAT WILL DESCRIBE EACH COST CURVE. THE INPUT NUMBER MAY BE PUT ANYWHERE IN THE FIVE SPACES PROVIDED. SHOW A DECIMAL POINT WITH EACH ENTRY.

26-35 COST ENTRY, IN MILLIONS OF DOLLARS, FOR 6-MONTH FUNDING PERIOD. THE INPUT NUMBER MAY BE PUT ANYWHERE WITHIN THE 10 SPACES PROVIDED. SHOW LOCATION OF DECIMAL POINT.

36-45 SAME AS ABOVE.

46-55 SAME AS ABOVE.

56-65 SAME AS AROVE.

66-75 SAME AS ABOVE.

77-80 SEQUENCE NUMBER. INPUT NUMBER AS SHOWN BELOW. NOTE THAT THE SYMBOL # HAS SPECIAL SIGNIFICANCE.

COST CURVE	INDIJT
NONRECURRING R AND D COST - SUB	SYSTEM LEVEL 602#
NONRECHERING FACILITIES COST -	SUBSYSTEM LEVEL 603#
<pre># = LAST DIGIT OF SEQUENCE NUMBER</pre>	. USE AS FOLLOWS-
L - IF LAST CARD IN SEQUEN	E OF ANY SIZE
1 - IF FIRST CARD IN SEQUE	ICE OF 2 OR 3 CARDS
2 - TE SECOND CARD IN SECU	INCE OF 3 CARDS

Appendix G

DETAILED INSTRUCTIONS FOR PREPARING EXPERIMENT INPUT CARDS

(Ref. Paragraph 10.6.3)

	CARD I.D. NO. 8-01 PAGE 1 OF 2
	EXPERIMENT INPUT - TYPE 1 CARD
THIS C	ARD IS USED TO INPUT SUMMARY DATA FOR EACH EXPERIMENT.
1	CARD TYPE. INPUT NUMBER 1.
4-11	EXPERIMENT NUMBER. INPUT AN EIGHT DIGIT NUMBER WHICH UNIQUELY IDENTIFIES THIS EXPERIMENT.
12-31	ABBREVIATED TITLE OF EXPERIMENT.
42	NUMBER OF MEN. MINIMUM NUMBER OF CREW PERSONNEL REQUIRED TO PERFORM THE CRITICAL OPERATIONS RELATED TO THIS EXPERIMENT.
45-48	EQUIPMENT MASS. INPUT TOTAL MASS, IN KILOGRAMS, NOT INCLUDING PACKAGING, STORAGE, OF TRANSPORTATION FACTORS. THE FOUR DIGIT CODE USES THE FIRST TWO DIGITS FOR THE SIGNIFICANT FIGURES, AND THE LAST TWO DIGITS FOR THE EXPONENT TO THE BASE 10. FOR NEGATIVE EXPONENTS ADD 50 TO THE ABSOLUTE VALUE.
	$\frac{1}{1} = 20 \text{ KG}$ $\frac{-1}{2051} = 2.0 \text{ X} (10) = 0.2 \text{ KG}$
49-52	EQUIPMENT VOLUME. INPUT TOTAL VOLUME, IN CUBIC METERS, OF OUTSIC ENVELOPE OF EQUIPMENT PACKAGE, NOT INCLUDING SPECIAL CONTAINER REQUIREMENTS. USE FOUR DIGIT CODE AS PREVIOUSLY DESCRIBED.
<u>53-56</u>	TOTAL EQUIPMENT ENERGY. INPUT, IN WATT-HRS, PRODUCT OF AVERAGE POWER DEMAND AND TOTAL OPERATING TIME. USE FOUR DIGIT CODE AS PREVIOUSLY DESCRIBED.
57-60	PEAK POWER. INPUT MAX POWER DEMAND, IN WATTS, FOR PERFORMING SPECIAL FUNCTIONS.
70	DEVELOPMENT TIME. INPUT THE NUMBER OF YEARS REQUIRED TO DEVELOP THE OPERATIONAL PIECE OF EQUIPMENT WHICH IS PACING FOR THE EXPERIMENT.
71-72	YEAR OF EARLIEST AVAILABILITY. LAST TWO DIGITS OF CALENDAR YEAR PACING FOULPMENT COULD BE AVAILABLE, INCLUDING PAST IF ALREADY AVAILABLE.

PAGE 2 OF 2

EXPERIMENT	INPUT -	TYPE 1	CARD	(CONT)	(NUED)
------------	---------	--------	------	--------	--------

- 73-74 NONRECURRING COST. A TWO-DIGIT CODE FOR COST TO DEVELOP ITEM(S) SUITABLE FOR SPACE USE. FIRST DIGIT IS SIGNIFICANT FIGURE, SECOND DIGIT IS FOR EXPONENT.
- 75-76 FIRST ITEM COST. A TWO-DIGIT CODE FOR THE ESTIMATED COST OF THE FIRST ITEM INTENDED FOR OPERATIONAL USE. USE CODING PROCEDURE AS PREVIOUSLY DESCRIBED.
- 77-80 TOTAL EARTH RETURNED MASS. INPUT MASS, IN KILOGRAMS, DE ALL SAMPLES, EMULSIONS, AND OTHER MATERIALS INCLUDING SUITABLE STORAGE CONTAINERS NOT INTEGRAL WITH THE SPACECRAFT. USE FOUP-DIGIT CODE PREVIOUSLY DESCRIBED.

. .

CARD I.D. NO. 8-02 PAGE 1 OF 2
EXPERIMENT INPUT - TYPE 3 CARD
THIS CARD IS USED TO INPUT LABOR OPERATION MANHOUP'S FOR EACH EXPERIMENT
1 CARD TYPE. INPUT NUMBER 3
4-11 EXPERIMENT NUMBER. INPUT AN EIGHT DIGIT NUMBER WHICH UNIQUELY IDENTIFIES THIS EXPERIMENT. USE SAME NUMBER AS ON 9-01 CARD.
THE NEXT EIGHT INPUTS ARE FOR TRAVERSE OPERATIONS ONLY.
12-15 SHIRTSLEEVE SETUP (PHASE 1). INPUT THE NUMBER OF MANHOURS TO SETUP THE EXPERIMENT IN A SHIRTSLEEVE ENVIRONMENT. THE FOUR DIGIT CODE USES THE FIRST TWO DIGITS FOR THE SIGNIFICANT FIGURES, AND THE LAST TWO DIGITS FOR THE
EXPONENT TO THE BASE 10. FOR NEGATIVE EXPONENTS ADD 50 TO THE ABSOLUTE VALUE.
EXAMPLE - $2001 = 2.0 \times (10)^{1} = 20 \text{ MAN-HOURS} - 1$
$2051 = 2.0 \times (10) = 0.2 \text{ MAN-HOURS}$
16-19 SHIRTSLEEVE OPERATE (PHASE 2). INPUT THE NUMBER OF MANHOURS TO OPERATE THE EXPERIMENT IN A SHIRTSLEEVE ENVIRONMENT. USE THE FOUR DIGIT CODE DESCRIBED ABOVE.
20-23 SHIRTSLEEVE ANALYSIS (PHASE 3). INPUT THE NUMBER OF MANHOURS TO ANALYZE THE EXPERIMENT IN A SHIRTSLEEVE ENVIRONMENT USE THE FOUR DIGIT CODE DESCRIPED ABOVE.
24-27 SHIRTSLEEVE TEARDOWN (PHASE 4). INPUT THE NUMBER OF MANHOURS TO TEARDOWN THE EXPERIMENT IN A SHIRTSLEEVE ENVIRONMENT
USE THE FOUR DIGIT CODE DESCRIBED ABOVE.
28-31 SPACESULT SETUP (PHASE 1). SAME AS 12-15 ONLY IN SPACESULT.
32-35 SPACESULT OPERATE (PHASE 2). SAME AS 16-19 ONLY IN SPACESULT.
36-39 SPACESHIT ANALYSIS (PHASE 3). SAME AS 20-23 ONLY IN SPACESUIT.
40-43 SPACESULT TEAPDOWN (PHASE 4). SAME AS 24-27 ONLY IN SPACESULT.
THE NEXT EIGHT INPUTS ARE FOR BASE OPERATIONS ONLY.
44-47 SHIRTSLEEVE SETUP (PHASE 1). SAME AS 12-15 ONLY AT BASE.
49-51 SHIPTSLEEVE OPERATE (PHASE 2). SAME AS 15-19 ONLY AT BASE.
52-55 SHIRTSLEEVE ANALYSIS (PHASE 3). SAME AS 20-23 ONLY AT BASE.

		CARD I.D. NO.	8-02	PAGE 2 OF 2	
	EXPERIMENT I	<u>NPUT - TYPE 3 (</u>	CARD (CONTINU	<u>ED)</u>	
56-59	SHIPTSLEEVE TEA	RDAWN (PHASE 4)	L. SAME AS 2	<u>4-27 ONLY AT BA</u>	<u>SE.</u>
60-63	SPACESUIT SETUP	(PHASE 1). 54	ME AS 29-31	ONLY AT BASE.	
64-67	SPACESULT OPERA	TE (PHASE 2).	SAME AS 32-3	5 ONLY AT BASE.	
68-71	SPACESULT ANALY	SIS (PHASE 3).	SAME AS 36-	30 ONLY AT PASE	
72-75	SPACESULT TEARD	OWN (PHASE 4).	SAME AS 40-	43 ONLY AT BASE	•
		<u></u>			
P _140_4	a da antiga da Canada da Angela da Angela da Angela da Angela da Angela da Angela da Angela da Angela da Angela				
				· · · · · · · · · · · · · · · · · · ·	
<u></u>		-			
	<u></u>				

-

Appendix H

DETAILED INSTRUCTIONS FOR PREPARING COMPUTER DATA UPDATE CARDS

(Ref. Paragraph 10.8)

COMPUTER CONTROL - TAPE RECORD IDENTIFIER

 THIS CARD IS USED BY THE COMPUTER TO FIND A SPECIFIC DATA RECORD

 WHICH WILL BE OPERATED UPON BY THE INPUT AND UPDATE COMPUTER ROUTINE.

 THE IDENTIFIER OF THAT DATA RECORD IS INPUT HERE.

 THE IDENTIFIER OF THAT DATA RECORD IS INPUT HERE.

 TO

 I. UPDATE THE EXISTING DATA RECORD

 2. DELETE THE ENTIRE DATA RECORD

 3. COPY THE ENTIRE DATA RECORD, DNTO ANOTHER TAPE

1-12 DATA RECORD IDENTIFIER. INPUT THE IDENTIFIER FOR THE SPECIFIC

DATA RECORD REQUIRED. INPUT AS FOLLOWS-

		•	
	DATA RECORD CATEGORY	COLUMNS	
· · · · · · · · · · · · · · · · · · ·	MISSION	1-8	
	MODE	1-12	
	EMERGENCY MODE	1-12	-
	SYSTEM	1-6	
· · · · · · · · · · · · · · · · · · ·	SUBSYSTEM S.D.T.A. ADVANCE	1-6	
	EXPERIMENT	1-12	
	والمتراجية فبتبعث والمراجب فالتار فترار الباري والباجلي فروار والمنجوب والمتحد ومتراجي والمراجع والمتراجع والمراجع	And the owner of the owner of the owner of the owner, where the owner, whe	-

77-80 SEQUENCE NUMBER. INPUT THE FOLLOWING NUMBER- 2021

CARD 1.D. NO. 2-03

COMPUTER CONTROL - A-FORMAT CHANGES

1

THIS CARD IS USED TO REVISE ANY DATA RECORD WORD THAT IS WRITTEN IN A-FORMAT DNLY.

- 1-12 DATA RECORD IDENTIFIER. INPUT IDENTIFIER OF SPECIFIC DATA RECORD FOR WHICH AN A-FORMAT CHANGE IS REQUIRED. USE SAME IDENTIFIER AS INPUT ON 2-02 CAPD.
- 16-18 DATA ADDRESS. INPUT THE ADDRESS OF THE FIRST A-FORMAT DATA WORD TO BE CHANGED. THIS ADDRESS CAN BE FOUND IN THE ACCOMPANYING TABLE FOR THE DESIRED DATA RECORD CATEGORY. BE SURE TO RIGHT-ADJUST THE INPUT NUMBER. THAT IS, BE SURE THAT THE LAST DIGIT OF THE INPUT APPEARS IN COLUMN 18.
- 24 NUMBER OF CONSECUTIVE ENTRIES (ALL A-FORMAT) TO BE REVISED START-ING WITH THE ENTRY WHOSE ADDRESS APPEARS IN COLUMNS 16-18 ABOVE. ONLY A MAXIMUM OF 8 CONSECUTIVE ENTRIES CAN BE CHANGED WITH ONE 2-03 CARD.
- 26-31 NEW A-FORMAT WORD DESIRED IN FIRST DATA ADDRESS. INPUT THE WORD EXACTLY AS YOU WISH IT TO APPEAR. START THE INPUT IN COLUMN 26.
- 32-37 NEW A-FORMAT WORD DESIPED IN THE NEXT DATA ADDRESS. INPUT THE WOPD EXACTLY AS YOU WISH IT TO APPEAR. START THE INPUT TO THE LEFT OF THE SPACE PROVIDED.
- 38-43 SEE ABOVE.
- 44-49 SEE ABOVE.

50-55 SEE 4BOVE.

56-61 SEE 130VE.

- 62-67 SEE ABOVE.
- 68-73 SEE ABOVE.

77-PO SEQUENCE NUMBER. INPUT THE FOLLOWING NUMBER- 203L

	CARD I.D. ND. 2-04
	COMPUTER CONTROL - E-FORMAT CHANGES
THIS CA	RD IS USED TO REVISE ANY NUMBER IN THE DATA RECORD THAT IS
WRITTEN	I IN F-FORMAT (FLOATING DECIMAL POINT).
1-12	DATA RECORD IDENTIFIER. INPUT IDENTIFIER OF SPECIFIC DATA RECORD FOR WHICH AN E-FORMAT CHANGE IS REDUIRED. USE SAME
· · · · · · · · · · · · · · · · · · ·	IDENTIFIEP AS INPUT ON 2-02 CARD.
16-19	DATA ADDRESS INDUT THE ADDRESS OF THE EIDST E-EDDMAT DATA
10-13	NUMBER TO BE CHANGED. THIS ADDRESS CAN BE FOUND IN THE
	ACCOMPANYING TABLE FOR THE DESIRED DATA RECORD CATEGORY.
	<u>SE SURE TO RIGHT-ADJUST THE INPUT NUMBER, THAT IS, BE</u>
	SURE THAT THE LAST DIGIT OF THE INPUT APPEARS IN COLUMN IN,
2.4	NUMBER OF CONSECUTIVE ENTRIES (ALL F-FORMAT) TO BE REVISED
	STARTING WITH THE ENTRY WHOSE ADDRESS APPEARS IN COLUMNS
	16-18 ABOVE. ONLY A MAXIMUM OF 5 CONSECUTIVE ENTRIES CAN
	NE CHANGED WITH ONE ZEUG CARD.
26-35	NEW F-FORMAT NUMBER DESIRED IN FIRST DATA ADDRESS. INPUT THE NUMBER EXACTLY AS YOU WISH IT TO APPEAR. INPUT THE NUMBER
	ANYWHERE IN THE 10 SPACES PROVIDED. BE SURE TO SHOW LOCATION
	DE DECIMAL PDINT. A BLANK ENTRY IN THE 10 SPACES WILL BE DEAD AS A ZERO BY THE COMPUTED.
36-45	NEW F-FORMAT NUMBER DESIRED IN THE NEXT DATA ADDRESS. SAME
46-55	SEE ABOVE.
56-65	SEE ABOVE.
66-75	SEE ABOVE.
77-90	SEQUENCE NUMBER. INPUT THE ECLLOWING NUMBER- 2041
Table H-1

LOCATION OF DATA ON MODE FIXED LENGTH RECORD

Address	Variable Name	Format	Original Input Card Name
3	Active Mode Crew Level	F	· · · · · · · · · · · · · · · · · · ·
4	Total Mode Crew Level	F	`
5	Mode Staytime	\mathbf{F}	
6	Sample Return Capability	F	Mode Input-General
7	Min. Decimal Fraction	\mathbf{F}_{\perp}	
8	Deliv. Date-Minor Sci. Equip.	F	
9	Flag		
10	No. of Payloads	F	
11	Total Time for Base Deploy.	F	
12	Deploy. mhr – Shirtsleeve	F	
13	Deploy. mhr – Spacesuit	\mathbf{F}	
14	Mhr of Suit Lifetime	\mathbf{F}	Mod Input-Operations
15	Mhr of Backpack Lifetime	\mathbf{F}	
16	Mass of Suit Replacement	\mathbf{F}	
17	Mass of Backpack Replacement	\mathbf{F}	
18 - 33	Nonrecur., Eng., Mfg., Test, Sup.	F	
34 - 49	Nonrecur., Facilities	\mathbf{F}	Mode
50 - 65	Recurring; Eng., Mfg., Test	\mathbf{F}	Cost Curve
66 - 81	Recurring; Oper. Support	F	
82	1st Payload Launch ID	F	
83	Related Launch ID	F	
84	Min. 6-mo Periods	F	
85	Max. 6-mo Periods	\mathbf{F}	
86	Mass Margin Available	F	
87-91	2nd Payload		Mode Input Payload
92-96	3rd Payload	•	Launch Identification
97-101	4th Payload	•	
102-106	5th Payload	•	

Table H-1 (Cont.)

Address	Variable Name	Format	Original Input Card Name
107-111	6th Payload		
112 - 116	7th Payload		
117 –1 21	8th Payload Launch ID	F	Mode Input Payload
			Launch Identification
	1st Payload, 1st System		
122	System Iden.	Α	
123	No. Men Supported	F	
124	Delivery Date	F	
125	Expend -Special Oper.	. F	
126-129	1st Payload, 2nd System		
•	•		· · ·
•	•		
• •	•		
178-181	1st Payload, 15th System		Mode Input-Systems on
182-185	2nd Payload, 1st System		Each Launch
•	•		
•	•		
•	•	,	
238-241	2nd Payload, 15th System		
242-301	3rd Payload		
302-361	4th Payload		
362-421	5th Payload		
422-481	6th Payload		
482-541	7th Payload		
542-601	8th Payload		
602-603	Identifier for Crew Rescue Emergency Mode	А	
604-605	Identifier for Logistics Emergency Mode	А	Mode Input-Emergency

H-6

Table H-2

.

LOCATION OF DATA FOR EMERGENCY MODE FIXED LENGTH RECORD

Address	Variable Name	Format ∕	Original Input Card Name
11-26	Nonrecurring R&D	F	
27 - 42	Nonrecurring Cost of Facilities	F	Emergency Mode Cost
43 - 58	Recurring Hardware Procurement	F	Cost Curve
59 - 74	Operations Support	F	
	Only Payload, 1st System		
75	System Identifier	А	
76	No. Men Supported	F	
77	Launch Lead Time	\mathbf{F}	
78 -	Expendables - Spec. Oper.	F	Systems on the Payload
79- 82	Only Payload, 2nd System	A, F, F,	
	•	F	
	•		
131-134	Only Payload, 15th System	A, F, F, F	

•

Table H-3

LOCATION OF DATA FOR SYSTEM FIXED LENGTH RECORD

Address	Variable Name	Format	Original Input Card Name
2-5	System Name	А	
6	Procurement Quantity Unit	F	System Input-General
7	Max. Mass for all Expend.	F	
8	Max. Mass for LFV Propellant	F	
9-38	Subsys. Ids., Deliv. Date - Pairs	A,F	System Input-Subsystem S.O.T.A. Iden.
39-54	Nonrecurring Cost	F	
55-70	Nonrecurring Facilities Cost	F	Cost System Curve
71-86	Recurring Cost	F	
. 87 – 102	Recurring Operations Support	F	
103-122	(Copies Nbrd., Dec. Frac.) – Pairs	(F,F)	Table of Hardware Cost Reduction for Add. Copies
123	Zero Pt. for Flight Table	F	R&D Flight Table
124-143	(No. Flights, Veh. Used) - Pairs	(F, A)	
144-150	1st Crew Level → Boiloff	F	
151-157	2nd Crew Level \rightarrow Boiloff	F	System Expendables
158-164	3rd Crew Level \rightarrow Boiloff	F	
165-171	4th Crew Level → Boiloff	F	
172-178	5th Crew Level \rightarrow Boiloff		
179-185	1st Crew Level → Non-Sci. Oper./Stop (Spacesuit)	F	
186-192	2nd Crew Level → Non-Sci. Oper./Stop (Spacesuit)	F	System Man-Hour Require-
193-199		•	ments for Nonscientific Operations
200-206		•	▲ ···-
207-213	5th Crew Level → Non-Sci. Oper./Stop (Spacesuit)		

LOCKHEED MISSILES & SPACE COMPANY

Table H-3 (Cont.)

Address	Variable Name	Format	Original Input Card Name				
214-223	(Crew Level, Max. Staytimes)—Pairs	(F,F)	System Input-Per. Shelter Only				
224-229	1st Crew Level → Flag	F					
230-235	2nd Crew Level \rightarrow Flag	F					
236-241	3rd Crew Level → Flag	F	System Input-LRV Only				
242-247	4th Crew Level → Flag	F					
248-253	5th Crew Level → Flag	F					
254-258	1st Crew Level → Flag	F	, ,				
259-263	2nd Crew Level \rightarrow Flag	F					
264-268	3rd Crew Level \rightarrow Flag	F	System Input-LFV Only				
269-273	4th Crew Level \rightarrow Flag	F					
274-278	5th Crew Level → Flag	F					
279-298	(No. Stops, Max. Single Excursion Range) – Pairs	(F , F)	System Input-LFV Stops vs. Range Table				
29 9-3 48	Experiment Iden.	А	System Input-Major Scien. Equip. Only				
349-360	(Flight System, Lead Time)—Pairs	(A,F)	System Input-Transporta- tion System Only				

.

Table H-4

LOCATION OF DATA FOR SUBSYSTEM SOTA ADVANCE FIXED LENGTH RECORD

Address	Variable Name	Format	Original Input Card Name
2-5	Subsystem Name (1st line)	A	Subsystem SOTA
6-9	Subsystem Name (2nd line)	A	Advance-General
10-25	Nonrucurring R&D	F	Subsystem Cost Curve
26-41	Nonrecurring Cost of Facilities	F	

Note: A total of 211 word locations have been set aside for subsystems, but only 41 are currently allocated.

LOCKHEED MISSILES & SPACE COMPANY

Table H-5

LOCATION OF DATA FOR EXPERIMENT FIXED LENGTH RECO	\mathbf{RD}
---	---------------

Address	Variable Name	Format	Original Input Card Name
3-6	Experiment Title	А	
7	Number Men Required	А	
8	Crew Skill	Α	
9	Total Equipment Mass	А	
10	Total Equipment Volume	А	
11	Total Equipment Energy	А	
12	Peak Power	А	Experiment Input-Type 1 Card
13	Dev. Time	А	
14	Year of Earliest Availability	Α	
15	Nonrecurring Cost	А	
16	Est. 1st Item Cost	Α	
17	Total Earth Return Mass	А	
18	Setup Shirtsleeve Traverse	А	
19	Operate Shirtsleeve Traverse	А	
20	Analysis Shirtsleeve Traverse	Α	
21	Take Down Shirtsleeve Traverse	А	
22	Setup Spacecuit Traverse	Α	
23	Operate Spacesuit Traverse	А	
24	Analysis Spacesuit Traverse	А	
25	Take Down Spacesuit Traverse	Α	Experiments Input-Type 3 Card
26	Setup Shirtsleeve Base	А	
27	Operate Shirtsleeve Base	А	
28	Analysis Shirtsleeve Base	А	
29	Take Down Shirtsleeve Base	А	
30	Setup Spacesuit Base	А	
31	Operate Spacesuit Base	А	
32	Analysis Spacesuit Base	А	
33	Take Down Spacesuit Base	А	

LMSC-A847943 Vol. I

Appendix I

DETAILED INSTRUCTIONS FOR PREPARING MISSION-MODE COMPARISON CONTROL CARDS

(Ref. Paragraph 10.9)

I-1

LOCKHEED MISSILES & SPACE COMPANY

PROGRAM LEVEL INPUT - MISSION-MODE IDENTIFICATION

THIS CARD IS USED TO INPUT THE REQUIPED IDENTIFIERS FOR EACH MISSION AND MODE MATCH THAT IS TO BE VERIFIED BY THE COMPUTER. THERE IS NO LIMIT TO THE NUMBER OF SUCH MATCHES THAT CAN BE VERIFIED IN ONE RUN.

- 1-12 EXPLORATION PPOGRAM IDENTIFIER. START THE IDENTIFIER IN COLUMN 1.
- 13-20 MISSION IDENTIFIER. INPUT AN IDENTIFIER IN THE EIGHT COLUMNS <u>SPECIFIED THAT WILL UNIONELY LABEL THE MISSION BEING CONSID-</u> ERED. THE FOLLOWING INPUT FORMAT WILL BEST MATCH THE COM-PUTER INTERNAL LOGIC AND OUTPUT PRINT ROUTINE.

13-15 SCIENTIFIC PROGRAM IDENTIFIER. USE ANY COMBINATION OF LETTERS AND NUMBERS TO LABEL THE SCIENTIFIC PROGRAM UNDER CONSIDERATION. START THE INPUT IN COLUMN 13.

> 17-18 MISSION MIX IDENTIFIER. IDENTIEY THE GROUP OF MIS-SIONS BEING DERIVED TO ACCOMPLISH THE SCIENTIFIC PROGRAM GIVEN ABOVE.

19-20 NUMBER OF THIS MISSION IN THE SEQUENCE OF MISSIONS.

25-36 MODE IDENTIFIER. USE THE NUMBER SHOWN IN THE MIMOSA DATA BOOK. START THE NUMBER IN THE FIRST SPACE PROVIDED.

50-55 SHIRTSLEEVE MAN-HOUR MULTIPLIER. THE MAN-HOUR DATA FOR EXPERIMENTS GIVEN BY NORTH AMERICAN DOES NOT ACCOUNT FOR ANY DIFFERENCE IN GRAVITY BETWEEN FARTH AND MOON. INPUT HERE THE FACTOP BY WHICH EXPERIMENT SHIPTSLEEVE MAN-HOURS WILL BE MULTIPLIED TO ACCOUNT FOR THIS DIFFERENCE. PLACE THE NUMBER ANYWHERE IN THE SPACE PROVIDED. BE SURE TO SHOW LOCATION OF DECIMAL POINT.

60-65 SPACESULT MAN-HOUR MULTIPLIER. THE MAN-HOUR DATA FOR EXPER-IMENTS GIVEN BY NORTH AMERICAN IS FOR 1 G SHIRTLSFEVE EFFORT. INPUT HERE THE FACTOR BY WHICH EXPERIMENT SPACESULT MAN-HOURS WILL BE MULTIPLIED TO ACCOUNT FOR SPACESULT EFFORT ON THE MOON. PLACE THE NUMBER ANYWHERE IN THE SPACE PROVIDED. BE SURE TO SHOW LOCATION OF DECIMAL POINT.

77-90 SEQUENCE NUMBER. INPUT THE FOLLOWING NUMBER- 211L

I-2

PROGRAM LEVEL INPUT - TRAVERSE-VEHICLE IDENTIFICATION

THIS CARD IS USED TO IDENTIFY THE TRAVERSES PERFORMED BY EACH FLYING AND ROVING VEHICLE IN THE MODE IDENTIFIED IN CARD I.D. 2-11. INPUT OF UP TO 24 TRAVERSES PER VEHICLE IS PERMITTED. AS THERE IS SPACE FOR IDENTIFYING 6 TRAVERSES PER CARD, A MAXIMUM OF 4 CARDS IS ALLOWED PER VEHICLE.

START WITH FLYING VEHICLES FIRST. IF THERE ARE NO FLYING OR ROVING VEHICLES USED IN THE MISSION, INPUT ONLY COLUMNS 1-12 AND 77-80.

1-12 MODE IDENTIFIER. USE THE NUMBER INPUT IN COLUMNS 25-36 OF THE 2-11 CARD.

13~18 TRAVERSE VEHICLE IDENTIFIER. USE THE NUMBER SHOWN IN THE MIMOSA DATA BOOK. USE ONLY THE SIX DIGITS. DO NOT INCLUDE THE DASH SEPARATING THE DIGITS.

19-24 IF THE VEHICLE IN COLUMNS 13-18 IS A TRAILER USED IN COMBINATION WITH AN LRV, INPUT THE LRV IDENTIFIER HERE. USE THE NUMBER SHOWN IN THE MIMOSA DATA BOOK. USE ONLY THE SIX DIGITS. DO NOT INCLUDE THE DASH SEPARATING THE DIGITS.

25-30 CREW LEVEL. INPUT THE NUMBER OF CREWMEN PARTICIPATING IN THE FOLLOWING LIST OF TRAVERSES. SHOW LOCATION OF DECIMAL POINT.

31-34 TRAVERSE IDENTIFIER. INPUT IDENTIFIER OF TRAVERSE PER-FORMED BY ABOVE VEHICLE AT THE SPECIFIED CREW LEVEL. THIS IDENTIFIER SHOULD TAKE THE FORM F-XX OR R-XX WHERE F = FLYING TRAVERSE

F = FLYING TRAVERSE R = ROVING TRAVERSE

XX = SEQUENCE NUMBER (01, 02, 03, ETC.) IN -INCREASING ORDER MAKE SURE THE IDENTIFIER GIVEN HERE CONFORMS TO DATA INPUT

WITH MISSION, PARTICULARLY CARDS 3-03 AND 3-04.

7-40 SEE ABOVE.

43-46 SEE ABOVE.

40-52 SEE AROVE.

55-58 SEE ABOVE.

61-64 SEE ABOVE.

X = 1, 2,..., L DEPENDING ON THE NUMBER OF CARDS USED. THE LETTER L MUST BE USED ON THE LAST 2-12 CARD FOR FACH MISSION.

Appendix J

DETAILED INSTRUCTIONS FOR PREPARING INTEGRATION AND COST CONTROL CARDS

(Ref. Paragraph 10.10)

CARD I. D. 2-20 INTEGRATION AND COSTING - GENERAL INPUT THIS CARD IS USED TO SUPPLY GENERAL CONTROL INFORMATION FOR THE INTEGRATION AND COSTING ROUTINE OF THE MIMOSA COMPUTER PROGRAM.

1-36 EXPLORATION PROGRAM NAME. START INPUT IN COLUMN 1. 41-50 NUMBER OF SATURN 5 LAUNCHES PERMITTED PER CALENDAR YEAR BY EXISTING LAUNCH FACILITIES WITHOUT REQUIRING LAUNCH RATE IMPROVMENT COSTING. INPUT THE NUMBER ANYWHERE IN THE 10 SPACES PROVIDED. BE SURE TO SHOW LOCATION OF THE DECIMAL POINT. EMERGENCY MODE USAGE FLAG. INPUT AS FOLLOWS-55 O (OR BLANK) = EMERGENCY MODES ARE NOT TO BE CONSIDERED. = EMERGENCY MODES ARE TO BE CONSIDERED 1 BACKUP SYSTEMS FLAG. INPUT AS FOLLOWS-60 O (OR BLANK) = BACKUP SYSTEMS ARE NOT TO BE PROCURED = BACKUP SYSTEMS ARE TO BE PROCURED 1 EXCEED LAUNCH RATE FLAG. INPUT AS FOLLOWS-65 O(OR BLANK) = LAUNCH RATE VS. CALENDAR YEAR (2-21 CARDS)MAY NOT BE EXCEEDED = LAUNCH RATE VS. CALENDAR YEAR (2-21 CARDS) 1 MAY BE EXCEEDED SEQUENCE NUMBER. INPUT THE FOLLOWING NUMBER-77-80 220L 1

CARD I. D. 2-21

PAGE 1 OF 2

LAUNCH RATE LIMIT VS CALENDAR YEAR

LOCKHEED HAS DEVELOPED TWO VERSIONS OF THE INTEGRATION AND COST ROUTINE. EACH VERSION ALLOWS CALCULATION OF COSTS FOR 21 CONSECUTIVE CALENDAR YEARS. ONE VERSION IS GOOD FOR 1967 THROUGH 1987, THE OTHER VERSION WORKS FOR 1930 THROUGH 1990.

THIS CARD IS USED TO TELL THE COMPUTER THE MAXIMUM NUMBER OF SATURN 5 LAUNCHES DESIRED IN EACH OF THE 21 YEARS FOR WHICH COSTS ARE CALCULATED. THE YEAR THAT IS INTENDED FOR EACH INPUT VALUE IS DETERMINED BY THE POSITION OF THE INPUT NUMBER ON THE CARD. THEREFORE CARE MUST BE TAKEN TO MAKE SURE THE INPUT ON THIS CARD AND THE DESIRED VERSION OF THE COST ROUTINE INSTRUCTIONS ARE COMPATIBLE.

DEPENDING ON THE VERSION OF THE INTEGRATION AND COST POUTINE TO BE USED, INPUT AS FOLLOWS. INSERT A NUMBER ON THE CARD IN THE SPECIFIED COLUMNS FOR THE MAXIMUM NUMBER OF SATURN 5 (AND UPRATED SATURN 5) LAUNCHES DESIRED FOR THE CALENDAR YEAR GIVEN IN THE TABLE ON THE FOLLOWING PAGE. PLACE EACH NUMBER ANYWHERE IN THE 6-COLUMN SPACE PROVIDED. BE SURE TO SHOW THE LOCATION OF THE DECIMAL POINT. DO NOT INPUT THE NUMBER OF THE YEAR (E.G., 1970) ITSELF. INPUT ONLY THE LAUNCH LIMIT.

BOTH CARDS MUST BE FILLED OUT. A BLANK IN ANY 6-COLUMN SPACE IS IN-TERPRETED AS A ZERO BY THE COMPUTER.

INSERT THE FOLLOWING SFOUENCE NUMBERS IN COLUMNS 77-80

CARD 1 - 2211 CARD 2 - 221L

CARD 1.0. 2-21

LAUNCH RATE LIMIT VS CALENDAR YEAR (CUNTINUED)

POSITION OF LAUNCH RATE LIMIT VALUES ON EACH REQUIRED CARD

COLUMN	<u>APPLICABL</u> 1967 - 19	E YEARS 37 VERSION	APPLICARLE YFARS 1970 - 1980 VERSION				
NUMBERS	CARD 1	CARD 2	CARD 1	CARD 2			
6	1967	1978	1970	1991			
7 - 12	1968	1670	1971	1982			
13 - 19	.1969	1980	1072	1983			
10 - 24	1970	1681	1973	1994			
25' - 30	1071	1092	1974	1695			
31 - 36	1972	1093	. 1975	1086			
37 - 42	1073	1984	1976	1997			
43 - 49	1974	1635	1977	1988			
<u>4° - 54</u>	1075	1986	1078	lçsc			
55 - 60	1976	1097	1079	1600			
61 - 65	1977		1080				

J-4

CAPD I. D. 2-22

MISSION-MODE COMBINATIONS

THIS CARD IS USED TO INPUT THE REQUIPED IDENTIFIERS FOR EACH MISSION-MODE COMBINATION TO BE INTEGRATED AND COSTED IN ONE EXPLORATION PROGRAM. UP TO 50 COMBINATIONS MAY BE INPUT AT ONE TIME. AS THERE ARE 3 COMBINATIONS INPUT PER CARD, UP TO 17 CARDS MAY BE USED. COMBIN-ATIONS MAY BE INPUT IN ANY ORDER PROVIDED THE SCHEDULING OF ONE MISSION DOES NOT DEPEND ON THE SCHEDULING OF ANOTHER MISSION (SEE 3-01 CAPD).

CAUTION-- TWO INSTRUCTION TAPES ARE AVAILABLE FOR COST CALCULATIONS. ONE WILL CALCULATE COSTS FROM 1967A THROUGH 19878. THE OTHER WILL CALCULATE COSTS FROM 1970A THROUGH 19903. BE SURE TO NOTE WHICH TAPE YOU ARE USING AND, MAKE SURE THAT NO MISSION LAUNCHES ARE SCHEDULED OUTSIDE THE PERMITTED RANGE OF DATES. IF LAUNCHES OCCUR OUTSIDE THE PERMITTED RANGE, THE COMPUTER RUN WILL BE ABORTED AUTOMATICALLY.

- 1-8 MISSION IDENTIFIER. INPUT THE IDENTIFIER OF A DESIRED MISSION EXACTLY AS IT APPEARS IN COLS. 1-8 OF THE 3-01 CARD.
- MODE IDENTIFIER. INPUT THE MODE WHICH IS TO BE COSTED WITH THE MISSION OF 1-8. MAKE SURE IT IS INPUT EXACTLY AS 9-20 IT APPEARS IN COLS. 1-12 OF THE 4-01 CARD.
- 21-25 MISSION START DATE OVERRIDE. IF YOU DESIRE TO CHANGE THE MISSION START DATE FOR THIS COSTING BOUTINE FROM WHAT IS SHOWN IN COLS. 31-35 OF THE 3-01 CARD, INPUT A NEW DATE HERE. IF NO INPUT IS MADE THEN THE DATE SHOWN ON THE 3-01 CARD WILL BE USED BY THE COMPUTER.
- SEE 1-9 ABOVE EXCEPT NEXT MISSION IN THE PROGRAM. 26-33
- 34-45 SEE 9-20 ABOVE EXCEPT NEXT MODE IN THE PROGRAM.
- SEE 21-25 ABOVE. 46-50
- SAME AS 26-33 ABOVE. 51-59
- 59-70 SAME AS 34-45 ABOVE.

SEE 21-25 ABOVE. 71-75

SEQUENCE MUMBER. INPUT THE MUMBER 2224 WHERE 77-80 ENP CARDS A = NUMBER OF THE CARD IN THE SET. 10-16, USE ONLY THE LAST DIGIT. BE SURE TO INPUT THE LETTER L IN COLUMN 90 OF THE LAST CAPD IN THE SET.

CARD I. D. 2-23

PAGE 1 OF 2

+

LAUNCH RATE IMPROVEMENT COST CURVE

TWO COST CURVES FOR FACILITY LAUNCH RATE IMPROVEMENT ARE INPUT FOR EACH
INCREMENTAL STEP IN LAUNCH RATE IMPROVEMENT. EACH COST CURVE MAY COVER
A PERIOD OF UP TO 7.5 YEARS. FACH CARD HAS PROVISIONS FOR FIVE COST
PATRIES OF 6 4TATHS DUPATTON FACE. THUS, A MAXIMUM OF 3 CARDS (15 ENTRIES
STEDS IN LAUNCH DATE IMDOOVEMENT MAY BE INDUT - HENCE, THIS SET MAY
COMPRISE A MAXIMUM OF AO CARDS.
13-16 ANNUAL LAUNCH RATE BEFORE IMPROVEMENT IS MADE. SEE NOTE *
BELOW.
17-20 ANNUAL LAUNCH RATE AFTER IMPROVEMENT IS MADE. SEE NOTE #
BELOW.
21-25 7ERD DOINT INCERT HERE THE NUMBER OF 6-MONTH TIME REPIONS
THAT OCCUR FROM START OF FUNDING UP TO AND INCLUDING THE
6-MONTH TIME PERIOD WHEN THE LAUNCH FACILITY FIRST OPERATES
AT THE NEW LAUNCH RATE SHOWN IN COLUMNS 17-20 ABOVE.
NOTE- THIS ZERO POINT ENTRY NEED ONLY BE PUT ON THE FIRST
CARD OF THE SET THAT WILL DESCRIBE EACH COST CURVE.
SEE NOTE * BELOW.
24-25 COST ENTRY IN MILLIONS OF DOLLARS EOD 6-MONTH EUNDING DEDIOD
SEE NOTE & RELOW.
36-45 SAME AS ABOVE.
46-55 SAME AS ABOVE.
56-65 SAME AS ABOVE.
·

			·											
<u> </u>			CAR	D I.D	• NO•	2-2	23			ΡΔ	GE 2	OF 2		
	LA	UNCH P	ATE	 Impro	VEMEN	TCC	IST (CURV	E (C	<u>ONTI</u>	NUED			
77-90	SEQUENCE	CUDDIN	<u>.</u> 				S SI		BEL	<u></u>		ATE		223X
	NONRE	CURRIN	NG FA	CILIT	IES C	OST	ÜF	1 S T	STEP	IN	LAUNC	H RA	ΤЕ	224X
	NONR E NONRE	CURR IN	NG R NG FA	AND D CILIT	COST IES C	OF OST	2ND 05	STE 2ND	P IN STEP	L AU I N	NCH F LAUNC	ATE H RA	TE	225X 226X
														• • •
	NONRE NONRE	CURRIN CHRRIN	IG R IG FA	AND D	COST IES C	nf OST	10TH	H STI LOTH	EP I STE	N LA P IN	UNCH LAUN	RATE	ATE	241X 242X
	WHERE	x = L/	AST D	IGIT	OF SE						AS FC	ILOW	<u>S</u> -	
		1 2	- [F - [F	FIRS	T CAR ND CA	D IN PD I	I SEI N SE		CE DI NCE (F 2 DF 3	DR 3 CARD	CARD S.	S.	
NOTE *	- PLACE J SURE	HE INF	PUT NU	JMBER E LOC	ANYW ATION	HERE OF	I N THE	THE	SPA(IMAL	CE P POI	ROVIČ NT.	ED.	B E	
											-			
													· ·	
												•••••		